# Sustainability Failures: The Challenge of Sustaining the NP Role and Other Innovations in Primary Health Care

by

Raelene Marceau

A thesis submitted in partial fulfillment of the requirements for the degree of

Doctor of Philosophy Faculty Nursing University of Alberta

© Raelene Marceau, 2020

#### Abstract

**Background:** Sustaining innovations in health systems is a topic of increasing importance to stakeholders interested in creating sustainable primary health care (PHC) reform. The Nurse Practitioner (NP) role, a PHC innovation, was initially introduced in Canada in the 1970's and re-implemented with PHC restructuring in the 1990's. Despite extensive evidence that NPs are a cost-effective means to providing comprehensive, high quality care, the role is not optimized throughout much of Canada. Understanding the contextual, policy and decision-making factors that influence sustainability of the NP role and other PHC innovations is an important addition to the current literature.

**Purpose:** The purpose of this thesis was to examine the sustainability failure of a PHC innovation, an NP primary care clinic in Alberta, Canada.

**Methods:** This thesis combines three linked scientific papers: 1) a scoping review of peer reviewed and grey literature of clinic closures involving NPs, as well as the sustainability of PHC innovations and the NP role, 2) a theoretically informed critical analysis of NP funding models in Canada and 3) a single, exploratory case study of a NP primary care clinic closure in Alberta, Canada.

**Findings/Conclusion:** Findings demonstrate that there is a paucity of peer reviewed literature informing the sustainability of the NP role and other PHC innovations. Sustainability failures in our health system are due to deficiencies in provincial and national policies in flawed policy environments. New sustainability knowledge; an updated understanding of the sustainability of

innovations in PHC and a newly visioned conceptual framework for stakeholders to use when assessing sustainability of PHC innovations, have emerged from this work

#### Preface

This dissertation presents the method and findings of original research completed by Raelene Deanne Marceau (RDM) under supervision by her Chair and in collaboration with her committee. Three consecutive academic papers present a scoping review of the literature, a critical analysis of NP funding mechanisms and a single case study that examined the contextual, policy and decision-making factors that influenced the closure of a NP primary care clinic in Alberta, Canada. The thesis includes the following five chapters;

#### Chapter One: Introduction

Chapter 1 provides a detailed outline of the thesis, background and theoretical foundation supporting this research.

#### Chapter Two: Sustaining Primary Health Care Programs and Services: A Scoping Review

Chapter 2 presents a scoping review which is in press with Policy, Politics, & Nursing Practice. The scoping review was undertaken to inform the case study described in Chapter 4. Raelene Marceau (RM) was the lead author. Co-authors Kathleen F. Hunter (KFH), Tammy O' Rourke (TO) and Stephanie Montesanti (SM), the thesis committee, collaborated on the conceptualization of the review, provided guidance and expert opinion throughout the review process, and critically analyzed and edited the manuscript.

Marceau, R., Hunter, K., Montesanti, S. and O' Rourke, T., 2020. Sustaining Primary Health Care Programs and Services: A Scoping Review. *Policy, Politics and Nursing Practice*, In Press. **Chapter Three:** A Critical Analysis of Funding and Compensation Models: Sustainability of the Nurse Practitioner Role in Canada

Chapter three presents a publication in process for the Journal of Nurse Practitioner. This publication describes a theoretically-informed critical analysis comparing Canada's current NP funding models. The analysis provides evidence of the need for future research to assist health care leaders, governments, funders and policy-makers in supporting multiple NP roles to meet the demands of a variety of patient populations. RM conceived and designed the analysis with

input from TO and KFH. RM performed the critical analysis and wrote the manuscript with expert guidance, input and edits from KFH, TO and SM.

Marceau, R., Montesanti, S., Hunter, K. and O' Rourke, T., 2020. A Critical Analysis of Funding and Compensation Models: Sustainability of the Nurse Practitioner Role in Canada. *Journal of Nurse Practitioners*, Manuscript in Preparation.

**Chapter Four:** Sustainability Failures in Primary Health Care: The Case of an NP Clinic Closure

Chapter four presents a draft manuscript in preparation for submission to Social Science and Medicine. This publication in process describes the methods and findings from a single case study examining a NP primary care clinic closure in Alberta, Canada. The case study provides an in-depth explanation of why the clinic closed, the contextual factors, the actors involved and the decision-making process that led to clinic closure. Lessons to improve understanding and better inform stakeholders about this sustainability failure and important concepts for consideration in sustaining the NP role and other PHC innovations are presented. RM conceived and designed the study and obtained ethics approval, with guidance and input from TO and KFH. RM, TO and KFH performed data analysis and RM wrote the manuscript with input and editing from KFH, TO and SM.

Marceau, R., Montesanti, S., O'Rourke, T. and Hunter, K., 2020. Sustainability Failures in Primary Health Care: The Case of an NP Clinic Closure. *Social Science & Medicine*, Manuscript in Preparation.

#### **Chapter Five: Conclusions and Future Research**

Chapter five presents a summary of the thesis findings and introduces a new conceptual framework, directions for future research and a knowledge dissemination plan is outlined.

#### Acknowledgments

What has felt at times to be a circuitous path is now coming to a close and the completion of my thesis is in a large part due to the contributions, help, support and patience of some pretty incredible humans – only some of whom I can mention here. With boundless gratitude and appreciation, I would like to extend a heartfelt thank you to my PhD Co-Supervisors Dr. Kathleen Hunter and Dr. Tammy O' Rourke. I am grateful for their steadfast encouragement and unshakeable faith that I could finish this journey. They are clinical and academic role models, scholars, innovators and creators. I am extremely lucky to have had their guidance, their willingness to share knowledge and their determined focus to get me through to the end. A sincere thank you to Dr. Stephanie Montesanti who climbed on board at the beginning as a core committee member, was so patient, supportive and provided an extremely valuable lens from which to view this research. Thank you to my fellow online doctoral 'cohort' friends for their online support, friendship and feedback.

To say the words 'thank you' to my family – it just doesn't seem like enough – those two words can't convey how incredibly grateful I am. To my husband and two children, my parents, my in-laws and my sisters...completing a PhD when you are from a small, rural, northern community requires an army of family members to help out. It requires re-arranging schedules, taking on the role of mom and dad, patience, understanding, and most of all it entails a huge amount of personal sacrifice. I may never be aware of all the sacrifices that have been made, especially by my children. I hope I have role-modeled that hard work, sacrifice and determination are worthwhile endeavors in achieving a goal and that lifelong learning is a valuable pursuit. Thank you to work colleagues. Thank you to best friends. In particular, those who encouraged me through the last final months with daily laughter and love. Thank you for the texts, phone calls, pep talks, the putting up with crankiness and stress; the changing schedules, missed weeks of work due to PhD commitments - together you all provided the support system I needed to bring this dissertation to completion.

	vii
Table of Contents	
ABSTRACT	II
PREFACE	IV
ACKNOWLEDGMENTS	VI
FIGURE 1-1: POLICY TRIANGLE FRAMEWORK DIAGRAM	Х
CHAPTER ONE: INTRODUCTION, BACKGROUND AND THEORETICAL FOUNDATION <b>BOOKMARK NOT DEFINED.</b>	ERROR!
CHAPTER ONE: INTRODUCTION	1
BACKGROUND METHODS CONCLUSION	1 8 10
CHAPTER TWO: SUSTAINING PRIMARY HEALTH CARE PROGRAMS AND SERVICES: REVIEW (PAPER 1)	A SCOPING 20
ABSTRACT ERROR! BOOKMA	ARK NOT DEFINED.
BACKGROUND	21
PRIMARY CARE AND PRIMARY HEALTH CARE	21
SUSTAINABILITY IN PHC	22
METHODS	23
RESULTS	26
DISCUSSION STRENGTHS AND LIMITATIONS OF THE REVIEW	28 32
CONCLUSION	33
CHAPTER TWO REFERENCES	34
CHAPTER THREE: A CRITICAL ANALYSIS OF FUNDING AND COMPENSATION MODE SUSTAINABILITY OF THE NURSE PRACTITIONER ROLE IN CANADA (PAPER 2)	ELS: 63
ABSTRACT ERROR! BOOKMA	
INTRODUCTION	64

	viii
Objectives	64
Background	65
Method	69
RESULTS	70
FACTOR 5: INNOVATION	72
DISCUSSION	73
FUTURE DIRECTIONS	73
CONCLUSION	76

# CHAPTER THREE: REFERENCES

# CHAPTER 4: SUSTAINABILITY FAILURES IN PRIMARY HEALTH CARE: THE CASE OF AN NP CLINIC CLOSURE (PAPER 3) 94

77

Abstract Background Methods Results Discussion Summary	Error! Bookmark not defined. 96 100 103 120 123
CHAPTER 4 REFERENCES	123
CHAPTER FIVE: SUMMARY AND SYNTHESIS	194
Contributions Minimizing Policy Failure: Recommendations for Action Evidence and Evaluation Pilot 'Projectitis' Government Accountability and Transparency Utilizing an Ethical Lens and Person-Centred Policy Approa Theoretical Contribution Knowledge Dissemination Conclusion	ACH 196 198 199 200 200 201 202 208 211
CHAPTER 5: REFERENCES	212
BIBLIOGRAPHY (All Cited References)	233

# List of Tables

Table 3-1: Explanation of Funding Models for Canadian NPs

Table 3-2: Canadian Jurisdiction Dominant NP and Other Funding Models

Table 4-1: Policy Triangle Components

Table 4-2: Sustainability of Innovation Framework Factors

Table 4-3: Case Study Interview Questions

Table 4-4: A-priori Preliminary Codes

Table 4-5: Case Study Ten Themes Aligning with SIF and PTF

Table 4-6: Case Study New Emerging Themes

Table 4-7: Participant Characteristics

Table 4-8: Document Summary

Table 4-9: Additional Participant Identified Quotes Motivation Clinic Closing

Table 4-10: Additional Participant Identified Quotes Innovation Barriers

Table 5-1: Knowledge Dissemination

# **List of Figures**

- Figure 1-1: Policy Triangle Framework Diagram
- Figure 1-2: Sustainability of Innovation Framework Diagram
- Figure 2-1: Prisma Diagram, Search Results, Article Selection
- Figure 3-1: Sustainability of Innovation Framework
- Figure 3-2: Image Representation of NP Funding Models
- Figure 4-1: Policy Triangle Framework Diagram
- Figure 4-2: Sustainability of Innovation Framework Diagram
- Figure 4-3: Patient/Stakeholder Research Information Letter/Consent
- Figure 4-4: Sheep River NP Clinic Closure Timeline
- Figure 5-1: Primary Health Care Sustainability Assessment Framework

#### **Chapter One: Introduction**

The term sustainability is frequently mentioned as an important concept in all sectors of our health system, particularly in reference to sustainability of Primary Health Care (PHC).<sup>1</sup> Sustainability is best defined as "the appropriate balance between the cultural, social, and economic environments designed to meet the health care needs of individuals and populations. This leads to optimal health care outcomes without compromising the ability of future generations to meet their own health care needs".<sup>1(p.13)</sup> While there is a growing body of sustainability research, few studies have examined the challenges involved in sustaining innovations and which factors may ensure sustainability of innovations within PHC.<sup>2</sup>

A common sustainability challenge in Canada has been the Nurse Practitioner (NP) role, which was first introduced in the 1970's and re-implemented in the early 1990's in response to an expanded interest in PHC reform.<sup>3</sup> Several studies have examined the NP role and despite evidence of effectiveness and quality of care, barriers towards full optimization of the role and its sustainability remain at risk.<sup>3</sup> This thesis includes three scientific papers which describe the current state of knowledge in the topic areas of PHC innovation sustainability, Canadian NP funding models and examining sustainability failure in a case study exploring a primary care NP clinic closure in Alberta, Canada. Sustainability failures are influenced by variable provincial initiatives and policies, are particularly relevant in primary care and PHC reform, and the inability of PHC innovations to sustain themselves is an ongoing issue.

Evidence to better inform governments, funders and policy-makers on sustaining innovations in PHC is essential towards achieving a lasting vision of PHC reform. New knowledge, concepts and a conceptual framework to improve our understanding of sustainability of PHC innovations such as the NP role are presented in this thesis.

#### Background

#### **Primary Health Care Research Focus**

PHC research informs clinical practice, improves system performance and patient care, promotes critical thinking, encourages multidisciplinary collaboration and supports health.<sup>4</sup> PHC refers to a foundational approach and the provision of a broad range of services beyond Canada's traditionally funded medical and hospital services. PHC stresses the importance of health promotion, disease prevention and illness care and provides a framework for the delivery of coordinated and collaborative health services for individuals, families and communities.<sup>5,6</sup> Given that national and international bodies have increasingly stressed the importance of strong, comprehensive PHC as the foundation to an effective and efficient health system, it is important to engage in academic research that recognizes the importance of PHC principles and the policies of PHC renewal. This thesis uses a PHC lens and the term PHC is used to indicate a broader concept, a philosophy and model for improving health care.<sup>5,7</sup>

#### Sustainability of Canada's Primary Health Care System

Sustainability is the continued use of program components and activities for the sustained achievement of desirable program and population outcomes.<sup>8</sup> There are some nuanced differences among definitions of sustainability, but they all generally refer to the continued use of program components and activities beyond their initial funding period.<sup>8–10</sup> Despite long-term interest in sustainability of health care services, the study of sustainability has not developed into a widely utilized program of research and this understudied area has been identified as a significant translational research problem.<sup>9</sup>

Linked to sustainability is innovation, a new and novel idea that promotes a paradigm shift or practice change and has the potential to improve the quality of care and health service delivery. Expanding the scale up and spread of any innovation to achieve, sustainable, largescale health system transformation remains an understudied area of research. The term scale-up commonly refers to the process in which the coverage and impact of an innovation are expanded to reach all potential beneficiaries. The term spread aims to communicate and implement an innovation, and usually involves adapting an innovation to a new setting. While, interest is

2

growing within health systems, to focus research on all three; the spread, sustainability and scaleup of any innovation, the focus of this thesis is on the concept of sustainability.

The term sustainability lacks clarity and there is some confusion due to variations in terminology, a lack of agreed upon measurement frameworks and methodological challenges in measuring the concept.<sup>8</sup> Research on the concept of sustainability can require several layers of data collection. This layered approach is required to capture the multiple components impacting the continuation of a PHC innovation and until recently, the sustainability field lacked a common set of definitions, research questions, measures, and conceptual frameworks.<sup>8-11</sup> In light of these conceptual challenges, little is known about the extent to which PHC innovations are sustained or whether their impact on health is maintained after the initial implementation.<sup>9</sup>

Current Canadian health care reform agendas prioritize the provision and sustainability of cohesive, comprehensive PHC services that are responsive to community and patient needs.<sup>5,12</sup> Primary care is usually the first point of contact with the health system and unsustainability of such services contributes to diminished access to health care and poorer health outcomes.<sup>12</sup> Primary care is typically the setting through which patients experience the tenets of PHC. Limited studies have examined the sustainability of PHC innovations in this context and very little is known about long term effects of successful planning and implementation, or the relative elements of sustainability of these services.<sup>13-17</sup> Inadequate funding dedicated to the evaluation of PHC innovations after initial implementation, challenges to ongoing surveillance and lack of validated measures, have further complicated the study of sustainability.<sup>14,17</sup> Stirman et al. suggest that more robust and targeted research, as well as ongoing evaluation, is necessary to better inform sustainability of PHC innovations.<sup>17</sup> The following section describes one PHC innovation, the Nurse Practitioner.

#### Sustainability of the Nurse Practitioner Role in Canada

The Canadian government recognizes the need to address challenges faced by our health system, as well as the need for an innovative and effective approach to continue to provide sustainable health services.<sup>18</sup> Despite reform recommendations, a continued lack of accessibility

to care, long patient wait-times and gaps in health services continue and Canada has made limited progress towards achieving operationalization of PHC principles.<sup>18-21</sup> PHC reform in Canada has been characterized by numerous small-scale pilot projects, ineffective advocacy of system wide change, and a failure to embrace progressive incremental changes.<sup>19</sup> This approach to reform has resulted in the introduction and implementation of PHC innovations positioned at the periphery of the system rather than as part of its core. For example, a highly cited PHC reform failure has been the integration of NPs in the Canadian health system.<sup>3,18,21,22</sup> Our failure in understanding how to sustain the NP role and other innovations in PHC ultimately continues to impact patient access to health services, patient outcomes and prevents much needed reform.<sup>22-23</sup>

In Canada, NPs possess an extensive body of knowledge and skills, they diagnose acute and chronic illness, prescribe medications and are accountable and responsible for their own professional practice.<sup>20,23-25</sup> NPs were first introduced into the Canadian health system during the 1960's; however, it was not until the early 1970s, during a nationwide physician shortage that NPs were officially recognized.<sup>19,25</sup> NPs provide health services in numerous positions and care delivery models across Canada and all ten provinces and three territories have legislation and regulations set out to support the role.<sup>20,23-25</sup> Despite an increase in numbers of practicing NPs in Canada as well as mounting evidence recognizing the value that NPs contribute to PHC focused change, the majority of Canadians still have no access to NP services.<sup>19,25,26</sup>

Several factors have been recognized as contributing to Canada's inability to unleash NP optimization as a PHC innovation. These factors include a lack of role awareness/clarity; a lack of funding mechanisms to sustain the role, and marginalization of the role across the country.<sup>25-30,</sup> The contributing root cause to these challenges are obstacles associated with the laws and conditions set out by the Canada Health Act (CHA). The act, which has been in existence since 1984, sets out conditions the provinces and territories must meet to receive federal health dollars and defines the principles of health care across Canada.<sup>31</sup> When the act was introduced in 1984, physician and hospital care were set as priorities and the language within the act reflected this by including Physicians and Dentists as the only health care providers eligible for reimbursement through our national health system. After extensive lobbying by the Canadian Nurses

Association, the inclusion into the act of the term "Health Care Practitioner", meant that health care providers other than physicians and dentists could be reimbursed. Despite the importance of this act, it is only a financial arrangement and as such, Canadian health care policy, including decisions about what services will be universally provided, how those services will be funded and who will be permitted to deliver services, is determined exclusively by provincial governments.<sup>31</sup>

There is near-universal agreement that CHA legislation is too vague to provide direction in terms of provincial and territorial health system policy options and the act itself may be restricting PHC reform.<sup>31,32</sup> Consequently, provinces tend to take a 'risk-averse' policy approach, which continues to support the same models of health service delivery across Canada.<sup>31</sup> Numerous calls for policy reform to enhance PHC restructuring have emanated from studies and well researched reports in Canada.<sup>33</sup> Yet most reform proposals have not been acted on in a substantial way and entrenched and outdated health policies continue to endure.<sup>31</sup> Lack of comprehensive, adequate funding through ongoing government financial support is the single most important obstacle affecting NP role sustainability and is blocking the NP role from becoming a substantive part of Canada's PHC strategy.<sup>33</sup>

#### **Sustainability and Policy Failure**

The failure to optimize and sustain the NP role across Canada is an issue of great concern. In this thesis, sustainability failure is defined as the breakdown in health care services or provision of health care that has the potential to harm or the ability to impede quality health care.<sup>34</sup> Related to sustainability failure is the concept of optimization. This term addresses challenges in health care by focusing on improving care, accountability, financial and resource management and transparency in order to build a sustainable health care system. Together, the terms optimization and sustainability provide the health care system the opportunity to improve clinical outcomes for patients, advance economic efficiency and address health policy issues in the delivery of care.<sup>34</sup> The interconnected concepts of optimization and sustainability, are

necessary to explore but beyond the scope of this research which focuses specifically on sustainability failure.

To better understand how to improve policy in the Canadian health system to sustain PHC innovations like the NP role, it is important to understand sustainability and policy failure.<sup>34</sup> There is no shortage of explanations for why a policy fails in normal health policy-making circumstances.<sup>34</sup> Early studies in policy failure focus on the cognitive limits of policy-makers as a factor which biases decision-making.<sup>34</sup> Other studies point to more structural factors such as routinization or institutionalization blocking innovation by restricting consideration of innovative alternatives.<sup>34</sup> More recent studies have combined the above behavioural and structural motivations, finding evidence that decision-makers often simply ignore trends, evidence and developments; they are risk averse, like to avoid failure and prefer to maintain the status quo.<sup>34</sup>

Policy failure is a concept that is challenging to define and, in many cases, it is unclear whether the policy has rightly failed. Howlett<sup>33</sup> states that a policy failure is considered to have occurred simply because a stated policy initiative did not correct or resolve a policy problem. Yet defining and understanding policy failure is rarely as straightforward as this definition suggests. For the purpose of this thesis, policy failure is categorized as a lack of evidence informed, supportive policy; a lack of policy response or inaction which does not correct or resolve a health system issue.<sup>34</sup>

Our failure in understanding NP role sustainability, as well as the sustainability of other innovations in PHC impacts patient access to health care, overall patient health outcomes and prevents much needed reform in PHC. This thesis seeks to explore the topic of NP role sustainability by examining Canadian NP funding models and the closure of a NP clinic in Alberta, Canada. The findings from this research are intended to further inform governments, funders and policy-makers about key concepts associated with sustaining the NP role and other innovations in PHC. The theoretical and policy analysis frameworks guiding this thesis are introduced below.

#### **Theoretical Frameworks**

Two frameworks are utilized in this dissertation: 1) the Policy Triangle Framework (PTF) by Walt and Gilson<sup>36</sup> and the Sustainability of Innovation Framework (SIF) by Fox, Gardner and Osborne.<sup>35</sup> The PTF scrutinizes the policy environment; what policies support decision making, the context in which decisions are made and the participants involved in the decision-making process (Figure 1-1). The PTF is used to support and strengthen the design and analysis of this research.<sup>36</sup> The SIF is used as a theoretical base for this thesis and examines the factors of sustainability (Figure 1-2). While this framework enhances understanding of the sustainability process, it was developed recently, has not been adequately tested, and does not comprehensively capture the environment where policy decisions are made

#### **Policy Triangle Framework**

The PTF, developed by Walt and Gilson, and analyzes the social, economic and political factors, which have influenced a policy-making decision. Walt and Gilson believe that health policy-making is an interactive process within special social-economic and cultural contexts where actors - individuals, groups and organizations - are at the center of this process. This framework stresses the importance of the process by which the policy was initiated, formulated, developed, implemented and evaluated, including the objectives of the policy, and the 'actors' involved in the decision-making process.<sup>35</sup> This simplified framework and approach to policy analysis assists researchers to understand and analyze health-related policies systematically.

#### Sustainability Innovation Framework

The SIF, merges the concepts presented by Greenhalgh and colleagues' systematic review of health service innovation,<sup>37</sup> and the Dynamic Sustainability Framework of Chambers, Glasgow and Stange.<sup>38</sup> This theoretical framework is suitable for the examination of sustainability of any PHC innovation and identifies five factors that impact sustainability. These factors are: political, organizational, financial, workforce and innovation specific and each factor has clear-cut questions and criteria used to assess sustainability of an innovation in PHC.<sup>35</sup>

7

#### **Dissertation Objectives**

This thesis was planned and is presented in manuscript format. Three manuscripts are provided in various stages from *accepted for publication (in press)* to *in preparation*. The objectives of this dissertation are the following:

### Study #1: Scoping Review of the Literature

Identify and synthesize peer-reviewed and grey literature in order to understand the factors that influence clinic closures involving NPs in PHC, as well as the factors that influence the sustainability of PHC innovations such as the NP role.

#### Study # 2: Critical Analysis of NP Funding Models in Canada

Critically analyze NP funding models across Canada utilizing a theoretical framework to understand the current state of these models and provide future directions for governments, funders and policy-makers on achieving funding sustainability of the NP role.

## Study # 3: Case Study Examining the Closure of a NP Clinic

Understand the contextual factors, the actors influencing, and the decision-making process of a NP primary care clinic closure in Alberta, Canada. A qualitative approach was used to obtain a comprehensive picture of a case to exemplify the sustainability failure of a PHC innovation. The overall aim of this work was to contribute to theory, knowledge and health policy development regarding sustaining PHC innovations including the NP role.

#### Methods

#### Study #1

The method used in Study #1 was a scoping review. In Study #1, 4 electronic databases were searched for peer-reviewed and grey literature related to three research questions. Our

search yielded 2550 articles. We included 24 peer-reviewed and grey literature articles in our final results that matched our review criteria.

#### Study #2

The method used in Study #2 was a critical analysis utilizing a theoretical framework. We provided an overview on the current funding models for NPs across all jurisdictions in Canada. This information was analyzed with the Sustainability of Innovations Framework by Fox, Gardner & Osborne.<sup>35</sup>

## Study #3

The method used in Study # 3 was a single, exploratory case study. Data collected was in the form of semi structured interviews, documents and field notes. We conducted 18 interviews and collected 31 documents for analysis. Participants included health organization leaders, professional organization leaders, health care professionals and patients. Two frameworks; the Sustainability of Innovation Framework and Policy Triangle Framework were utilized to inform Yin's case study analysis approach as the study design.<sup>35, 36, 39</sup>

#### **Researcher Positionality**

An important characteristic of qualitative research is the researcher and the role they play in the research process. A researcher must acknowledge they are a human instrument and the primary research tool.<sup>40</sup> In the interest of full disclosure, this researcher acknowledges that their professional experience as a NP may bias the methodological approach and analysis. Therefore, utilizing a reflexive method, specifically, a reflexive journal, where the researcher logs the details of how he or she may have influenced each portion of the research design, was part of the study design. This approach to reflexivity sensitized the researcher to her prejudices while fully informing the research team of the impact of these influences on the credibility of the research outcomes.<sup>40</sup>

# Ethics

This study was conducted with full compliance of research ethics, codes and practices established by the University of Alberta, Human Research Ethics Board. Ethical approval to conduct the study in Chapter 4 was received from the University of Alberta Research Ethics Board (Pro00087461).

## Conclusion

In this chapter, the background, objectives, and rationale were provided. Subsequent chapters (2-4) contain the manuscripts that represent the outputs of this thesis. Each manuscript reflects the linked studies described in this chapter. The final chapter of this dissertation contains a summary of the findings, main conclusions drawn from each study, main limitations, contributions this research makes to knowledge, theory, practice and policy, and the next steps in the form of a knowledge translation plan for dissemination of the findings.

**Chapter 1 References** 

- Stonebridge C, Brichta J, Dinh T. A Road Map to Health System Sustainability: CASHC Compendium Report, 2011–16. The Conference Board of Canada - Economic forecasts, public policy, and organizational performance. <u>https://www.conferenceboard.ca/e-library/abstract.aspx?did=8816&AspxAutoDetectCookieSupport=1</u>. Published 15 May 2017. Accessed February 13, 2020
- Martin GP, Weaver S, Currie G, Finn R, Mcdonald R. Innovation sustainability in challenging health-care contexts: embedding clinically led change in routine practice. *Health Serv Manage Res.* 2012;25(4):190-9. doi: 10.1177/0951484812474246
- Sangster-Gormley E, Martin-Misener R, Burge F. A case study of nurse practitioner role implementation in primary care: what happens when new roles are introduced? *BMC Nursing*. 2013;12(1). doi:10.1186/1472-6955-12-1.
- Brown LJ, Mcintyre EL. The contribution of Primary Health Care Research, Evaluation and Development-supported research to primary health care policy and practice. *Aust J Prim Health.* 2014;20(1):47-55. doi:10.1071/py12058
- Canadian Nurses Association. Primary Health Care The Time Has Come. https://www.cna-aiic.ca/~/media/cna/page-content/pdffr/primaryhealthcare sept 2003 e.pdf?la=en. Published September 2003
- World Health Organization. International Conference on Primary Health Care (1978 : Alma-Ata, USSR), World Health Organization & United Nations Children's Fund (UNICEF). (1978). Primary health care: report of the International Conference on Primary Health Care, Alma-Ata, USSR, 6-12 September 1978 / jointly sponsored by the

World Health Organization and the United Nations Children's Fund. World Health Organization. <u>https://apps.who.int/iris/handle/10665/39228</u>

- Canadian Nurses Association. Primary care, Primary Health Care. Canadian Nurse. <u>https://canadian-nurse.com/articles/issues/2015/june-2015/cna-now?page=8</u>
- Scheirer MA, Dearing JW. An Agenda for Research on the Sustainability of Public Health Programs. *American Journal of Public Health*. 2011;101(11):2059-2067. doi:10.2105/ajph.2011.300193
- Shelton RC, Cooper BR, Stirman SW. The Sustainability of Evidence-Based Interventions and Practices in Public Health and Health Care. *Annual Review of Public Health*. 2018;39(1):55-76. doi:10.1146/annurev-publhealth-040617-014731
- Walugembe DR, Sibbald S, Ber MJL, Kothari A. Sustainability of public health interventions: where are the gaps? *Health Research Policy and Systems*. 2019;17(1). doi:10.1186/s12961-018-0405-y
- Fleiszer AR, Semenic SE, Ritchie JA, Richer M-C, Denis J-L. The sustainability of healthcare innovations: a concept analysis. *Journal of Advanced Nursing*. 2015;71(7):1484-1498. doi:10.1111/jan.12633
- 12. Buykx P, Humphreys JS, Tham R, et al. How do small rural primary health care services sustain themselves in a constantly changing health system environment? *BMC Health Services Research*. 2012;12(1). doi:10.1186/1472-6963-12-81
- Lennox L, Maher L, Reed J. Navigating the sustainability landscape: a systematic review of sustainability approaches in healthcare. *Implementation Science*. 2018;13(1). doi:10.1186/s13012-017-0707-4

- 14. Sibthorpe BM, Glasgow NJ, Wells RW. Emergent themes in the sustainability of primary health care innovation. *Medical Journal of Australia*. 2005;183(S10). doi:10.5694/j.1326-5377.2005.tb07185.x
- 15. Wakerman J. Innovative rural and remote primary health care models: What do we know and what are the research priorities? *Australian Journal of Rural Health*. 2009;17(1):21-26. doi:10.1111/j.1440-1584.2008.01032.x
- 16. Wakerman J, Humphreys JS. Sustainable primary health care services in rural and remote areas: Innovation and evidence. *Australian Journal of Rural Health*. 2011;19(3):118-124. doi:10.1111/j.1440-1584.2010.01180.x
- 17. Stirman SW, Kimberly J, Cook N, Calloway A, Castro F, Charns M. The sustainability of new programs and innovations: a review of the empirical literature and recommendations for future research. *Implementation Science*. 2012;7(1). doi:10.1186/1748-5908-7-17
- Hutchison B, Levesque JF, Strumpf E, Coyle N. Primary health care in Canada: systems in motion. *Milbank Q.* 2011;89(2):256-88. doi: <u>10.1111/j.1468-0009.2011.00628.x</u>
- 19. Canadian Nurses Association. (2019). Advanced practice nursing: a pan-Canadian framework. Ottawa: Canadian Nurses Association. <u>https://www.cna-aiic.ca/-/media/cna/page-content/pdf-en/apn-a-pan-canadian-framework.pdf</u>
- 20. Donald F, Martin-Misener R, Bryant-Lukosius D, et al. The Primary Healthcare Nurse Practitioner Role in Canada. *Nursing Leadership*. 2010;23(sp):88-113. doi:10.12927/cjnl.2013.22271
- 21. El-Jardali F, Lavis JN. Issue Brief: Addressing the Integration of Nurse Practitioners in Primary Healthcare Settings in Canada. Hamilton, Canada: McMaster Health Forum, 6

July 2011. https://www.mcmasterforum.org/docs/default-source/productdocuments/issue-briefs/nurse-practitioners-in-primary-healthcare-in-canadaib.pdf?sfvrsn=2

- 22. Dicenso A, Bourgeault I, Abelson J, et al. Utilization of Nurse Practitioners to Increase Patient Access to Primary Healthcare in Canada – Thinking Outside the Box. *Nursing Leadership.* 2010;23(sp):239-259. doi:10.12927/cjnl.2010.22281
- 23. Kaasalainen S, Martin-misener R, Kilpatrick K, et al. A historical overview of the development of advanced practice nursing roles in Canada. *Nurs Leadersh* (Tor Ont).
  2010;23 Spec No 2010:35-60. DOI: 10.12927/cjnl.2010.22268
- 24. Dicenso A, Auffrey L, Bryant-Lukosius D, et al. Primary health care nurse practitioners in Canada. *Contemporary Nurse*. 2007;26(1):104-115. doi:10.5172/conu.2007.26.1.104
- 25. Edwards N, Rowan M, Marck P, Grinspun D. Understanding Whole Systems Change in Health Care: The Case of Nurse Practitioners in Canada. *Policy, Politics, & Nursing Practice.* 2011;12(1):4-17. doi:10.1177/1527154411403816
- 26. Mattison CA, Wilson MG. Rapid Synthesis: Enhancing Health System Integration of Nurse Practitioners in Ontario.
- 27. Ashby M. (2006). Barriers to nurse practitioner reimbursement. *The Pennsylvania Nurse*, 61(2), 19.
- 28. Brassard A. Scope of practice wins in Iowa, Nevada, Oregon and Rhode Island. *The American Nurse*, 2013:11.

- 29. Gould ON, Johnstone D, Wasylkiw L. Nurse practitioners in Canada: Beginnings, benefits, and barriers. *Journal of the American Academy of Nurse Practitioners*. 2007;19(4):165-171. doi:10.1111/j.1745-7599.2007.00210.x
- 30. Canadian Nurses Association. The Primary Health Care Approach. <u>https://www.cna-aiic.ca/-/media/cna/page-content/pdf-en/fs02\_primary\_health\_care\_approach\_june\_2000\_e.pdf?la=en&hash=C31F23CD2036</u>6955F267C04D737B32163F7D5623
- 31. Esmail N, Barua B. Is Canada's Health Act a Barrier to Reform. Fraser Institute. https://www.fraserinstitute.org/studies/is-the-canada-health-act-a-barrier-to-reform
- 32. Martin-Misener R, Bryant-lukosius D, Harbman P, et al. Education of advanced practice nurses in Canada. *Nurs Leadersh* (Tor Ont). 2010;23 Spec No 2010:61-84. DOI:<u>10.12927/cjnl.2010.22269</u>
- 33. Tapper L, Trevoy J. Key Issues Impacting Nurse Practitioner Practice and Integration in Alberta. Nurse Practitioner Association of Alberta. <u>https://albertanps.com/wpcontent/uploads/2015/03/Key-Issues-for-Nurse-Practitioners-of-Alberta.pdf</u>
- 34. Howlett M. The lessons of failure: learning and blame avoidance in public policymaking. *International Political Science Review*. 2012;33(5):539-555. doi:10.1177/0192512112453603
- 35. Fox A, Gardner G, Osborne S. A theoretical framework to support research of health service innovation. *Aust Health Rev.* 2015;39(1):70-75. doi:10.1071/ah14031
- 36. Walt G, Gilson L. Reforming the health sector in developing countries: the central role of policy analysis. *Health Policy Plan*. 1994;9(4):353-70. doi:10.1093/heapol/9.4.353

- 37. Greenhalgh T, Robert G, Bate P, Kyriakidou O, Macfarlane F, Peacock R. How to spread good ideas - A systematic review of the literature on diffusion, dissemination and sustainability of innovations in health service delivery and organisation. *Report for the National Co-ordinating centre for NHS Service Delivery and Organisation R & D* (NCCSDO), 1-426
- 38. Chambers DA, Glasgow RE, Stange KC. The dynamic sustainability framework: addressing the paradox of sustainment amid ongoing change. *Implementation Science*. 2013;8(1). doi:10.1186/1748-5908-8-117
- 39. Yin RK. Case study research: Design and methods (5th ed.). Thousand Oaks, CA: Sage.
- 40. Bourke B. Positionality: Reflecting on the research process. *Qualitative Report*, 2015: 19(33).

Figure 1-1: Policy Triangle Framework<sup>35</sup>





# Chapter Two: Sustaining Primary Health Care Programs and Services: A Scoping Review (Paper 1)

20

Raelene Marceau Dr. Kathleen Hunter Dr. Stephanie Montesanti Dr. Tammy O' Rourke

Marceau, R., Hunter, K., Montesanti, S. and O' Rourke, T., 2020. Sustaining Primary Health Care Programs and Services: A Scoping Review. *Policy, Politics and Nursing Practice,* In Press.

# Sustaining Primary Health Care Programs and Services: A Scoping Review Informing the Nurse Practitioner Role in Canada

Sustainability is an important concept in health care delivery, in particular when examining the principles of PHC and health care reform. Efforts aimed at restructuring Canada's health system have led to the introduction and implementation of numerous PHC programs and services, however, many of these are not sustained.<sup>1</sup> An example is the NP role, which was introduced in Canada in 1967 as a means of addressing the primary care (PC) physician shortage.<sup>2</sup> Utilization of the NP role has fluctuated over the past 40 years and although NPs are acknowledged as an important addition to the health care system, sustainability of the professional role remains at risk.<sup>3</sup> A scoping review was undertaken as part of a research project to examine the closure of an NP clinic in Alberta, Canada with the aim to explore evidence in three main areas: a) the factors which influence decisions to close clinics that include NPs; b) factors which influence the sustainability of PHC programs and services and; c) factors which influence the sustainability of the NP role in PHC. This review summarizes the state of the literature on sustainability of the NP role and other PHC programs and services and identifies existing gaps in knowledge to provide direction for future research.

# Background

#### **Primary Care and Primary Health Care**

NPs were first introduced into Canada's health system to increase access to PC and to utilize their expertise in PHC service delivery.<sup>4</sup> PC and PHC are terms often used interchangeably, yet denote distinct entities and concepts.<sup>5</sup> For the purpose of this review, PC is defined as "as the provision of integrated, accessible health care services by clinicians who are accountable for addressing a large majority of personal health care needs, developing a sustained partnership with patients, and practicing in the context of family and community".<sup>6</sup> In contrast, PHC is defined as essential health care with basic elements and objectives that help attain better

health services for all, is universally accessible to any individual at a cost the community can afford, and includes all areas that interact with health; such as social, cultural, environment, and lifestyle factors.<sup>5,7,8</sup> This scoping review will encompass a PHC focus on sustainability but will include PC as a necessary element to strengthen and enhance the exploration of all relevant literature.<sup>7</sup> To better inform sustainability of the NP role and other PHC programs and services in Canada, literature from Canada, United Kingdom and Australia were included in this review as these countries all have some form of publicly funded health care.

## Sustainability in PHC

Sustainability, in the broadest sense, has been defined as the ability to maintain, uphold or support something at a certain level for a long period of time.<sup>1</sup> A comprehensive definition consolidates new and emerging constructs to advance a definition in which sustainability occurs after a defined period of time, when a program or implementation strategy continues to be delivered, involves individuals or organizations whose behaviour is maintained, and may evolve or adapt while continuing to produce benefits for individuals and systems.<sup>9</sup>

Discussion about program and services sustainability often arises within Canada's health care system in reference to PHC reform and understanding how to address sustainability issues is "one of the most significant translational research problems of our time"<sup>9 (p.1)</sup>. However, not all programs and services are sustainable, even with excellent research and evidence. Efforts aimed at restructuring Canada's health system have led to the introduction and implementation of new PHC programs and services, for example, utilization of the NP role. However, little is known about whether, or for how long, many of these programs and services are sustained.<sup>1</sup> Coupled with a rapidly changing health policy environment, the lack of coordinated effort to study PHC programs and services, contributes to a limited understanding of sustainability and seriously compromises Canada's ability to achieve meaningful PHC reform.<sup>1</sup>

## Primary Health Care Reform and the NP Role

NPs are essential health professionals with advanced education, such as a Master's or Doctoral degree.<sup>10,11</sup> In Canada, NPs possess a wide range of knowledge and skills, diagnose acute and chronic illness, prescribe medications, and are accountable and responsible for their own professional practice.<sup>10,11</sup> In the mid-1990s to early 2000s, amidst the call for major PHC reform, the use of NPs was identified as a means to improving access to cost-effective, highquality health care. It was during this time that all Canadian jurisdictions launched a variety of new PHC-focused programs and services that included NPs and each jurisdiction developed their own legislation and regulation for the NP role.<sup>3,4,10,11</sup> In 2010, there were 2,486 NPs practicing in Canada, up from 1,129 in 2006, and in 2018, there were over 5,000 NPs across the country.<sup>12</sup>

Despite this growth, a number of obstacles continue to challenge the sustainability of NPs in Canada.<sup>3,10,11</sup> These obstacles include an absence of policy to support sustainable funding, a lack of role awareness by patients, providers, and policymakers; marginalization of the role by corporations, physician associations and government organizations; and a lack of standardization of the role across all jurisdictions.<sup>3,13,14</sup> PHC programs and services that include NPs have also been erratically introduced and in most jurisdictions the establishment of the role has occurred in a 'piecemeal' fashion with no overarching strategy or long-term funding solution in sight. <sup>3,4,11,14</sup> A comprehensive review of the literature on the factors supporting or adversely influencing NP role sustainability is needed, has the potential to further inform governments and policymakers about PHC programs and services and could stimulate action and change towards health system transformation.

#### Methods

Arksey and O'Malley's (2005) scoping review method was chosen to guide this review, map existing literature and identify research gaps.<sup>15</sup> This method enables examination of all applicable literature regardless of study design. There are five stages in the methodological framework: (a) identify the research question(s), (b) search for relevant studies, (c) selection of relevant studies, d) analysis including charting, collecting and summarizing the data, and e) reporting the results .<sup>15</sup> This five-stage format adopts a rigorous process of transparency, enables replication of the search strategy and increases the reliability of the study findings .<sup>15</sup> In addition to peer reviewed articles, we elected to include news articles and media items as part of this review to deepen our understanding on the scope of NP clinic closures that may not have been captured by research studies.

#### **Stage 1: Identify Research Questions**

For the purpose of this review, three research questions were formulated to guide the review. What is known about the factors, which influence the closure of clinics specifically using NPs? What is known about factors, which support and/or adversely influence the sustainability of PHC programs and services? What is known about factors, which support and/or adversely influence the sustainability of the NP role in PHC?

#### **Stage 2: Identifying Relevant Studies**

The key terms *clinic, closure, nurse practitioner, primary health care, primary care and sustainability,* were expanded to include *Canada, United Kingdom and Australia* as it was felt these countries had a parallel experience with implementation of the NP role in PHC. Since PHC is a term not routinely utilized in the United States (U.S.) as defined in this review, it was not included as a key term. The key terms and synonymous words were then used to guide the search strategy. Databases MEDLINE, CINAHL, ACADEMIC SEARCH COMPLETE and HEALTH POLICY REFERENCE CENTRE were chosen for comprehensiveness. A search of the grey literature was conducted over the same period using the databases Canadian Newsstream, News and Newspapers, News, Policy and Politics Magazine, and Google/Google Scholar search engines.

#### **Inclusion and Exclusion Criteria**

Studies, newspaper articles and media items were eligible for inclusion if they addressed the chosen key terms. Both PHC and PC were utilized as key terms as they are often used interchangeably within the health literature. A publication date limit of 1990 to 2019 was set for the database and grey literature search to obtain the most up to date evidence and to assist with further narrowing of the search results .<sup>15</sup> English only limits were set to represent key term countries with similar NP experiences. Studies were excluded if they were outside limit dates, did not contain key terms, were non-English and focused on: hospital closure, private clinic or abortion clinic closure and clinic closure due to physician shortage. Studies using the terms environmental/ecological sustainability, sustainable development (or development goals) and use of any term or designation for the NP role other than nurse practitioner, were also excluded (Inclusion/Exclusion Table 2-1). Relevant peer reviewed studies of any study design were sought, as well as any applicable editorial, letter, policy brief, newspaper article, interview or media item from the grey literature search. The purpose of searching the grey literature, defined as research that is either unpublished or has been published in non-commercial form, was to minimize bias, to create a broad evidence base, and to integrate a complementary perspective utilizing mainstream and peer reviewed publications.<sup>15</sup>

#### **Stage 3: Selection of Studies**

A total of 2,550 citations (duplicates excluded) were retrieved, with one author (RM) initially screening by title. Non-relevant items were removed, leaving a total of 215 potential relevant items. Subsequently, these items were screened by two authors (RM, KH), first by abstract (if available), yielding 76 articles after non-relevant items were removed. All applicable articles were then given full-text review by two authors (RM, KH). Each full-text article was evaluated for relevance according to the inclusion and exclusion criteria. Reference lists of included studies were hand searched, with two additional relevant peer reviewed studies identified. A total of 24 articles (2 primary studies, 2 systematic reviews, 20 news items) met the inclusion criteria and were included in the review (See Figure 2-1). As our intent was to map

both the research and grey literature to inform our understanding and answer our questions, we did not undertake a quality appraisal.

#### **Stage 4: Charting the Data**

Summary tables were created to chart the data for both peer reviewed and grey literature sources, and data from the sources was extracted to the tables. The tables were analyzed for commonalities addressed in the sources, and grouped into three main topic areas.

## Results

This scoping review explored the existing literature concerning three research questions. The results provide a description of the evidence from the three research questions and are divided into results for NP clinic closures, sustainability of PHC programs and services and sustainability of the NP role in PHC.

#### **NP Clinic Closures**

There is a lack of published, peer reviewed evidence to better understand clinic closures specifically using NP services, despite numerous newspaper articles and one media source detailing clinic closures in the grey literature (n=20). <sup>16–35</sup> (See Table 2-2). These sources include newspaper articles and a radio interview transcript, and are described in further detail below. Most clinics that included NPs were recorded as closed due to lack of funding (n=15),<sup>16–30</sup> and/or funding termination (n=1).<sup>31</sup> Other reasons cited for closure include patient underutilization (n=1),<sup>32</sup> and clinic mismanagement (n=1).<sup>33</sup> In two of the sources, the rationale for clinic closure was indiscernible.<sup>34,35</sup>
## **Sustainability of PHC Programs and Services**

There is limited peer reviewed literature pertaining to sustainability of PHC programs and services and the existing evidence is poorly developed. A total of five sources pertaining to the sustainability of PHC programs and services were included, all were peer-reviewed articles. <sup>36–39</sup> (See Table 2-3). Of the four, two were systematic reviews.<sup>37,38</sup> The other two include a longitudinal evaluation of PHC service sustainability,<sup>36</sup> and a discussion paper outlining a systems-based solution to PHC sustainability.<sup>39</sup>

Buykx et al<sup>36</sup> used a mixed method, longitudinal design to explore how one PHC service proactively managed change to achieve sustainability. These authors identified six factors influencing sustainability; alignment with government policies, assessment of service performance, maintaining engagement with the local community, maximizing opportunities for alternative funding, initiating workforce succession planning, and actively partnering in research and evaluation designed to ensure quality improvement.<sup>36</sup> Sustainability success did not occur as the result of any one key success factor, but rather through proactively seizing opportunities to strengthen the service and implementing practical responses to risks and threats.<sup>36</sup>

Gruen et al<sup>37</sup> described the degree to which PHC program sustainability is dependent on system connectivity and alignment.<sup>37</sup> Unsustainable PHC programs were the result of an imbalance in the system, represented by misalignment/disconnection between population health, programming and the program administrators. Well-designed PHC programs are vulnerable to withdrawal of funding or community support if stakeholder facilitation is neglected, however, PHC programs that recognize and respond to the needs of key stakeholders at the cost of commitment to health improvement will likely to run out of support. Ultimately, a comprehensive approach to PHC program sustainability is based on the availability of research that would benefit policymakers, funders, and program managers.<sup>37</sup>

Lennox et al<sup>38</sup> acknowledge that the sustainability research 'landscape' is fraught with frameworks, models and tools to support and monitor sustainability, yet there is little direction to guide this area of research. This systematic review aimed to identify approaches available to assess and influence sustainability in health systems and to describe the different perspectives,

applications, and constructs within these approaches to guide their future use. Study findings revealed that selecting a sustainability method poses a challenge because of the diverse approaches reported in the literature. Based on these findings, a consolidated framework for sustainability in health care was developed as a resource and tool for researchers, health care professionals and improvement practitioners.<sup>38</sup>

Wakerman and Humphreys <sup>39</sup> also suggest that more research and evaluation is necessary to better inform sustainability of PHC services.<sup>39</sup> From their synthesis of evidence on remote and rural PHC services in Australia, a system-based solution to PHC sustainability was suggested as necessary. Achieving sustainability requires a proactive approach including alignment of health organizations; collaboration amongst health care providers, researchers and consumers; and strong association with the external policy environment.<sup>39</sup>

#### **Sustainability of NPs**

Despite a thorough assessment, no peer-reviewed or grey literature sources were selected pertaining to sustainability of the NP role in PHC and there is no evidence to improve understanding on sustainability of NPs in PHC.

## Discussion

The key findings of this review include: (a) a lack of published, peer reviewed research to advance the understanding of clinic closures specifically using NPs, despite numerous newspaper articles and one media source detailing clinic closures in the grey literature, (b) limited evidence from peer reviewed research pertaining to sustainability of PHC programs and services, and (c) a lack of peer reviewed, published research informing sustainability of the NP role in PHC is identified.

The subsequent discussion summarizes the results to provide a comprehensive overview of what is known about the state of the literature regarding clinic closures specifically using NP services and sustainability of PHC programs and services, including the NP role. It also draws attention to prominent gaps in the literature with consideration given to the implications and limitations of this review.

## **Knowledge Gaps**

Although a significant number of sources were assessed in this scoping review, the vast majority of selected sources were from the grey literature. As a result, this review revealed several knowledge gaps. Given the number of clinic closures detailed in popular media sources, as well as the dearth of applicable peer reviewed studies, a lack of rigorous research to broaden understanding on clinic closures is apparent. Multiple news items indicate clinics with and without NPs close frequently. However, there are no peer reviewed published studies that addressed the closure of a clinic specifically using NP services or the closure of clinics utilizing NPs. The lack of substantial evidence on clinic closures suggests that understanding the factors which influence these closures has yet to be explored and such research has potential to support PHC advancement and a more evidenced-informed health policy process within Canada.<sup>1</sup>

There was no literature which articulated a comprehensive, well-defined sustainability definition applicable to PHC and it is recognized the lack of standard, comprehensive sustainability definitions and the use of a wide variety of synonyms, is a significant issue in this area of study.<sup>9</sup> The discrepancy in the use of the term sustainability results in ambiguous understanding of the word, impacts the health policy process and impairs the ability of organization, policymakers, practitioners and researchers to understand how to support programs and services in PHC reform.<sup>9</sup>

This review further confirms a gap in research specific to sustainability of PHC programs and services and the literature on this topic is "fragmented and underdeveloped.<sup>40</sup> There is also scant information on the unintended consequences of sustainability in PHC, which priorities are favored, why some services are sustained and how long a service must remain to be considered sustained.<sup>38–42</sup> Limited funding for examining PHC programs and services after initial implementation, challenges to real-time surveillance and lack of validated measures, make the study of sustainability in PHC complex and difficult.<sup>40,41</sup>

The literature addressing sustainability of the NP role is limited, not explicit to PHC and focuses on discussing the prevailing barriers or strategies to enhance sustainability of the NP role. It does not advance a definition of sustainability that is inclusive and aligned with the NP role in PHC and does not discuss sustainability of the NP role within the context of health policy or PHC reform. The literature also demonstrates a lack of prospective, theoretical and evaluative research on NP role sustainability in PHC, and the paucity of research examining sustainability of the NP role in PHC raises "important questions about this little explored field". <sup>43-45</sup> Consequently, the above strategies require further examination and evaluation to advance NP role sustainability within Canadian PHC reform. <sup>44</sup>

#### **Implications for Future Research**

Further research is required on clinic closures specifically using NP services. This area of study could contribute to informing and improving the sustainability of the NP role by examining the reasons for closure, the implications of closures on the health of communities and patients, and the impact on health care professionals and organizations. <sup>36–41</sup> It is equally crucial for organizations, policymakers, and funders to engage in the knowledge generation process when a clinic closes. As evident in the grey literature, clinics close frequently, yet all too often there is no review, evaluation or investigation conducted post closure. The generation of this type of post clinic closure knowledge could have a significant and valuable impact on the Canadian health policy process, could assist with advocacy for NP role sustainability and enhance PHC reform.<sup>36,37,39,41</sup>

With no clear consensus on how to define or influence sustainability and little overarching direction for research, it is clear there is also a substantial need for research specifically focused on sustainability of PHC programs and services.<sup>38</sup> This research should emphasize a definition of sustainability that builds on a standard explanation explicit to PHC and addresses the structures and challenges of this setting.<sup>9,40</sup> Researchers must use frameworks, tools and models which advance the sustainability of PHC programs and services, are the most appropriate for their projects, include multi-level measurements of sustainability and allow for greater methodological

rigor and interpretability of findings.<sup>40</sup> Work in this area of research is evolving, and sustainability frameworks and models exist within the literature; however, there is a risk in continually creating 'novel' approaches with similar constructs divided by semantics and individual interpretations. <sup>37,38,44,46–48</sup> Thus, researchers should take in account the literature and available frameworks/models before 'reinventing the wheel'. Continuous production of new frameworks/models may lead to further division and confusion in the literature resulting in fewer robust studies on sustainability in PHC.<sup>38</sup>

An important opportunity exists to examine clinic closures and the concept of sustainability together, where one could inform and impact understanding of the other. Studies examining sustainability in health systems and PHC are an emerging science and there is no peer-reviewed literature on clinic closures. The intersection of this research may have several benefits; first, most studies have measured sustainability at a single point in time and masked a potentially dynamic phenomenon<sup>38,40</sup> Studying why a clinic has closed and assessing the concept of sustainability over several years rather than at a single time point can capture important information and further develop the PHC sustainability field. Second, combining the two areas of research also has the potential to influence health care organizations, practitioners, policymakers and funders in decision making about future PHC programs and services and their sustainability.<sup>38,40,42</sup>

Given the paucity of published research on sustainability of the NP role in PHC it is important to correlate research findings. A deeper exploration and examination of sustainability of the NP role happens through connection and correlation with other PHC sustainability studies, further enhancing and informing the sustainability of each. Sustainability of the NP role in PHC is complex and is best understood through the use of an evaluative, theoretical framework or conceptual model to structure research questions; more operational definitions with clear measurement guidelines for sustainability; and more robust designs and rigorous methods for testing sustainability concepts. This is an important future step and is crucial towards building a solid evidence base for organizations, policymakers and funders to further support NP role sustainability.<sup>49</sup> The topic of NP role and PHC programs and services sustainability is at an early stage and more robust research informed by theory, is required to continue to expand knowledge in this field.<sup>49</sup> For too long, sustainability has been a concern only at the "tail end" of the implementation process and thus, is insufficiently addressed.<sup>49(p12)</sup> Continued efforts are needed to bridge the work in the PHC sustainability field, consolidate what is known, identify unanswered research questions, and formulate a plan for accelerating an evidence base to sustain PHC focused programs and services such as the NP role.<sup>49</sup>

## Strengths and Limitations of the Review

This review provides improved understanding of the paucity of research on the topic of sustainability of the NP role and other PHC programs and services. A comprehensive approach to this review was taken and includes utilizing two full article reviewers, developing a search strategy in consultation with a librarian, and searching for broad literature from an inclusive group of databases. This approach added rigour to the scoping process and as such, serves as its strength. Despite rigour in the scoping process certain limitations are evident. Scoping reviews offer a distinct opportunity to retrieve and scan a broad range of literature to answer research questions. One potential limitation is that by searching the titles, abstracts, and subject headings only, we may have missed relevant papers. The second limitation is the challenge of comparing literature across countries and health systems. Canada, the UK and Australia all have parallel NP experiences in PHC. The U.S. was purposefully left out of the search due to its divergent health system and focus on PC. Including the key term U.S. in the review would have yielded more information and results but is not as reflective of the NP role in PHC as the other selected countries. The third limitation is that only English-language sources were retrieved and reviewed. This was due to the English-speaking nature of the countries selected as part of the inclusion criteria for the review, as well, the limited resources for translation services. Lastly, while the processes described in this paper are sufficiently transparent to allow the research to be replicated, the quality of the articles examined was not evaluated as per the procedure of a scoping review.<sup>15</sup>

#### Conclusion

The paucity of literature on clinic closures, and the literature on sustainability of the NP role and PHC programs and services, is in its nascent stage. This review finding emphasizes the need for more robust examination to improve our understanding of the contextual factors which influence clinic closures that involve NP services and thus impede the sustainability of PHC programs and services led by NPs. The pattern of PHC initiatives being generated, implemented and often un-sustained continues at great cost to the health care system.<sup>1,48</sup> It produces an even greater cost to patients, whose access to care and overall health outcomes can be significantly affected. It is important to recognize that these topic areas, when better informed through theoretically driven research, have great capacity to support and advance necessary health system reform.

#### **Chapter Two References**

- Hovlid E, Bukve O, Haug K, Aslaksen AB, Von plessen C. Sustainability of healthcare improvement: what can we learn from learning theory?. *BMC Health Serv Res.* 2012;12:235. <u>https://doi.org/10.1186/1472-6963-12-235</u>
- Dicenso A, Auffrey L, Bryant-Lukosius D, et al. Primary health care nurse practitioners in Canada. *Contemporary Nurse*. 2007;26(1):104-115. doi:10.5172/conu.2007.26.1.104

#### https://doi.org/10.5172/conu.2007.26.1.104

- Donald F, Martin-Misener R, Bryant-Lukosius D, et al. The Primary Healthcare Nurse Practitioner Role in Canada. *Nursing Leadership*. 2010;23(sp):88-113. doi:10.12927/cjnl.2013.22271
- Sangster-Gormley E, Martin-Misener R, Burge F. A case study of nurse practitioner role implementation in primary care: what happens when new roles are introduced? *BMC Nursing*. 2013;12(1). doi:10.1186/1472-6955-12-1
- Muldoon, L. K., Hogg, W. E., & Levitt, M. (2006). Primary care (PC) and primary health care (PHC): what is the difference? *Canadian Journal of Public Health/Revue Canadienne de Sante'e Publique*, 409–411. https://www.jstor.org/stable/41995754
- Starfield, B. (1998). Primary care: balancing health needs, services, and technology. Oxford University Press.

- Canadian Nurses Association. (2015). Primary care, Primary Health Care. Canadian Nurse. https://www.cna-aiic.ca/~/media/cna/page-content/pdf-en/primary-health-careposition-statement.pdf
- World Health Organization. (1978, September 6–12). Primary health care: report of the International Conference on Primary Health Care, Alma-Ata, USSR, 6–12. <u>https://apps.who.int/iris/handle/10665/39228</u>
- 9. Moore JE, Mascarenhas A, Bain J, Straus SE. Developing a comprehensive definition of sustainability. *Implementation Science*. 2017;12(1). doi:10.1186/s13012-017-0637-1
- Canadian Nurses Association. Advanced practice nursing: a pan-Canadian framework. Ottawa: Canadian Nurses Association. https://www.cna-aiic.ca/-/media/cna/page-content/pdf-en/apn-a-pan-canadian-framework.pdf
- Dicenso A, Bourgeault I, Abelson J, et al. Utilization of Nurse Practitioners to Increase Patient Access to Primary Healthcare in Canada – Thinking Outside the Box. *Nursing Leadership.* 2010;23(sp):239-259. doi:10.12927/cjnl.2010.22281
- 12. Canadian Federation of Nurses Unions. Media Release. https://nursesunions.ca/wpcontent/uploads/2017/11/2017-11-13\_NursePractitionerWeek\_EN-1.pdf
- Gould ON, Johnstone D, Wasylkiw L. Nurse practitioners in Canada: Beginnings, benefits, and barriers. *Journal of the American Academy of Nurse Practitioners*. 2007;19(4):165-171. doi:10.1111/j.1745-7599.2007.00210.x
- Sangster-Gormley E. Interprofessional Collaboration: Co-workers' Perceptions of Adding Nurse Practitioners to Primary Care Teams. *Quality in Primary Care*, 2015: 23(3). http://primarycare.imedpub.com/interprofessional-collaboration-

#### coworkersperceptions-of-adding-nurse-practitioners-toprimary-care-

#### teams.php?aid=5485

- Arksey H, O'Malley L. Scoping studies: Towards a methodological framework. *International Journal of Social Research Methodology*, 2005: 8(1), 19–32. <u>https://doi.org/10.1080/1364557032000119616</u>
- 16. Arenson A. Questionable logic behind clinic closure. *Winnipeg Free Press*. Published04 August 2017. Accessed January 30, 2018
- Bruineman M. Clinic suffers from lack of funding: *Examiner*. Published 06 September 2001. Accessed January 30, 2018
- Bruineman M. Clinic closure creates anxieties for seniors: Sandy cove facility closing Nov 19. *Examiner*. Published 18 September 2001. Accessed January 30, 2018
- Conrad K. Lack of funding closes clinic. Okotoks Western Wheel. <u>http://www.westernwheel.com/article/Lack-of-funding-closes-clinic-20151211.</u> Published 11 December 2015. Accessed January 30, 2018
- Gilchrist L. Two medical clinics facing closure. *Peterborough this Week*. Published 16 September 2005. Accessed January 30, 2018
- 21. Gilchrist L. Clinics' pending closure to see 2,000 lose primary health care.*Peterborough this Week.* Published 23 December 2005. Accessed January 30, 2018
- 22. Guest A, Australia's first stand-alone nurse practitioner clinic closes [Radio broadcast]. *The World Today, ABC News*, 17. <u>https://www.abc.net.au/radio/programs/worldtoday/australias-first-stand-alone-nurse-practitioner/4017410</u>. Published 17 May 2012. Accessed January 30, 2018

- 23. King K. Family health teams prepare for transition. *Peterborough Examiner*. Published16 December 2005. Accessed January 30, 2018
- 24. Kovach J. Ottawa not to blame if medical centre closes: MP. *Peterborough Examiner*. Published 22 December 2001. Accessed January 30, 2018
- 25. Looming Walk In. Looming walk-in closure disappoints. Vernon Morning Star. <u>https://www.vernonmorningstar.com/news/looming-walk-in-closure-disappoints/</u>. Published 5 December 2015. Accessed January 30, 2018
- 26. Marchen, J. Minister vows to look into clinic's dilemma. *Peterborough Examiner*Published 18 December 2010. Accessed January 30, 2018
- 27. McEachern L. Project redefines nursing role: Nurse practitioner pilot project wrapping up final year. *Evening News*. Published 21 October 2002. Accessed January 30, 2018
- Punch R. Funding runs out for clinics. *Peterborough Examiner*. Published 15 September 2005. Accessed January 30, 2018
- 29. Punch R. Nowhere to go for patients: *Peterborough Examiner*. Published 16 September2005. Accessed January 30, 2018
- Wedley B. Clinic's closure causes concern. *Peterborough Examiner*. Published 12 January 2006. Accessed January 30, 2018
- Herhalt C. Care clinic no longer needed. *Waterloo Region Record*. Published 21 April 2015. Accessed January 30, 2018
- 32. Ramsay J. Elmvale nurse practitioner clinic closes. *Barrie-Advance*.
   <u>https://www.simcoe.com/news-story/4247236-elmvale-nurse-practitioner-clinic-closes/</u>
   Published 29 November 2013. Accessed January 30, 2018

- 33. Hutton R. Doors to close on Niagara Falls nurse practitioner clinic. Niagara This Week. <u>https://www.niagarathisweek.com/news-story/4040234-doors-to-close-on-niagara-falls-nurse-practitioner-clinic/.</u> Published 20, August 2013. Accessed January 30, 2018
- 34. MEI. (2016, April 12) Super nurse clinics: When bureaucracy prevents better access to care. <u>https://www.iedm.org/59836-super-nurse-clinics-when-bureaucracy-preventsbetter-access-to-care/</u>
- 35. Carter, M. (2015, August 27). Dawson creek clinic facing closure. *Prince George Citizen*
- 36. Buykx P, Humphreys JS, Tham R, et al. How do small rural primary health care services sustain themselves in a constantly changing health system environment? *BMC Health Services Research*. 2012;12(1). doi:10.1186/1472-6963-12-81
- Gruen RL, Elliott JH, Nolan ML, Lawton PD, Parkhill A, McLaren CJ, Lavis JN.
   Sustainability science: an integrated approach for health-programme planning. *The Lancet*. 2008:372(9649), 1579–1589. https://doi.org/10.1016/S0140-6736(08)61659-1
- Lennox L, Maher L, Reed J. Navigating the sustainability landscape: a systematic review of sustainability approaches in healthcare. *Implementation Science*. 2018;13(1). doi:10.1186/s13012-017-0707-4
- Wakerman J, Humphreys JS. Sustainable primary health care services in rural and remote areas: Innovation and evidence. *Australian Journal of Rural Health*.
   2011;19(3):118-124. doi:10.1111/j.1440-1584.2010.01180.x
- 40. Stirman SW, Kimberly J, Cook N, Calloway A, Castro F, Charns M. The sustainability of new programs and innovations: a review of the empirical literature and

recommendations for future research. *Implementation Science*. 2012;7(1). doi:10.1186/1748-5908-7-17

- 41. Sibthorpe BM, Glasgow NJ, Wells RW. Emergent themes in the sustainability of primary health care innovation. *Medical Journal of Australia*. 2005;183(S10). doi:10.5694/j.1326-5377.2005.tb07185.x
- 42. Martin GP, Weaver S, Currie G, Finn R, Mcdonald R. Innovation sustainability in challenging health-care contexts: Embedding clinically led change in routine practice. *Health Serv Manage Res. 2012;25(4):190-9*.doi:10.1177/0951484812474246.
- 43. Considine J, Fielding K. Sustainable workforce reform: case study of Victorian nurse practitioner roles. Aust Health Rev. 2010;34(3):297-303.
  https://doi.org/10.1071/AH08727
- 44. Fox A, Gardner G, Osborne S. A theoretical framework to support research of health service innovation. Aust Health Rev. 2015;39(1):70-75. doi:10.1071/ah14031
- 45. Keating SF, Thompson JP, Lee GA. Perceived barriers to the sustainability and progression of nurse practitioners. *International emergency nursing*, 2010 18(3), 147-153. <u>https://doi.org/10.1016/j.ienj.2009.09.003</u>
- 46. Fleiszer AR, Semenic SE, Ritchie JA, Richer M-C, Denis J-L. The sustainability of healthcare innovations: a concept analysis. *Journal of Advanced Nursing*. 2015;71(7):1484-1498. doi:10.1111/jan.12633
- 47. Greenhalgh T, Robert G, Macfarlane F, Bate P, Kyriakidou O. Diffusion of innovations in service organizations: systematic review and recommendations. *Milbank Q*. 2004;82(4):581-629. https://doi.org/10.1111/j.0887-378X.2004.00325.x

- 48. Scheirer, M. A., Hartling, G., & Hagerman, D. (2008). Defining sustainability outcomes of health programs: Illustrations from an on-line survey. *Evaluation and Program Planning*, 31(4), 335–346. <u>https://doi.org/10.1016/j.evalprogplan.2008.08.004</u>
- 49. Proctor E, Luke D, Calhoun A, et al. Sustainability of evidence-based healthcare: research agenda, methodological advances, and infrastructure support. *Implement Sci.* 2015;10:88. <u>https://doi.org/10.1186/s13012-015-0274-5</u>

Table 2-1: Inclusion and Exclusion Criteria

nclusion Criteria	Exclusion Criteria
• Primary research reports published in peer -	• Study dated before 1990
reviewed journals of any study design that were relevant to the research questions.	• Any other country other than those listed in inclusion list
• Relevant reviews (systematic, scoping,	• Non English studies and articles
integrative, narrative) published in peer reviewed	• Published reports, relevant reviews, policy documents,
journals	editorials, briefs, interviews and newspaper articles which
• Relevant policy documents from the included	focused on the following:
countries	Hospital Closure, Conversion or Consolidation.
• Relevant editorials, briefs, interviews, policy	• Hospital or clinic closure due to physician shortage,
documents, interviews, newspaper and any other	physicians leaving or difficulty recruiting physicians.
media article.	•
• Dates: 1990-2018	• Literature that focused on environmental/ecological sustainability, sustainable development (or development
English Only	goals).

- Use of the term Nurse Practitioner, clinics, closure, primary health care, primary care and sustainability.
- Use of any term or designation for the Nurse Practitioner role other than Nurse Practitioner (Advanced Practice Nurse, Clinical Nurse Specialist)
- Private Clinics, Abortion Clinics

Table 2-2: Summary of Clinic Closure Grey Literature Articles (Google, Google Scholar,Canadian Newsstream, News and Newspapers, News, Policy and Politics)

Author	Setting	Source	Key Points Addressed	Comments
Arensen (2016)	Winnipeg,	Newspaper	Winnipeg Regional	• The factors which impacted
	MB, Canada	Article	Health Authority closing	clinic closure are lacking.
		Winnipeg	4 Quick Care Clinics and	• Stakeholders, providers and
		Free Press	a Primary Care Clinic	patients were not consulted
			staffed with Nurse	prior to the decision being
			Practitioners stating 'to	made.
			save money' was the	
			cause of closure.	
D : (2001 )		N		
Bruineman (2001a)	Barrie, ON,	Newspaper	• North Innisfil Health	• An interesting funding
	Canada	Article	Services Clinic in Barrie,	approach to seniors' care but
		The	Ontario faced closure	one that was ultimately not
		Examiner	without cash infusion of	sustainable.
			\$450,000.	• No answers as to why it was
				not sustainable.

				44
Author	Setting	Source	Key Points Addressed	Comments
			• The provincial	• First article about Sandy
			government provided the	Cove Clinic closure
			clinic with temporary	
			funding which runs out	
			in mid-November.	
			• The clinic is a family	
			practice designed for	
			seniors and is staffed by	
			a doctor and nurse	
			practitioner who work on	
			salary. This allows them	
			to spend more time with	
			their older patients, who	
			generally require more	
			attention.	
Bruineman	Barrie ON,	Nougpopor	Article mentions senior's	• This is the second article
	Canada	Newspaper Article		• This is the second article about the closure of the
(2001b)	Callada	Alucie	family practice closing.	
			Both NP and Physician	Sandy Cove Clinic.

Author	Setting	Source	Key Points Addressed	Comments
Carter (2015)	Prince George, BC, Canada	The Examiner Newspaper Article, Prince George Citizen	<ul> <li>funded by government and salaried wages.</li> <li>Dawson Creek Medical Clinic set to close.</li> <li>Lone physician is retiring.</li> <li>Nurse Practitioner also at this clinic has to leave.</li> </ul>	<ul> <li>Questions arise as to why clinic sustainability was not maintained with the NP once the physician retired.</li> <li>There was a lack of information on why the NP</li> </ul>
				had to leave and was it due to funding cuts or personal choice.
Conrad (2015)	Okotoks, AB, Canada	Newspaper Article Okotoks Western Wheel	<ul> <li>Closure of the Sheep River Nurse Practitioner Clinic.</li> <li>No funding model in place to fund NPs and keep clinic open.</li> </ul>	• Lack of funding model for NP practice at issue.

Author	Setting	Source	Key Points Addressed	Comments
Hutton (2013)	CAN	Newspaper	Niagara Nurse	Mismanagement of funding
		Article	Practitioner clinic	at issue.
Niagara This Week			closing.	
			• The Ministry of Health is	
			shutting down the clinic	
			in the wake of a forensic	
			audit which found	
			evidence funding for the	
			clinic provided by the	
			Ministry had not been	
			spent appropriately.	
Gilchrist	Peterborough,	Newspap	• 1 <sup>st</sup> Article on Greater	Lack of information as to
(2005a)	ON, Canada	er	Peterborough Health	whether this was a pilot
(2003a)	OIV, Canada	Article,	Care Alliance NP Clinic	project and how the clinic
		Peterboro	(and a peds clinic)	was funded.
		ugh This	written a month before a	was funded.
		Week	scheduled closure.	
		week	scheduled closure.	

Author	Setting	Source	Key Points Addressed	Comments
Gilchrist	Peterborough,	Newspap	• 2 <sup>nd</sup> Article details	• Same clinic as above. Lack
(2005b)	ON, Canada	er	Greater Peterborough	of information as to whether
(2005b)		Article,	Health Care Alliance NP	this was a pilot project and
		Peterboro	clinic closure 1 week	how the clinic was funded.
		ugh This	before scheduled closure	
		Week	and transfer of patients	
			to potential family health	
			teams.	
Guest	Australia	Radio	• Interview to discuss why	• This article points out that
(2012)		Interview	the nation's first stand-	funding remains one of the
		Transcrip	alone nurse practitioner	main barriers that exist for
		t	clinic is closing and	NPs and one of the huge
			reverting to a traditional	roadblocks to sustainability
			GP model.	of the role.
			• Access to funding	
			rebates for NPs in	
			Australia is limited and	
			the NP clinic was not	

				40
Author	Setting	Source	Key Points Addressed	Comments
			profitable compared to a	
			physician funded clinic.	
Herhalt (2015)	Kitchener,	Newspap	Primary Care Clinic	• Physician in the article states
	ON, CAN	er	located at Grand River	that "NP will be able to apply
		Article,	Hospital closing and NP	on a comparable position"
		Waterloo	position will be cut. This	
		Region	is due to more medical	
		Record	residents graduating in	
			the area so there is less	
			of a need for clinic and	
			NP position.	
			• A satellite clinic in	
			downtown Kitchener	
			greatly improved access	
			to care and was given as	
			a reason for closure.	
King	Peterborough,	Newspap	• 3 <sup>rd</sup> Article from a	Greater Peterborough Health
(2005)	ON, Canada	er	different Newspaper	Care Alliance NP Clinic
		Article,	about the Greater	(same clinic as 2 others

				49
Author	Setting	Source	Key Points Addressed	Comments
		Peterboro	Peterborough Health	above). Lack of information
		ugh	Care NP Clinic. Written	as to whether this was a pilot
		Examiner	a month before	project and how the clinic
			scheduled closure	was funded.
Kovach (2001)	Peterborough,	Newspap	• Clinic employs a nurse	• Another article detailing
	ON, Canada	er Article	practitioner and support	ending of funding for NP
		Peterboro	staff. It provides medical	Clinic.
		ugh	care to more than 3,000	
		Examiner	people in the community	
			who have no family	
			doctor.	
			• There's no money left to	
			run the clinic past	
			January.	
			• MPP Gary Stewart	
			places blame on federal	
			government and there is	
			no word if province will	
			kick in funding.	

				30
Author	Setting	Source	Key Points Addressed	Comments
"Looming	Vernon, BC,	Newspap	• Vernon Walk in Clinics	• Owner of the walk in clinics
Walk-In", 2015	Canada	er	to close due to shortage	identifies the need for more
		Article,	of physicians and lack of	NPs
The Morning		The	funding for nurse	
Star		Morning	practitioners.	
		Star	• Owner of Walk In	
			Clinics proposes that BC	
			health needs to recruit	
			more doctors and	
			provide funding for	
			nurse practitioners.	
Marchen	Peterborough,	Newspap	• Funding dilemma may	• Article detailing a funding
(2001)	ON, Canada	er	close a NP clinic in	issue as reason for potential
		Article.	Havelock-Belmont-	NP clinic closure.
		Peterboro	Methuen.	
		ugh		
		Examiner		
McEacheran	Pictou West,	Newspap	Primary care pilot	• This article highlights how
(2002)	Nova Scotia	er Article	project draws to a close	one community, through a

				51
Author	Setting	Source	Key Points Addressed	Comments
		Evening	in Pictou West, Nova	NP primary care project,
		News	Scotia. It is in its final	began to understand exactly
			year of funding.	what NPs do.
			• Community now better	• Questions arise as to why this
			understands what a nurse	pilot project was not
			practitioner does.	sustainable.
Punch	Peterborough,	Newspap	• 4 <sup>th</sup> Article on Greater	• This article is dated from
(2005a)	ON, Canada	er Article	Peterborough Health	2005 and Health Centre
		Peterboro	Care Alliance NP Clinic	Chair notes that they may be
		ugh	• 2,400 patients (1,700	getting close to fitting NPs
		Examiner	children) could be	into the system, yet years
			without primary health	later the same issues still
			care if the health unit	exist.
			follows through on plans	
			to close two nurse	
			practitioner clinics.	
			• Health Centre Chair	
			states 'we're not in a	
			position to actually fit	

				32
Author	Setting	Source	Key Points Addressed	Comments
			nurse practitioners	
			immediately into the	
			system. We think it's	
			getting close. Hopefully	
			the solution is not too far	
			away,'	
			• Funding is not flowing to	
			the clinics due to	
			logistical issues. The	
			Ministry is not	
			responding effectively to	
			the community.	
Punch	Peterborough,	Newspap	• 5 <sup>th</sup> Article on Greater	•
(2005b)	ON, Canada	er Article	Peterborough Health	
		Peterboro	Care Alliance NP Clinic	
		ugh	Closure	
		Examiner		
Ramsay (2013)	Barrie, ON,	Newspap	• After being open once a	• This was a nurse practitioner
	Canada	er	week for two years, the	led clinic that was closed due

Author	Setting	Source	Key Points Addressed	Commonto
Author	Setting	Source	Key Points Addressed	Comments
		Article,	Elmvale nurse-	to a staffing shortage and
		Barrie	practitioner led clinic has	underutilization. The article
		Advance	closed.	does not mention any
			• The closure was due to a	consultation with
			staffing shortage and	stakeholders or patients prior
			underutilization of the	to the clinic being closed.
			clinic backs up the	
			decision.	
MEI, 2016	Ottawa, ON,	Newspap	• Quebec's Health	• This article is an example of
	Canada	er Article	Department is	the barriers that continue to
		Canada	senselessly blocking the	exist for NPs. In this
		Newswir	opening of clinics run by	situation/article it is not clear
		e	nurse practitioners who	why the Quebec Government
			specialize in front-line	is blocking the opening of
			care.	these clinics.
			• These doctor-less super	
			clinics would respond to	
			real needs among the	

				54
Author	Setting	Source	Key Points Addressed	Comments
			population, access to	
			front-line care being one	
			of the main failings of	
			Quebec's health system.	
			• Moreover, a nurse	
			practitioner costs the	
			health care system	
			around $1/3$ of what a	
			general practitioner	
			costs.	
Wedley	Peterborough,	Newspap	• 6th article on Greater	• Details about the same NP
(2006)	ON, Canada	er Article	Peterborough Health	clinic closure and patients
		Peterboro	Care Alliance NP Clinic	will have to use ED while
		ugh This		transitions happen to family
		Week		health teams.

# Table 2-3

Summary of Studies PHC Sustainability Included from CINAHL, Medline, Academic Complete,

## Health Policy Reference

Author	Method	Objectives and Aims	Key Findings	Comments
Country	Design	Setting		
	Sample			
Buykx et	Mixed	• Evaluation to	• Study was only	• This article highlights the
al. (2012)	Methods	explore and	underway for four	benefits of working with health
		illustrate how	years and it was	organizations and stakeholders
AUS	6 Year	one primary	already evident that	to benchmark and use primary
	Longitudinal	health care	the Elmore Primary	health care evidence to
	Evaluation	service has	Health Service is	formulate health policy and
		proactively	exemplary compared	programs designed to meet the
		managed	with many small rural	needs of the community.
		change to	community health	
		remain	services that have	
		sustainable.	struggled to serve their	
			communities	
			adequately.	

5	6
.)	0

Author	Method	Objectives and Aims	Key Findings	Comments
Country	Design	Setting		
	Sample			
		•	• The EPHS model has	
			demonstrated its	
			capacity to be	
			proactive rather than	
			simply reactive	
			through aligning its	
			approach with	
			government policies,	
			ongoing monitoring of	
			its service	
			performance,	
			maintaining strong	
			engagement with the	
			local community,	
			maximizing	
			opportunities for	
			alternative funding,	

Author	Method	Objectives and Aims	Key Findings	Comments
Country	Design	Setting		
	Sample	2000.8		
	Sample			
			initiating workforce	
			succession planning	
			and actively	
			partnering research	
			evaluation designed to	
			ensure quality	
			improvement.	
Gruen et al.	Quantitative	• Using a SR to	• Sustainability is	• While the focus of this research
(2008)		derive a	increased to the degree	is not on program sustainability,
	Systematic	practical	to which the	the information in this article
AUS	Review	conceptual	components of the	may be useful to apply to NP
		framework for	system are connected	role sustainability.
		understanding	and aligned—an	
		health-program	indication of system	
		sustainability;	equilibrium.	
		and to use this	• Unsustainable	
		framework to	programs are a form	

				58
Author	Method	Objectives and Aims	Key Findings	Comments
Country	Design	Setting		
	Sample			
		propose an	of disequilibrium, in	
		approach to	which the health status	
		planning for	of a population, the	
		health-program	program implemented	
		sustainability.	within the population,	
			and the drivers of the	
			program are	
			disconnected and	
			misaligned.	
			Conceptual model	
			framework was	
			devised.	
Lennox,	Mixed	• This study	• In total, 62	• This is the first review to
Maher &	Methods	aimed to	publications	consolidate available
Reed		identify what	identifying a	approaches for sustainability
(2018)	Systematic	approaches are	sustainability	across diverse health care
	Review	available to	approach were	settings.

				57
Author	Method	Objectives and Aims	Key Findings	Comments
Country	Design	Setting		
	Sample			
UK		assess and	included in this review	• This review could serve as a
		influence	(32 frameworks, 16	sustainability knowledge base.
		sustainability	models, 8 tools, 4	
		in health care	strategies, 1 checklist	
		• Where have	and 1 process).	
		approaches	Constructs across	
		come from and	approaches were	
		how have they	compared and 40	
		been	individual constructs	
		developed?	for sustainability were	
		• What are the	found.	
		approaches	• Although similarities	
		key	were found, no	
		characteristics?	approaches contained	
			the same combination	
			of constructs nor did	
			any single approach	

Author Country	Method Design Sample	Objectives and Aims Setting	Key Findings	Comments
Wakerman & Humphreys (2011)	Mixed Methods	<ul> <li>To highlight how evidence from studies of innovative</li> </ul>	<ul> <li>capture all identified constructs.</li> <li>From these results, a consolidated framework for sustainability constructs in health care was developed.</li> <li>66% of approaches reviewed saw sustainability as a process rather than an end state.</li> <li>Authors propose a systems-based solution to PHC sustainability</li> </ul>	

1	1	
h		
υ	1	

Author	Method	Objectives and Aims	Key Findings	Comments
Country	Design	Setting		
2	Sample	C		
	Discussion of	rural and	• Highlights the paucity	
AUS	a Previous	remote models	of comprehensive,	
	Study	of service	high-quality	
	Evaluating	provision can	evaluations of PHC	
	Innovative	inform global	health services in	
	PHC models.	health system	order to better inform	
		reform in order	PHC sustainability	
		to develop	• An effective systemic	
		appropriate,	approach to	
		accessible	sustainability relies on	
		PHC services	alignment of all levels	
		to 'difficult-to-	of health services with	
		service'	the external policy	
		communities.	environment.	



Prisma Diagram, Search Results and Article Selection


# Chapter Three: A Critical Analysis of Funding and Compensation Models: Sustainability of the Nurse Practitioner Role in Canada (Paper 2)

Raelene Marceau Dr. Stephanie Montesanti Dr. Kathleen Hunter Dr. Tammy O' Rourke

Marceau, R., Montesanti, S., Hunter, K. and O' Rourke, T., 2020. A Critical Analysis of Funding and Compensation Models: Sustainability of the Nurse Practitioner Role in Canada. Manuscript in Preparation, to be submitted to the *Journal of Nurse Practitioners*.

#### Introduction

The number of Nurse Practitioners (NPs) in Canada has increased substantially over the past decade and the role saw the highest growth rate of 8.02% between 2017 and 2018.<sup>1</sup> Integrating NPs into health care environments has been shown to improve patient outcomes and contribute to high-quality care. The importance of NPs to organizational and system outcomes is also well documented and NPs bring significant added value to various models of health care delivery and the health care system in Canada.<sup>2</sup>

Optimizing the contribution of NPs, including ensuring that they are able to work to their full scope of practice, remains a challenge across the country.<sup>1,2</sup> In spite of the positive evidence indicating the NP role can increase access to quality, cost-effective care for patients, barriers to full implementation of the role persist.<sup>2,3,53</sup> The lack of a sustainable funding model has been cited as a primary barrier to the optimization of NPs in Canada. Sustainable funding and adequate infrastructure and resources, have been cited as the most critical factors to support the NP role going forward.<sup>2,3,53</sup>

We conducted a critical analysis of the current state of Canada's NP funding models utilizing the Sustainability of Innovation Framework (SIF).<sup>4</sup> Our analysis compares Canada's current NP funding models using a theoretical approach. Our results and discussion propose future directions to assist healthcare leaders, funders and policy-makers in supporting a variety of NP roles in Canada to meet the demands of different patient populations in our health system.

#### Objectives

Significant jurisdictional variation exists in terms of NP funding models within the Canadian health care system and it is apparent this variability creates significant challenges associated with the NP role. This includes: difficulty in defining and legitimizing the role; underutilization of NPs; scope of practice and liability concerns; insufficient funding for practicerelated costs; marginalization of the role; and significant NP wage and benefit disparities across the country.<sup>2,5</sup> The purpose of this analysis is to create a comprehensive review of the funding models for NPs in a variety of healthcare settings, to compare and analyze these NP funding models across all Canadian jurisdictions and to propose future directions to assist in alleviating the funding challenges Canadian NPs are experiencing.<sup>6-11</sup> We use the SIF to guide our analysis and this theoretical framework was developed to identify factors influencing primary health care (PHC) innovation sustainability.<sup>4</sup> This research is expected to advance our knowledge in regards to the development of sustainable, modernized NP funding models to support the work of health care leaders, funders and policy-makers across Canada.

#### Background

#### **Overview of the Nurse Practitioner Role in Canada**

The NP role in Canada has had a chaotic history with periods of inactivity and near eradication in the 1970s and '80s; coupled with periods of growth and role re-integration in the 1990's and 2000's.<sup>12</sup> Canadian NPs are required to have a minimum Master's level preparation with a growing number also holding doctoral degrees. NPs possess an extensive body of knowledge and skills, diagnose acute and chronic illness, prescribe medications, are accountable and responsible for their own professional practice, and all Canadian jurisdictions<sup>1</sup> have legislation and regulation supporting the role.<sup>13</sup> Literature indicates that NPs provide high quality care, are cost effective and work well with, and greatly complement skill sets of other professionals when working within collaborative teams.<sup>14-18</sup>

The number of NPs in Canada is increasing steadily and the role is the fastest developing sector of the nation's health human resource pool with over 6,000 NPs currently licensed to practice in Canada.<sup>19,20</sup> Despite substantial progress, the number of NPs is still insufficient to meet the growing needs of Canada's population and there are considerably less NPs practicing in rural and remote areas of the country.<sup>20</sup> Considerable opportunity exists to improve expanded NP role integration within the Canadian health care system.<sup>2,21</sup>

<sup>&</sup>lt;sup>1</sup> The term jurisdiction indicates the ten provinces and three territories in Canada who have the authority and power to regulate, oversee and provide health care delivery in their specific geographic area.

#### **Overview of Canadian NP Funding Models**

The integration and optimization of NPs within the health care system depends on the NP funding model within a jurisdiction. A funding model typically has three defining characteristics: a) it generally revolves around a primary source of funding, such as governments or organizations, b) within this primary source of funding, there are decision makers; individuals who dictate the flow of funds (such as government leaders), and c) it focuses on the motivations of decision-makers who control the flow of funds.<sup>22</sup>

Based on these characteristics, an NP funding model involves the payment or remuneration for NP services or employment. Models are based on numerous factors including base salary, bonuses, incentives, health care practice and individual/team employment settings.<sup>6</sup> Every Canadian jurisdiction has their own NP funding model, however, there are significant variations in the defining characteristics of these models across the country.<sup>21,22</sup> It has been proposed that these variations may be necessary to accommodate jurisdictional differences in NP scope of practice, responsibility and accountability, as well as the needs of unique patient population characteristics.<sup>24</sup>

The majority of Canadian NPs are funded either directly or indirectly by jurisdictional governments. These funding models are structured to comply with legislation and regulations set out by their respective jurisdiction. Most NPs are funded indirectly through the use of government money disbursed to an organization. These organizations typically include hospitals, regional health authorities or health care agencies. NPs in these organizations are employees and typically receive an hourly wage or salary. See Figure 3-2 for an image overview of NP funding models in Canada.

A variety of government funding envelopes have been introduced as "special funding" to support NP role implementation. These "special funding" envelopes typically support pilot initiatives driven by perceived and actual physician shortages.<sup>12</sup> Relatively few Canadian NPs are employed by private or non-profit corporations. Only three Canadian jurisdictions report this type of NP employment (Alberta, Newfoundland and the Northwest Territories).<sup>2</sup> NPs in Prince Edward Island, Ontario, and Saskatchewan may work with a physician in a fee for service environment, in this type of model the physician pays the NP an hourly wage or salary from the revenue generated from physician billing. Very few NPs in Canada are self-employed, these NPs receive funds from direct billing of patients.<sup>6,25,26</sup>

Two jurisdictions, British Columbia and Alberta, have recently introduced a new NP funding model. In 2018, British Columbia approved the allocation of funds for NPs as contract providers. These funds can be used by NPs to join a pre-existing primary care team or to set up a new team. The duration of these contracts is three years, and the goal of this funding model is to increase the number of NP positions in the jurisdiction's primary care system by 200 NPs. These NPs are expected to provide full-scope primary care services and increase health care access to British Columbia residents.<sup>27,28</sup>

In early 2019, the Alberta government announced the Nurse Practitioner Support Program which was also created to increase the number of NPs in primary care through a revised team funding formula. The government has committed \$38.5 million in funding over 3 years, to support the continued employment of NPs in existing primary care teams and encourage them to hire more NPs (n=50). These funds are distributed to primary care networks (PCNs) through an application to pay for NP roles within interprofessional teams.<sup>29</sup>

The most innovative NP funding model is found in Ontario, the largest jurisdiction in the country. NP-led clinics which were introduced in 2007, receive funding directly from the government to implement a team-based, PHC initiative. NP-led clinics were introduced to improve access to primary care in areas where a large proportion of the population are without a family physician.<sup>30</sup> Within this model, NPs are the most responsible primary care provider, NPs provide clinical and operational supervision and represent a significant interest on the board of directors.<sup>31</sup>

Every Canadian jurisdiction has an NP funding model, however, there are significant variations in the structure and processes of these models.<sup>23</sup> It is proposed that these variations may be necessary to accommodate jurisdictional differences in NP scope of practice, responsibility and accountability, as well as the needs of unique patient populations.<sup>24</sup> Although the number of NPs continues to grow, the ratio of NPs to citizens remains low (14 NPs per 100,000 population) and the number is insufficient to meet the growing needs of the nations' population.<sup>2</sup>

#### **NP Funding Models: An International Perspective**

There are similar barriers to NPs role utilization in most countries. These barriers include opposition from stakeholders, vested interests (e.g., physician workforce), government legislative/ regulatory/policy barriers, financing and reimbursement issues, and challenges in uptake at organizational levels. It is useful to review what is happening from a funding perspective in other countries with similar NP experiences. The NP role was first introduced in the U.S. in the mid to late 1960s, followed by the United Kingdom (U.K.) in the mid-1980s, then Australia in the 1990s.<sup>32</sup> In the U.S., regulation and funding of NP roles vary by state and there is significant variation in wages and compensation models. Most NPs are funded through thirdparty payers who directly reimburse for services. These payers include: a) Medicare, b) Medicaid, c) commercial indemnity insurers, d) commercial managed care organizations and d) businesses or schools wanting health services for employees or students.<sup>33–35</sup> Direct reimbursement is available to NPs (individually or through an employer, clinics or group practice) who provide service to patients enrolled in Medicare at 85%, or Medicaid at 60% of the physician rate. Managed care organizations' financial arrangements may include fee-for-service and risk-bearing models and compensation are provided via salary or per member-per-month agreements.33,35

In Australia, the majority of NPs are employed in public hospitals, these NPs are funded through State Government salaries. In primary care, NP services are remunerated through the national Medicare Benefits Schedule (MBS). Funds are accessed by private NPs and the reimbursement rate is at least fifty percent less than physicians. This discrepancy in MBS reimbursement for NP services has resulted in unsustainable business models for NPs practicing in primary care.<sup>32,35</sup>

In the UK there is no regulation or national requirements for obtaining or using the NP title other than a regulatory framework for nurse prescribing.<sup>35</sup> Essentially, the diverse, unregulated and flexible nature of the NP role within the UK varies somewhat from the state of NP practice in other western countries. Funding for NPs in the UK is provided through the National Health Service (NHS) which is the publicly funded national health system overseen by

the Department of Health.<sup>34,35</sup> This international comparison of funding models from other countries illustrates the diversity of funding for NPs in western countries and is useful to use as a tool to guide our results and discussion below.

#### Method

The critical analysis methodology is well-developed and broadly applied in qualitative health research. Critical analysis is the ability to recognize or question, taken-for-granted assumptions and their effects. It attempts to expose hidden power, embraces multiple perspectives, gives space to unheard voices, and stimulates social criticism.<sup>36</sup> The process of critical analysis allows for the interaction of specific topic elements to be examined while at the same time closely studying individual components.<sup>36</sup> Theoretical frameworks are used as an analytical tool to provide guidance for the critical analysis process. The SIF by Fox, Gardner and Osborne was chosen to examine and interpret the data collected to better inform our understanding of Canadian NP funding models.<sup>4</sup>

#### **Theoretical Framework**

The NP role is frequently cited as one of the most innovative solutions to Canada's health care system challenges, yet sustainability of the role remains in question and implementation of the role has been cited as the biggest innovation failure in the country.<sup>2</sup> Given these challenges, the SIF developed by Fox, Gardner and Osborne was chosen as the most suitable tool to guide this analysis.<sup>4</sup> The framework merges the concepts presented by Greenhalgh and colleagues' systematic review of health service innovation,<sup>37</sup> and the Dynamic Sustainability Framework by Chambers, Glasgow and Stange (See Figure 3-1).<sup>4,38</sup> It provides a clear framework to guide an in-depth examination of health care innovations and describes five factors that influence sustainability.<sup>4</sup>

Each sustainability factor includes specific elements required to achieve health care innovation sustainability. Factor #1 Political: appropriate government and policy alignment and evidence indicate that a political focus on one particular policy can have a strong influence on the sustainability of a health innovation. Factor # 2 Organizational: Innovation has to be

adaptable to support the local context and address perceived needs. Factor #3 Financial: The innovation is dependent on a specified source and length of funding. Factor #4 Workforce: Innovations aligned with the values and needs of the existing human resource pool are more readily adopted and sustained. Factor #5 Innovation Specific: Innovations must be able to respond to dynamic changes in funding and policy environment, and continue to evolve.<sup>4</sup>

#### **Data Collection and Analysis**

Data to inform this critical analysis was collected from individuals involved with NP education and regulation, NP professional organizations and documents and articles via an internet search. The data were reviewed and relevant NP funding information was organized and summarized into <u>Table 3-1</u> and <u>3-2</u>. This data is presented in an overview format. Using a critical lens, each factor in the Sustainability of Innovation Framework was used to systematically assess and examine current jurisdictional funding models to determine whether these models address the factors necessary to sustain the NP role.

#### Results

An overview of the existing NP funding models in Canada was created and compared across the nation. This data was critically examined and the results are organized according to the five factors in the SIF theoretical framework.<sup>4</sup>

#### **Factor 1: Political**

Jurisdictional government funding is the main NP reimbursement source and there is variation in how each jurisdiction implements NP designated funding. This type of funding has been instituted with no long term, sustainable vision for NP role utilization. The continued lack of a long-term vision and funding strategy for Canadian NPs is the result of outdated and entrenched health policies that fail to fully advance the NP role.<sup>2</sup> The lack of transformation in Canadian health policy fails to facilitate optimization of the NP role and reinforces barriers that

prevent NPs from working to their full scope of practice, and achieving a sustainable funding solution across Canada.<sup>2</sup>

### **Factor 2: Organizational**

Many Canadian organizations and corporations employ NPs and have multi-level, historic structures and processes that limit change and flexibility in the funding models provided. These types of long-established organizational structures and processes impact the uptake and sustainability of the NP role. These restrictions are especially salient in organizations that adopt a physician-led hierarchical infrastructure in which physicians have final decision-making authority.<sup>39</sup> A prime example of this type of hierarchy are Alberta Primary Care Networks (PCN). These networks are government-funded and physician-led and operated.<sup>40</sup> Despite legislation and regulation to support independent NP practice, the 'physician driven' practices of these organizations and funding models typically restrict NP utilization.<sup>39</sup>

#### **Factor 3: Financial**

As demonstrated in Figure 3-1 and Table 3-1, most Canadian NP positions are funded through jurisdictional government-based funding and NPs receive a salary for their services. These salaries are administered through a government agency in each jurisdiction. There is considerable jurisdictional variation in the structure and processes of this funding and there is even more variability in "special funding" envelopes. These grant-type NP funding initiatives continue to be short-term and generally unsustainable.<sup>2</sup> There is a rigid structure to jurisdictional funding which often lacks a comprehensive, team-based focus to support NP and interprofessional integration in different types of PHC environments. To add to the lack of sustainability in NP funding models, NP compensation is not keeping up with market value and there are substantial differences in wage and benefit reimbursement across the country. All Canadian jurisdictions fail to meet the recommendations made in the Hay Group Report (2013) which has specific evidence-based compensation guidelines for NPs.<sup>41</sup>

#### **Factor 4 Workforce**

There is a lack of transparency in NP human resource information and no apparent coordinated jurisdictional or national NP workforce strategy. Very few NP organizations or affiliations have up to date workforce information and some NP workforce information is buried within Registered Nursing statistics.<sup>42</sup> Unfortunately, the literature demonstrates that current Canadian NP human resource planning suffers from deficiencies and tends to be intermittent, based on incorrect information, and is often not linked to appropriate research evidence.<sup>43</sup> NP human resource efforts are not consistent across the country nor are efforts comprehensive enough to fully address sustainability issues of the NP role.<sup>2</sup> Additionally, there are areas in Canada's northern jurisdictions (Yukon, North West Territories and Nunavut) where NPs are grossly underutilized and underrepresented.<sup>2</sup>

#### **Factor 5: Innovation**

Aside from the NP-led clinic initiative in one Canadian jurisdiction (Ontario) there are no other examples of sustainable NP innovation in Canada. The key word being sustainable. Many other jurisdictions have tried to implement similar models; however, these models were set-up to fail through the use of "special envelope" funding that did not include planning for continuance of the funding past the original grant timeline. For example, a newly established NP-led academic faculty practice model was established in Alberta for seniors. The NP-led clinic was embedded within an existing social service agency, however, with their funding set to end March 2020, the government failed to identify funds for continuance of this innovation. The clinic closure means that many seniors are left with no access to primary care. This lack of access is particularly dangerous in our current Covid19 health crisis. NP roles and associated models of care continue to be organized on the basis of policy tradition, power and politics rather than in relation to the evidence of how best to meet current health needs. Furthermore, there is no consideration given to whether NPs should be integrated into existing service delivery models or whether they are better suited to their own innovative models of care.<sup>2</sup>

#### Discussion

This critical analysis of current Canadian NP compensation models highlights the following findings: 1) jurisdictional government funding is the standard remuneration mechanism for NPs across Canada. This type of funding is rigid, inflexible, often not based on evidence informed policy making and also perpetuates entrenched policies that continue to inhibit solutions towards addressing sustainability of the NP role, 2) considerable jurisdictional variation exists in NP funding models, including variability of jurisdictional NP initiatives and health policies. This is most obvious in the disbursement of "special envelope" funding for pilot projects or grant type NP funding initiatives across the country. These initiatives have occurred in a piecemealed type fashion, are short term and generally not sustained, 3) current funding models as well as NP compensation (wages and benefits) are not keeping up with market value, 4) organizations with hierarchical infrastructure that support physician-led teams and have total decision making authority, negatively impact NP role utilization, 5) there is a lack of transparency in NP workforce strategy, and 6) there is a lack of innovative, interprofessional models such as the NP led Clinic across the nation.<sup>2</sup>

#### **Future Directions**

#### **Funding Model Reform**

This research suggests that Canadian jurisdictions have implemented a tapestry of funding models and no two provinces are the same in their approach to the integration of NP roles within an antiquated health care system. This analysis also demonstrates that jurisdictional government funding is the prevailing remuneration mechanism for NPs across Canada. This type of funding is rigid, inflexible and frequently not evidenced based. Furthermore, global health budgets provide little incentive for innovation or to improve efficiency of care. Jurisdictions must transition to evidence based, flexible funding models that support PHC environments and variations in NP practice. These funding models must include comprehensive team-based funding where appropriate, as well as increase the number of NP-Led Clinics across the country.44-47

Funding model reform, regardless of setting, should ensure that the money follows the patient, not the provider, and improvements should aim to better support patient-centred care.<sup>2</sup> A mandate to protect funding must focus on stability, sustainability and the potential for long-term evaluation, once funding models have been incorporated into a health care organization or structure.<sup>3,8</sup> There is an urgent need to amend and align Canadian national and jurisdictional health policies to facilitate transformation of NP funding models and considerable opportunity exists to address entrenched and outdated policy directives which continue to impact the optimization and sustainability of the NP role.<sup>2,8</sup>

There is an equal need to enhance and harmonize NPs' salaries and benefit compensation across all health care settings, within each jurisdiction, and to increase the representation of NPs in currently underserved areas.<sup>2, 48</sup> In determining what constitutes appropriate wage compensation, utilizing best research evidence, accounting for NPs' formal education and experience, their scope of practice, professional responsibilities and their accountability as autonomous health care providers is important.<sup>2,48</sup>

#### **Interprofessional Team Expansion**

Increasing the number and expanding the role of NPs on interprofessional teams, facilitates the adoption by policy-makers and funders, of multiple, innovative funding models. The influence of existing policies on the success of collaborative practice models needs consideration. Reimbursement structures for NPs have to ensure financial viability of NPs in PHC to increase the motivation for physicians to work in collaboration. The inclusion of NPs on interprofessional teams has the potential to remove some of the current funding barriers, to promote sustainability for the NP role and fosters the creation of positive relationships between NPs, their physician colleagues and other allied health professionals.<sup>49,50</sup> In this scenario, funding models are readjusted to ensure fair compensation across professions and interprofessional teams.<sup>8</sup> This requires integrated consultation where interprofessional teams negotiate compensation together to establish what model of care needs to be implemented, and what funding model(s) will achieve fair compensation for all parties.<sup>8</sup>

#### **Comprehensive Workforce Strategy**

Implementing a comprehensive health human resource strategy to optimize the use of Canadian NPs is required. Health human resource planning has traditionally focused on individual health care professional groups rather than taking an integrated, interprofessional approach.<sup>43</sup> Much of this planning has been dominated by supply-oriented thinking and outdated patterns of use. Planning of this nature results in the development of a health human resource allocation that has little relation to actual community needs. Interprofessional health human resource planning needs to be more closely linked to population health outcomes if it is going to produce a desired impact. Linked to this issue is the need for transparent, valid, reliable and current HR data to inform the requirements of a human resource strategy.<sup>43</sup>

It is important to recognize that no single funding model will address all NP funding model barriers and sustainability of the role in Canada's health system. Instead, it is the thoughtful design and integration of a variety of funding models, as well as a modernized, pluralistic and flexible approach that will foster a culture of change, innovation, and hold hope for a sustainable future for NPs.<sup>51</sup> To facilitate formation of these models, and increase the accessibility of quality health care, the nursing community, other health professions groups, and policymakers must establish common ground to remove barriers to NP practice, improve NP wage and benefit compensation and increase interprofessional collaboration.

#### **Implications for Research**

Cross jurisdictional variability in funding models points to the need for research to further analyze and examine these exact trends across Canada.<sup>24</sup> Results from these types of studies could better inform policymakers about best practices with respect to funding models, and this may lead to more desirable NP role sustainability.<sup>23</sup> Research related to NP funding models in all settings is required to show the impact of these models on sustainability of NP practice and should be a priority area for investigation. Research also needs to examine interprofessional, integrated funding models and the effect these models have on patient access,

patient outcomes and quality of care.52

The information collected and reviewed for the purpose of this critical analysis was challenging to locate due to the significant contrast in how jurisdictions collect and report NP workforce data. The collection and reporting of NP this data is often included with that of other nursing professions and it is challenging to find and separate.<sup>41</sup> Further research on NP funding models relies on current, detailed, transparent workforce data and describes, monitors and compares the practice patterns, trends and funding information of NPs working in each jurisdiction across the country.

#### Conclusion

This critical analysis is intended to improve our understanding of current NP funding models in Canada. It analyzes the current state of NP funding models across Canada and proposes improvement of NP optimization by employing modern, flexible, interprofessional and integrative models to reflect community health needs and the broad variety of NP practice. It is important to conduct further evaluative research on current NP funding models and to begin to build a robust NP workforce strategy to help ensure sustainability of the NP role in Canada.

#### **Chapter Three: References**

- Canadian Nurses Association (2020). Nursing Statistics. <u>https://www.cna-aiic.ca/en/nursing-practice/the-practice-of-nursing/health-human-resources/nursing-statistics</u>
- Canadian Federation of Nurses Union (2018). Fulfilling nurse practitioners untapped potential in Canada's healthcare system. <u>https://nursesunions.ca/wp-</u> content/uploads/2018/06/CFNU\_UntappedPotential-Final-EN.pdf
- Carter N, Martin-Misener R, Kilpatrick K, et al. The role of nursing leadership in integrating clinical nurse specialists and nurse practitioners in healthcare delivery in *Canada. Nurs Leadersh* (Tor Ont). 2010;23 Spec No 2010:167-85. https://doi.org/10.12927/cjnl.2010.22274
- 4. Fox A, Gardner G, Osborne S. A theoretical framework to support research of health service innovation. *Aust Health Rev.* 2015;39(1):70-75. doi:10.1071/ah14031
- 5. Canadian Nurses Association. (2003). Health Human Resource Component, Literature Review Report, Recruitment and Retention of Primary Health Care Nurse Practitioners in Canada. https://cna-aiic.ca/~/media/cna/page-content/pdfen/04 hhr appendixc.pdf?la=en
- 6. Canadian Nurses Association (2016, August). The Canadian Nurse Practitioner Initiative: A10-year Retrospective. <u>https://cna-aiic.ca/~/media/cna/page-content/pdf-en/canadian-nurse-practitioner-initiative-a-10-year-retrospective.pdf?la=en</u>

- Dicenso A, Auffrey L, Bryant-Lukosius D, et al. Primary health care nurse practitioners in Canada. *Contemporary Nurse*. 2007;26(1):104-115. doi:10.5172/conu.2007.26.1.104
- Dicenso A, Bourgeault I, Abelson J, et al. Utilization of Nurse Practitioners to Increase Patient Access to Primary Healthcare in Canada – Thinking Outside the Box. *Nursing Leadership.* 2010;23(sp):239-259. doi:10.12927/cjnl.2010.22281
- Donald F, Martin-Misener R, Bryant-Lukosius D, et al. The Primary Healthcare Nurse Practitioner Role in Canada. *Nursing Leadership*. 2010;23(sp):88-113. doi:10.12927/cjnl.2013.22271
- Edwards N, Rowan M, Marck P, Grinspun D. Understanding Whole Systems Change in Health Care: The Case of Nurse Practitioners in Canada. *Policy, Politics, & Nursing Practice*. 2011;12(1):4-17. doi:10.1177/1527154411403816
- Gould ON, Johnstone D, Wasylkiw L. Nurse practitioners in Canada: Beginnings, benefits, and barriers. *Journal of the American Academy of Nurse Practitioners*. 2007;19(4):165-171. doi:10.1111/j.1745-7599.2007.00210.x
- 12. De Witt L, Ploeg J. Critical analysis of the evolution of a Canadian nurse practitioner role. *Canadian Journal of Nursing Research Archive*. 2005;1:37(4).
- Wong ST, Farrally V. The utilization of nurse practitioners and physician assistants: A research synthesis. Michael Smith Foundation for Health Research; 2014.
- 14. Green LV, Savin S, Lu Y. Primary care physician shortages could be eliminated through use of teams, nonphysicians, and electronic communication. *Health Affairs*. 2013; 1;32(1):11-9.

- Geurts-Laurant MG, Reeves D, Hermens RP, Braspenning JC, Grol RP, Sibbald BS. Substitution of doctors by nurses in primary care. *Cochrane Database of Systematic Reviews*. 2004;4:CD001271.
- 16. Mundinger MO, Kane RL, Lenz ER, Totten AM, Tsai WY, Cleary PD, Friedewald WT, Siu AL, Shelanski ML. Primary care outcomes in patients treated by nurse practitioners or physicians: a randomized trial. *Jama*. 2000 Jan 5;283(1):59-68.
- Peterson LE, Phillips RL, Puffer JC, Bazemore A, Petterson S. Most family physicians work routinely with nurse practitioners, physician assistants, or certified nurse midwives. *J Am Board Fam Med* 2013; 26:244 –245.
- Reay T, Patterson EM, Halma L, Steed W. Introducing a nurse practitioner: experiences in a rural Alberta family practice clinic. *Can J Rural Med.* 2006;11(2):101-7.
- 19. Nurse Practitioner Association of Canada. (2019, May 24 email). President's Newsletter.
- Canadian Institute for Health Information. *Health Workforce in Canada, 2019 Quick Stats*. Ottawa, ON: CIHI; 2020.
- 21. DiCenso A, Bryant-Lukosius D. Clinical nurse specialists and nurse practitioners in Canada. *Canadian Health Services Research Foundation*. 2010 Jun.
- 22. Foster W, Kim P, Christenson D. Ten Non Profit Funding Models. Stanford Social Innovation Review. 2009: https://ssir.org/articles/entry/ten\_nonprofit\_funding\_models#
- 23. Canadian Nurses Association (2009b). Recommendations of the Canadian Nurse Practitioner Initiative: A Progress Report.
- 24. Delvin ME, Braithwaite S, Plazas PC. Canadian nurse practitioner's quest for identity: A philosophical perspective. *International journal of nursing sciences*. 2018: 10;5(2):110-4.

- 25. Dueck, S. Timely Care Care Clinic Now Serving Steinbach Patients. 2017 https://steinbachonline.com/local/timely-care-clinic-now-serving-steinbach-patients
- 26. Murphy GT, Birch S, Mackenzie A, Rigby J, Langley J. An integrated needs-based approach to health service and health workforce planning: applications for pandemic influenza. *Healthcare Policy*. 2017:13(1):28.
- 27. Divisions of Family Practice. (n.d.). Contracting 200 NPs into Primary Care Networks. https://www.divisionsbc.ca/sites/default/files/inlinefiles/NP%20Backgrounder%20and%20FAQ%20-%20Final%20181023.pdf
- 28. Sangster-Gormley E, Martin-Misener R, Burge F. A case study of nurse practitioner role implementation in primary care: what happens when new roles are introduced? *BMC Nursing*. 2013;12(1). doi:10.1186/1472-6955-12-1
- 29. Government of Alberta. PCN Nurse Practitioner Support Program. 2019: https://www.alberta.ca/assets/documents/health-pcn-nurse-practitioner-support-programguide.pdf
- 30. Martin-Misener R, Donald F, Kilpatrick K, Bryant-Lukosius D, Rayner J, Landry V, Viscardi V, McKinlay R. Benchmarking for nurse practitioner patient panel size and comparative analysis of nurse practitioner pay scales: update of a scoping review. Hamilton, ON: McMaster University. 2015 Mar.

https://fhs.mcmaster.ca/ccapnr/documents/ScopingReviewNPPanelSize-FINALPrinted17042015SJ 000.pdf

31. Heale R, Butcher M. Canada's First Nurse Practitioner-Led Clinic: A Case Study in Healthcare Innovation. *Nursing leadership* (Toronto, Ont.). 2010;23(3):21-9.

- 32. King, R., Tod, A., & Sanders, T. Regulation of advanced nurse practitioners: understanding the current processes in the UK. *Nurs Stand*, 2017: 32(14), 43-50.
- Buppert, C. Billing for nurse practitioner services: Guidelines for NPs, physicians, employers, and insurers. *Medscape Nurses*, 2002: 4(1).
- 34. Delamaire, M. L., & Lafortune, G. Nurses in advanced roles: a description and evaluation of experiences in 12 developed countries. Organization for Economic Cooperation and Development, 2010.
- 35. Kleinpell R, Scanlon A, Hibbert D, et al. Addressing Issues Impacting Advanced Nursing Practice Worldwide. Online J Issues Nurs. 2014;19(2):5. DOI: 10.3912/OJIN.Vol19No02Man05
- 36. Nixon SA, Yeung E, Shaw JA, Kuper A, Gibson BE. Seven-Step Framework for Critical Analysis and Its Application in the Field of Physical Therapy. Phys Ther. 2017;97(2):249-257. DOI:10.2522/ptj.20160149
- 37. Greenhalgh T, Robert G, Bate P, Kyriakidou O, Macfarlane F, Peacock R. How to spread good ideas - A systematic review of the literature on diffusion, dissemination and sustainability of innovations in health service delivery and organisation. *Report for the National Co-ordinating centre for NHS Service Delivery and Organisation R & D* (NCCSDO), 1-426
- Chambers DA, Glasgow RE, Stange KC. The dynamic sustainability framework: addressing the paradox of sustainment amid ongoing change. *Implementation Science*. 2013;8(1). doi:10.1186/1748-5908-8-117

- Norful AA, De jacq K, Carlino R, Poghosyan L. Nurse Practitioner-Physician Comanagement: A Theoretical Model to Alleviate Primary Care Strain. *Ann Fam Med*. 2018;16(3):250-256. doi: 10.1370/afm.2230.
- Poghosyan L, Liu J, Norful AA. Nurse practitioners as primary care providers with their own patient panels and organizational structures: A cross-sectional study. *Int J Nurs Stud.* 2017;74:1-7. doi: 10.1016/j.ijnurstu.2017.05.004
- 41. HayGroup. (2013). Final Report Submitted to: AOHC, AFHTO and NPAO: Developing a Provincial Compensation Structure for Primary Care Organizations. https://www.afhto.ca/wp-content/uploads/Developing-a-Provincial-Compensation-Structure-for-IPCOs-Final-Report-June-2013.pdf
- 42. Canadian Institute of Health Information. (2018). Nursing in Canada. https://www.cihi.ca/en/nursing-in-canada-2018
- 43. Canadian Nurses Association. (n.d.). Health human resource component literature review report. Canadian Nurse Practitioner Initiative. https://www.cna-aiic.ca/-/media/cna/pagecontent/pdf-

en/04\_hhr\_appendixc.pdf?la=en&hash=A824BB5E65D3BB8CE9456F1E6FBC72AD97 CB6971

- 44. Hutchison B. A long time coming: primary healthcare renewal in Canada. *Health Pap*. 2008;8(2):10-24. DOI: 10.12927/hcpap.2008.19704
- 45. Hutchison B, Abelson J, Lavis J. Primary care in Canada: so much innovation, so little change. *Health Aff* (Millwood). 2001;20(3):116-31. DOI:10.1377/hlthaff.20.3.116
- 46. Hutchison B, Levesque JF, Strumpf E, Coyle N. Primary health care in Canada: systems in motion. *Milbank Q.* 2011;89(2):256-88. doi: <u>10.1111/j.1468-0009.2011.00628.x</u>

- 47. El-Jardali, F & Lavis, J. (2011). Issue Brief: Addressing the Integration of Nurse Practitioners in Primary Healthcare Settings in Canada. *McMaster Health Forum*. https://www.mcmasterforum.org/docs/default-source/product-documents/issuebriefs/nurse-practitioners-in-primary-healthcare-in-canada-ib.pdf?sfvrsn=2
- 48. Martin-Misener R, Donald F, Kilpatrick K, Bryant-Lukosius D, Rayner J, Landry V, Viscardi V, McKinlay R. Benchmarking for nurse practitioner patient panel size and comparative analysis of nurse practitioner pay scales: update of a scoping review. Hamilton, ON: McMaster University. 2015 Mar.
- 49. Johnson JE. Working together in the best interest of patients. J Am Board Fam Med.2013;26(3):241-3. doi: 10.3122/jabfm.2013.03.130075..
- Naylor MD, Kurtzman ET. The role of nurse practitioners in reinventing primary care. Health Aff (Millwood). 2010;29(5):893-9. doi: 10.1377/hlthaff.2010.0440.
- 51. Aggarwal M, Hutchison BG. Toward a primary care strategy for Canada. Ottawa, Ontario: Canadian Foundation for Healthcare Improvement; 2013 Jan.
- 52. Harkless G, Vece L. Systematic review addressing nurse practitioner reimbursement policy: Part one of a four-part series on critical topics identified by the 2015 nurse practitioner research agenda. *J Am Assoc Nurse Pract*. 2018;30(12):673-682. doi:10.1097/JXX.00000000000121

## Figure 3-1: Sustainability of Innovations Framework<sup>4</sup>





## Figure 3-2: Image Representation of NP Funding Models in Canada

## 85

## Table 3-1: Explanation of Funding Models for Canadian Nurse Practitioners

Explanation of Funding Models for Canadian NPs	Jurisdiction Utilizing
	Approach
Government Funded:	BC, AB, SK, MN,
Most NPs across Canada are employed in a hospital, health region or health care agency environment within an	ON, QC, NFLD,
organization run by the government. The majority of these positions are salaried.	NB, NS, PEI, NWT,
	YT,
Capitation:	AB
Allocates funding for health services based on population demographics. A fixed payment made at regular intervals	
by the government for each enrolled patient, regardless of services provided. More recent NP integration strategies	
have involved the payment of the NP's salary by the government to work in primary healthcare practices where	
physicians are paid through mechanisms other than FFS, such as capitation.	
Fee For Service:	None
This is a payment model in which NPs charge separately for each service they perform. They direct bill these	
services to the provincial health insurance plan. There is no fee for service NPs in Canada	
Shadow Bill:	BC, AB, SK, MN,
Refers to the process where non-fee-for-service health practitioners such as NPs submit	PEI
claims to Provincial Health Care for insured services provided to patients. Claims are paid at zero and are for	

tracking purposes only.	
Private Pay Option:	AB, MN, NFLD
The client has access to any healthcare service that is provided by the NP, but will utilize their personal funds or	
insurance plans to pay for services. Services can be customized to fit the needs of the client.	
Non Profit/ Private Corporation: a private or community clinic where NPs are employees of the clinic	AB, NFLD, YT,
	NWT
Interprofessional NP Led Clinic: This model is the incorporation of nursing leadership within an interprofessional	ON
team and is a formal clinic where the NP independently and collaboratively provides primary care.	
Academic Institution Funding:	AB

Table 3-2: Canadian Jurisdiction Dominant NP	and Other Funding Model; Median NP Salary
--	---

Jurisdictio n	DOMINANT NP FUNDING MODEL	OTHER COMPENSATION MODELS	MEDIAN NP SALARIES
British Columbia	<ul> <li>Government funding for full or partial NP salary, benefits, overhead and administrative costs and professional development expenses.</li> <li>Majority of B.C. NPs are paid to work in community based/primary health care settings working with specialized populations.</li> </ul>	<ul> <li>2012: NP4BC, 45 new NP positions per year over three years were made available (2012/13 – 2014/15). Integration of NP roles was introduced to support increased access to primary health care services. Salaried model where HA receives partial funding for NP positions (salary, benefits only. HA global budget must cover overhead/admin., professional development)</li> <li>2019: 200 new NP positions as part of a shift to a team-based primary health-care system. Positions are being supported by \$115 million over 3 yrs, to secure NPs' employment in primary care settings throughout B.C.</li> </ul>	\$105, 383
Alberta	• Government funding for full or partial funding of NP salary which typically	• 2019 PCN NP Support Program. Government of Alberta (GOA) announced 50 funded NP	\$112,000

Jurisdictio n	DOMINANT NP FUNDING MODEL	OTHER COMPENSATION MODELS	MEDIAN NP SALARIES
	<ul> <li>includes salary, benefits, overhead and administrative costs and professional development expenses.</li> <li>Majority of Alberta NPs are paid to work in acute care settings.</li> </ul>	positions through PCNs. Interested PCN's must apply for funding (\$125,000 for 1 FTE) via an Expression of Intent form which must be approved by GOA.	
Saskatche wan	<ul> <li>Government funds most RN(NP) positions throughout the province.</li> <li>Government funding for full or partial funding of NP salary which typically includes salary, benefits, overhead and administrative costs and professional development expenses.</li> </ul>	• There are a few RN(NP) positions that are receive payment for their services paid for by from physician specialists, who employee incorporates or work with RN(NP)s in their practice setting.	\$114,200
Manitoba	• Government funds most of the NPs employed in either autonomous or collaborative teams' positions (which are run by health authorities or attached to private physician clinics).	No Additional Information	\$110,000

Jurisdictio n	DOMINANT NP FUNDING MODEL	OTHER COMPENSATION MODELS	MEDIAN NP SALARIES
Ontario	<ul> <li>NPs in Ontario are in salaried positions funded by the Ministry of Health and Long-Term Care (MOHLTC). 100 NP positions in long-term care, approximately 500 positions in FHTs, 425 in Community Health Centres, 117 in Community Care Access Centres and 145 in other community settings</li> <li>NPs in acute care are funded from a global hospital budget.</li> </ul>	<ul> <li>NP Led Clinics, an innovative model for delivery of comprehensive primary health care services. Funding is provided through the MOHLTC and NPs are the lead primary care providers of the clinic's inter-professional team of healthcare professionals. There are currently 27 NP Led Clinics in the province.</li> <li>Some NPs are employed directly by physicians in their solo practices and are paid from physician salary or FFS revenue.</li> </ul>	\$103, 822
Quebec	• Government funds NP services in primary care in public organizations, hospital-based family medicine units and family medicine groups (GMFs).	One extended pilot project Super Nurse Clinic. Led by NP and extended funding by RHA in 2016	\$85,000
Newfoundl and	• Government funding for full or partial NP salary, benefits, overhead and	• There are private corporations who employ a small number of NPs.	\$101,622

Jurisdictio n	DOMINANT NP FUNDING MODEL	OTHER COMPENSATION MODELS	MEDIAN NP SALARIES
New Brunswick	<ul> <li>administrative cost and professional development expense</li> <li>Government funds NPs and they are employed by one of two RHAs in a variety of community and clinic settings.</li> </ul>	<ul> <li>One NP in the province is funded by a private physician's office a</li> <li>Provincial Department of Justice employs two NPs</li> <li>In 2015, the Department of Health announced a pilot project whereby a physician could hire an NP and be reimbursed for each client/NP visit at half the usual rate.</li> </ul>	\$100,000
Prince Edward Island	• The dominant funding mechanism is the provincial government who hires NPs and they are salaried according to the provincial nurses' union collective agreement.	<ul> <li>One NP has been hired through private funding to work in a clinic operated through a non-profit Family Centre.</li> <li>There are two NPs working in a FFS practice who are hired privately by the practice physicians. These NPs are unable to bill the provincial health plan directly and the physician bills for the work the NP does.</li> </ul>	\$96,500

Jurisdictio n	DOMINANT NP FUNDING MODEL	OTHER COMPENSATION MODELS	MEDIAN NP SALARIES
Nova	• The government funds majority of	• While NPs can be employed privately, their FFS	\$103,000
Scotia	NPs	physician employer must be an active participant	
needs		in the care of the patient.	
update		• 2017: 23 new NP positions announced at	
		collaborative health clinics across the province	
		• 2019: New Nursing Act passed NPs and the	
		requirement to have a documented collaborative	
		practice for NPs was removed (NP formerly must	
		have a collaborative relationship agreement with	
		physician in order to practice).	
Yukon		• Only 3 practicing NPs. Two NPs work in	n/a
		collaboration with medical clinics. One NP is	
		contracted in a three-year pilot project through	
		the Yukon government	
North	• 48 salaried positions through the NWT	• one NP position for a private mining corporation.	\$106,000
West	government.		
Territories	• The salaried positions are distributed		
	throughout family clinics, hospitals and		
	health centres.		

Jurisdictio n	DOMINANT NP FUNDING MODEL	OTHER COMPENSATION MODELS	MEDIAN NP SALARIES
Nunavut	• No available information	• No available information	\$126,000

## Chapter 4: Sustainability Failures in Primary Health Care: The Case of an NP Clinic Closure (Paper 3)

Raelene Marceau Dr. Stephanie Montesanti Dr. Tammy O' Rourke Dr. Kathleen F. Hunter

Marceau, R., Montesanti, S., O'Rourke, T. and Hunter, K., 2020. Sustainability Failures in Primary Health Care: The Case of an NP Clinic Closure. Manuscript in Preparation, to be submitted to *Social Science & Medicine* 

•

•

#### Sustainability Failures in Primary Health Care: The Case of an NP Clinic Closure

Canada's health care crisis demands changes to the health system that are patient-centred, cost effective; changes that promote health system stability, and long-term sustainability.<sup>1</sup> Canada can no longer afford ongoing increases to the cost of health care. This financial burden is influenced by persistent inadequate access to all health care services, increasing aging populations, and high incidences of chronic diseases; all which tax an already congested health system. The Canadian government is well aware of the need to address these long-term challenges, as well as the critical importance of introducing innovative, effective and sustainable solutions to save our health system.<sup>2</sup> Sustainability is frequently mentioned as a critical concept in health system reform, yet little is known about how to sustain innovations to improve this system.<sup>3</sup> While there is a growing body of sustainability research, few studies have considered the challenges involved in sustaining innovations and the factors that influence and ensure sustainability of innovations within primary health care (PHC).<sup>4</sup>

One of the most cited sustainability failures in Canada's health system is the Nurse Practitioner (NP) role, which was first introduced in the 1970's and re-implemented in the 1990's.<sup>5,6</sup> Despite indications of NP effectiveness in Canada's health system, barriers towards full implementation of the role and its sustainability remain at risk. These barriers and risks are particularly visible in primary care settings, which limit NPs ability to take on the most responsible care provider role, greatly impacting patient access to health care and NP role sustainability. <sup>6-10, 12</sup> Using a single, exploratory case study, this research examines the closure of a NP clinic in western Canada and seeks to understand the contextual factors influencing the decision to close the NP clinic, which ultimately led to sustainability failure. The case study analysis was guided by a theoretical framework informed through an in-depth review of the literature examining the sustainability of innovations and by a policy framework which systematically considers the factors that may impact the policy environment, and is suitable for policy analysis.<sup>11,13</sup> The findings contribute to the limited literature on sustainability failures, may inform future innovation planning and assessment, and have the potential to enhance governments, funders and policy-makers understanding of the elements that contribute to NP role sustainability and sustainability of other important PHC innovations.

The use of the NP role has fluctuated throughout the past forty years with periods of substantial growth and periods of relative inactivity. The role has been impacted by health care system restructuring, increasing costs of health care, health human resource shortages, and a demand for health spending accountability.<sup>15</sup> All ten provinces and three territories have legislation and regulations set out to support the role of NPs and NPs provide health services in a variety of settings across Canada.<sup>16-21</sup> Currently, over 6,000 NPs practice in Canada and it is estimated that the number of NPs have increased by 300% since 2007.<sup>16,22,23</sup> Despite the increase in numbers and growing evidence recognizing the value that NPs contribute to quality of care, barriers continue to exist towards full optimization of the role. Most individuals in Canada don't have access to, or the option of NP care. This is reflected by a ratio of approximately 14 NPs per 100,000 population and illustrates that the number of NPs is insufficient to meet the growing needs of Canada's population.<sup>11,24</sup>

#### Background

Despite longstanding and ongoing recommendations for PHC reform, accessibility issues and a lack of comprehensive health service delivery continue to contribute to Canada's absence in fulfilling a vision for PHC integration. <sup>2,16,19,25</sup> The optimization of health care professional scopes of practice and innovative models of care have the potential to initiate transformation and contribute to a health system to better meet patient, community and population needs. However, current institutionalization of scopes of practice has shielded the system from radical reform and there is limited incremental change across Canada in the expansion of scopes of practice for health care professionals, such as NPs. NPs possess an extensive body of knowledge and skills, and they have advanced education, such as a Master's or Doctoral degrees.<sup>16,17</sup> Within their scope of practice, NPs diagnose acute and chronic illness, prescribe medications and are accountable and responsible for their own professional practice.<sup>16</sup>

#### Sustaining Progress: NP access for all Canadian's

The current Canadian health reform agenda prioritizes the provision and sustainability of cohesive, comprehensive PHC services that are responsive to community and patient needs.<sup>26,27</sup> As PHC health services are usually the first point of contact for patients with the health system, the loss of services contributes to diminished access to health care and poorer health outcomes.<sup>27</sup> The challenges associated with this agenda are many, and in spite of Canada's success to develop, plan and implement many PHC innovations, its ability to sustain, spread and scale up these innovations has been inadequate.<sup>27</sup> A glaring example of this failure to progress is demonstrated by Canada's limited success in implementation of the NP role.

Sustainability is an extensively debated topic and is challenging to define. The application of this concept is particularly difficult in the context of health systems. Sustainability is defined as the "continued use of program components and activities for the continual achievement of desirable program and population outcomes".<sup>28(p2060)</sup> Few studies have examined how best to sustain PHC innovations in Canada.<sup>28-30</sup> There is also a lack of knowledge about successful implementation of innovative models of care and the factors that influence their sustainability. <sup>29–33</sup> Stirman et al.<sup>29</sup> suggest that more research and evaluation is necessary to inform the sustainability of PHC innovations since early efforts to study the sustainability of health programs illustrates that the evidence base is 'fragmented and underdeveloped'. <sup>(p. 3)</sup> Similarly, there is a lack of scholarly debate and research on the factors affecting the capacity of health systems to evolve, transform, or adapt to change, which is necessary information for evaluating PHC innovations.<sup>4</sup> Therefore, more targeted research and evaluation is necessary to better inform knowledge on the sustainability of PHC innovations.<sup>29</sup>

This paper presents a qualitative case study of a NP clinic located in a southern Alberta community in Canada. It represents an in-depth examination of a sentinel innovative PHC model that was implemented to address primary care access issues at a time when the community experienced an increase in their population density and a subsequent shortage of physicians. The innovation was funded by the Calgary Rural Primary Care Network (CRPCN) in the community of Okotoks, Alberta in 2010. Despite strong community support and evidence of outcome attainment, the decision to close the clinic was made by the CRPCN in 2015. A description of the community context and the history of the clinic is described below.

#### The Case: The Sheep River Nurse Practitioner Clinic

The Sheep River NP Clinic in Okotoks, Alberta opened in October 2010 as a pilot project funded by the CRPCN; a member of the Alberta Primary Care Network. The network was an initiative created and funded by the provincial Government in 2003 as a health reform program to improve patient primary care access.<sup>33</sup> Okotoks is located 18 kms south of the city of Calgary, Alberta and its current population is 28,833 residents. Between 1996 and 2016 the population of Okotoks increased by 192.6% (compared to a 62.2% increase for Alberta) when it was considered one of the fastest growing communities in the country.<sup>34</sup>

The Sheep River NP clinic was introduced to address the physician shortage in Okotoks and increase access to primary care services; while ultimately reducing emergency department wait times.<sup>35–36</sup> The clinic was solely operated by NPs who independently served 1,600 patients and shared a physical space with two established family physicians. It was highly anticipated by funders and organizational leaders that this NP initiative would propel the provincial government towards producing a sustainable funding solution for NPs across the province.<sup>35</sup> Beyond the initial pilot funding period, there was no dedicated budget or definitive project end date.<sup>37</sup> Unfortunately, in October 2015, the decision was made by the CRPCN to close the Sheep River NP clinic in January 2016, despite vigorous community lobbying and support from the local Member of the Legislative Assembly (MLA).<sup>38</sup> The CRPCN sited the following as a reason for closure:

"As the PCN has matured as an organization, we have developed other initiatives and programs, many of which integrate interprofessional health teams directly into the offices of member physicians. These programs also provide critical linkages to programs and services that benefit patients throughout our rural communities. In preparing our 2015-2018 business plan, the Physician's Corporation and Board of Directors jointly determined that, in the absence of a stable alternative funding source, long term support for the nurse practitioner program would not be a financially feasible option" (Document 20, 2015)

"Currently the clinic operates at a cost of \$297,557.48 per annum, caring for a panel of approximately 1600 patients. Unfortunately, funding the program at this amount prevents
the local community from funding other programs and services that also benefit patients of PCN member physicians practicing in the area" (Document 20, 2015)

# **Research Objectives and Questions**

Case study allows for in-depth examination of sentinel events, such as a sustainability.<sup>39</sup> The primary purpose of this case study was to understand how the abrupt closure of the Sheep River NP clinic happened and why evidence did not influence sustainability of this pilot project. The knowledge gained from this study may have implications for the sustainability of the NP role in Canada, as well as other PHC innovations. Our research was guided by the following three research questions:

- Why did the Sheep River NP Clinic close and what were the contextual factors influencing clinic closure?
- What was the decision-making process that led to the clinic closure and who were the actors influencing this decision?
- What factors influenced the sustainability failure of the Sheep River NP clinic?

### **Theoretical and Policy Analysis Frameworks**

The focus in this case study was to apply a theoretical framework to support the concept of sustainability and utilize a policy analysis framework to advise health policy. The Policy Triangle Framework (PTF)<sup>13</sup> and the Sustainability of Innovation Framework (SIF)<sup>11</sup> were chosen to inform study development and were used to frame data analysis. Other frameworks, which focus on implementation and scale up and spread, may provide provide an alternative perspective on the sustainability or policy analysis process. However, the SIF and PTF are simple, applicable to PHC settings and it was felt they best support the research questions to examine how sustainability of PHC innovations can be improved. These frameworks will be described in this section.

### **Policy Triangle Framework**

The PTF encourages analysis of the social, economic and political factors, which have influenced a policy-making decision.<sup>13</sup> Walt and Gilson believe that health policy-making is an interactive process within special social-economic and cultural contexts where *actors* - individuals, groups and organizations - are at the center of this process. The PTF stresses the importance of the *process* by which the policy was initiated, formulated, developed, implemented and evaluated, including the objectives of the policy, and the 'actors' involved in the decision-making process.<sup>40</sup> This simplified framework and approach to policy analysis assists researchers to understand and analyze health-related policies systematically.<sup>40</sup> The framework consists of four elements, context (why), content (what), process (how) and actors (who). See Figure 4-1 for framework diagram and Table 4-1 for framework elements.<sup>41</sup>

### **Sustainability Innovation Framework**

The SIF merges the concepts presented by Greenhalgh and colleagues' systematic review of health service innovations,<sup>42</sup> and the Dynamic Sustainability Framework of Chambers, Glasgow and Stange.<sup>43</sup> This theoretical framework is suitable for the examination of sustainability of any health service innovation and identifies five factors that impact sustainability. These factors are: political, organizational, financial, workforce and innovation specific, and each factor has clear-cut questions and criteria to assess sustainability of an innovation in PHC. See Figure 4- 2 for framework diagram, Table 4- 2 for framework factor explanation.<sup>11</sup>

#### Methods

# Design

This study used an exploratory, single case study design and adhered to Yin's rigorous approach to case study research.<sup>39</sup> Documents and fieldnotes relevant to the research questions for this case study were also sub-units of analysis. The case study methodology was chosen because the aim was to analyze the results arising in this individual case to allow the discovery of richer, more detailed and useful information about the Sheep River NP Clinic closure. The

case study approach was also used to gain deep insights into this contemporary and complex issue within a real-life context.<sup>39</sup>

## **Data Collection**

Data were collected from key informant interviews and qualitative document analysis of publicly available and internal documents to promote an in-depth understanding of the phenomenon of interest, and to ensure analytical rigor.<sup>39,44</sup> Data collection occurred from May to June 2019. In-depth, semi structured interviews were conducted with 18 participants regarding their knowledge of the Sheep River NP clinic closure. The interview guide was developed a-priori based on the SIF and PTFs (Table 4-3). Interviews were tape recorded, transcribed verbatim and verified for accuracy of transcription, before being imported into the qualitative data analysis package MAXQDA to facilitate data handling. The research team met to organize and validate the findings, identify any gaps and key themes. Documents were gathered from provincial government documents, government/organizational policies, health organization annual reports, scholarly publications, media reports and opinion pieces (e.g. editorials in newspapers or letters to the editor). Researcher field notes were another data source used as the notes were considered 'analytical in themselves' because they contain 'immediate and later perceptions and thoughts' about the research.<sup>45(p3)</sup>

### **Participants**

We identified a purposive sample of informants using a sampling frame (Palinkas et al., 2015). This sampling frame included health care providers, government and organizational employees/leaders, community members and professional organization stakeholders that were especially knowledgeable about the Sheep River NP clinic closure. Emails with an attached study letter were sent to potential participants with publicly available emails inviting them to take part in the study (Study Letter Figure 4-3). In addition to purposeful sampling, snowball sampling was utilized. Participants identified through purposive sampling were asked during their interviews to identify other appropriate participants and if interviewed, these participants were asked to identify other potential participants.<sup>46</sup> This sampling process continued until no further participants were identified. Baker, Edwards, & Doidge,<sup>47</sup> suggested aiming for a sample

of around 15- 30 interviews to achieve saturation and this was achieved by obtaining a total 18 interviews.

#### **Data Analysis**

In keeping with Yin's case study approach, three analytical strategies were employed for this case study.<sup>39</sup> The first was pattern matching, the second was developing a case description using an explanation building strategy, and the third was utilizing a time series analysis. A deductive (using an a-priori pre-set coding scheme) and inductive (coding derived from the data) analysis approach were used as the SIF had not been previously tested. While the SIF guided our understanding of the factors that influence sustainability of PHC innovations, we applied the PTF to understand the role of actors and the social, political, economic and cultural contexts in which the decision to close the NP clinic was made.<sup>14</sup> An a-priori coding framework was developed for deductive analysis using categories from the SIF and PTF (Table 4-4). A multiphase approach was utilized for analysis. In phase one, data were identified from the interview transcripts and documents, pattern matching was utilized and data was coded using the previously developed apriori framework. The data was matched under the a-priori categories in this initial process and then a second phase of coding was inductively conducted using thematic analysis by Braun and Clark.<sup>48</sup> In this second phase, using an open coding process, interview transcripts and documents were coded to identify any additional relevant codes based on the three research questions. In phase three, the codes from phase one and two were merged to create an explanation building strategy to construct links amongst the data. We overlaid the data from phase 1 codes from the apriori analysis, with phase 2 codes and confirmed the fit of the categories to the a-priori coding framework. This merging process included the researcher collapsing data to categories, building initial themes, reviewing themes, and defining and naming the themes.<sup>48</sup> There were categories that did not fit the a-priori framework and these were collapsed into new and emerging themes. In the final phase, a timeline for the clinic closure was constructed from the interview transcripts and document data. The timeline plotted the fit of the themes against the events occurring in the case study to further support explanation building and a case description.

## Validity and Reliability

According to Yin, the quality of any case study, depends on construct validity, internal validity, external validity, and reliability.<sup>39</sup> Construct validity was achieved in this study through utilizing multiple sources of evidence, and maintaining a strong chain of evidence.<sup>49</sup> Internal validity was achieved through the use of pattern matching, explanation building and a timeline as an analysis strategy, utilizing field notes as a type of audit trail and using multiple sources of evidence.<sup>39</sup> External validity was achieved through utilizing a supportive conceptual and analytical framework to allow for data analysis that was thick and rich in description. Two strategies for ensuring reliability of this case study included: 1) utilizing a case study protocol, and 2) using the MAXQDA database to store all study interviews and documents.<sup>39</sup> Triangulation occurred in the form of multiple sources of data (interviews, documents and field notes), utilizing two coders during the analysis phase, and utilizing one theoretical and one policy analysis framework to support the development of interview questions and analysis of the data.

# **Ethical Considerations**

This study was conducted with full compliance of research ethics, codes and practices established by the University of Alberta, Human Research Ethics Board. This study involved human participants, commencing with face-to-face or telephonic interviews. Every research participant was given a letter of information outlining the purpose and procedures of the study, primary investigator role and contact information, funding sources, and information about how the results of the study would be disseminated and used. The letter also advised participants of their right to receive additional information and procedures for retracting information or withdrawing participation were also explained. Anonymity and confidentiality were explained and no individual participants would be identified in the final research report. Participation in the study was voluntary, and informed consent was discussed with all participants.

#### Results

The primary purpose of this study was to understand factors that led to the abrupt closure of the Sheep River NP Clinic and why sustainability of the clinic failed. During interviews with participants and through analysis of key documents and research field notes, ten key themes emerged. Most of these themes aligned with the previously described SIF and PTFs and the concepts they identified and defined. These themes included: Motivation<sup>2</sup> (*Context; PTF*), Details of Day to Day operations (*Process; PTF*), Just a Pilot Project (*Content/Process, PTF; Political, SIF*), Patient, Political and Provider Champions (*Actors, PTF; Political, SIF*), Stakeholder Uncertainties (*Political, SIF; Organizational, SIF*), Insufficient Consultation (*Organizational, SIF*) Funding Failure (*Financial, SIF*), The Right Model (Workforce; Innovation Specific, SIF), Innovation Barriers (*Innovation Specific, SIF*), and Profound Social Proof (*Innovation Specific, SIF*). See Table 4-5 for SIF and PTF aligned themes.

Three new themes not previously described in either the SIF or PTF emerged from the analysis: Emotional/Aftermath, Professional Turf and Technicolor Data. Participant and document characteristics, and a timeline presenting the chronological events relevant to the Sheep River Clinic inception and closure are discussed, twelve themes emerging from this research study are described. See Table 4-6 for new emerging themes.

# **Participant Characteristics**

Nineteen participants signed written consents and one participant withdrew their interview, for a total of eighteen interviews. <u>Table 4-7</u> presents an overview of participant characteristics. Participants (females n=14; males n=4) represented healthcare professionals, health organization leaders, professional organization leaders, and patients/community members.

## **Document Characteristics**

A total of 31 public documents were analyzed, these documents included; media articles (n=4), government reports (n=2), social media postings (n=9,) letters (n=4), a video (n=1), and other special reports (n=11). A summary of all the documents is provided in <u>Table 4-8</u>. Using the search terms Sheep River Nurse Practitioner Closure, documents were retrieved and published between 2010 and 2017. In addition, there were five documents with no date but they were deemed relevant to the closure of the Sheep River NP Clinic and thus included in the analysis.

 $<sup>^2</sup>$  Themes are presented as their constructed concept first followed in brackets by the framework in which they are aligned. Some themes align with both frameworks.

## **Case Timeline**

A timeline of significant events relevant to the introduction, implementation and closure of the Sheep Rive NP Clinic that occurred between 2009 and 2016 are reported in Figure 4-4. Analysis of this timeline took place by plotting the ten key and three emerging themes on the timeline along with significant events, decisions, and the Ministers of Health (MOHs) in office during the time the clinic was open. The planning for the Sheep River NP Clinic began in November of 2009 with the signing of a project charter agreement by the CRPCN. The clinic officially opened in October of 2010 and an evaluation conducted by the CRPCN was completed in 2012. The project charter agreement was renewed in March of 2013, with no indication of a review date for next renewal and no funding information. In the latter part of 2014, the decision was made by CRPCN to close the Sheep River NP clinic in January 2015, citing that funds were redirected to other priorities within their organization. However, in January of 2015 an order by the MOH was sent to the CRPCN advising them to keep the clinic open using surplus budget funds.

In February 2015, CRPCN responded to this Ministerial Order by agreeing to fund the Sheep River NP Clinic until April of 2016, hoping the Alberta Government would take the next 14 months to create a sustainable funding model for NPs across the province. During this time period the CRPCN organization went through significant changes, consolidating individual community authority, and creating a more centralized budget and decision-making process. In May of 2015 a provincial election saw a switch from the Progressive Conservative party to the New Democrat Party. This government changeover meant that a new Minister of Health was sworn in on May 15<sup>th</sup>, 2015. Lobbying took place by patients, stakeholders and Okotoks Member of Legislative Assembly (MLA). However, with no sustainable funding solution in sight, the NPs were informed by CRPCN in October of 2015 that the clinic would be closing ahead of schedule in January, instead of April 2016. On November 6<sup>th</sup>, 2016 a letter from the two NPs was sent out to patients informing them of the imminent clinic closure. On January 15<sup>th</sup>, 2016 the Sheep River NP Clinic closed its doors for good.

Analysis of the themes plotted on the timeline show that sustainability of the Sheep River NP clinic cannot be viewed as a static or linear state. Rather, sustainability in this case is better understood as a fluid and evolving process, one which remained dynamic and had varying degrees of progress and regression throughout the time the clinic was open.<sup>4</sup> The Sheep River NP clinic was a small innovative service in a health system constantly undergoing big changes and found itself susceptible to the decision making of key stakeholders. It is evident that without flexibility and responsiveness to wider pressures, the capacity of the Sheep River NP Clinic to adapt to the changing political, economic and health system conditions was lacking.<sup>4</sup> The Sheep River NP clinic closure validates the challenges of trying to sustain and embed an PHC innovation beyond initial implementation and substantiates the current evidence on the complexity of sustaining innovations in PHC. <sup>4</sup> The dynamic nature of sustainability makes it challenging for governments, funders, policy makers and health service planners to systematically address the sustainability of innovations in our health system. This results in economic loss, a lack of coordinated health reform effort and a lack of health policy reform.<sup>4</sup>

# **Conceptually Aligned Themes**

Ten of the thirteen final themes aligned with concepts originally identified and described in the SIF or PTF. This section presents each of these themes and provides evidence of participant quotes that support the label and definition for the theme.

## Motivation Clinic Opening and Closing (Context, PTF)

Motivation provides contextually relevant data as to why the clinic was developed, implemented and eventually closed. Documents provided and explanation and the primary motivation for the development and implementation of the Sheep River NP clinic was threefold; a perceived shortage of family physicians, lack of access to primary care and a huge population boom in Okotoks putting pressure on existing primary care services (Document 30, 2009; Document 19, 2012). Most participants (n=12) confirmed that the primary motivation for the development and implementation for the clinic was a perceived shortage of physicians.

"Yeah. Well, I don't know if this is the right answer, but my impression was ... I know for sure there was a doctor shortage, and they were looking to address that shortage by bringing in some nurse practitioners to fill in some deficiencies" (Participant 16; Patient) "[There was an] absolute shortage of primary care access within the community" (Participant 11; Health Care Professional)

The motivations for the closure of the Sheep River NP clinic was less well defined and participants spoke about many possible contextual reasons for the closure. Three primary motivations for the clinic closure were identified and several participants spoke about these motivations. First, 11 of the 18 participants spoke about the lack of a sustainable funding mechanism for NPs provincially. Second, participants (n=5) felt that NPs were being utilized as "physician gap fillers" in primary care, and when the perceived physician shortage improved, the clinic could be closed. Third, a provincial election and change in the core government party and philosophy as well as a Minister of Health who was in a new portfolio and had no previous government or health policy experience (n=8). The CRPCN NP Family Clinic project charter (Document 30, 2009) and evaluation (Document 19, 2012) for the Sheep River NP Clinic validated the shortage of physicians in primary care. Only 2 of the 18 participants said that they did not know why the clinic closed. Participants talked about five other possible reasons for the clinic closure, these motivations are summarized in <u>Table 4-9</u>. These motivations could not be confirmed in discussions with other participants and documents provided no information on these possible motivations.

# Just a Pilot Project (Process, PTF)

The Sheep River NP Clinic was referred to as a 'Pilot Project' by 12 of the 18 participants. Documents confirmed the original CRPCN budget for the clinic was designated to fund a one-year pilot initiative (Document 30, 2009). Three primary purposes of this pilot project were defined in the original charter for the development of the clinic. This pilot project was meant to test whether the model could: 1) provide care attachment, improve access and continuity of care for the residents of Okotoks who were in need of a primary care provider; 2) establish NPs as the first point of contact for a defined panel of patients; and 3) demonstrate the effectiveness, efficiency of NP care in a community of practice, where each health care professional shares information to provide the best possible care in the community. Twelve of the eighteen participants (n=12) understood that the clinic was a one-year pilot project, four participants (n=4) didn't know the nature of the project and two participants (n=2) stated that it was not a pilot project. The pilot project status of the clinic was confirmed through the document analysis which showed that the funding provisions were set from September 11, 2010 when the charter was signed and for a one-year time period with a community review to be held in March of 2011 (Document 30, 2009).

"[The clinic was] initially opened as a pilot project to fulfill a need: 20% of people in Okotoks without a family physician" (Participant 2; Health Care Professional)

"The nurse practitioner program was developed as a pilot program, aimed at enhancing access to a primary care provider in the Town of Okotoks during a period of physician shortage" (Document 30).

"It was never a pilot project. It was a PCN supported initiative. It only started to become viewed as temporary once there were more physicians in Okotoks" (Participant 6; Health Organization Leader)

Alignment of innovations with policy is an important feature of sustainability and only two participants (n=2) were able to affirm that the implementation of the pilot project aligned with the province's policy priority for primary care networks (PCNs).

"Well, I would say that the mandate of that clinic and the service they provided were completely aligned with the directions in Alberta health at the time" (Participant 1; Health Care Professional)

These policies included increased access for patients to a primary care provider, increased health promotion, disease and injury prevention, improved coordination of primary care, and promoting a multidisciplinary team environment (Document 30, 2009; Document 19, 2012; Document 20, 2015). An important negative finding from this study, is that the majority of participants (n=14) stated they 'didn't know' about policy alignment or had limited knowledge in regards to whether there were organizational policies to support the planning and implementation of the pilot project.

"No, I don't know whether policies were aligned" (Participant 17, Patient)

"I do not know that [if policies were aligned], but I don't think so" (Participant 5, Patient) "I don't know whether policies aligned" (Participant 3; Professional Organization Leader Provider)

# Details of Day to Day Operations (Content, PTF)

Participants were asked questions about the basic organizational structure, funding and relationships of the Sheep River NP Clinic to the Sheep River Medical Clinic. The Sheep River NP clinic staff included two NPs and one Medical administrative assistant (Document 19, 2012) A patient participant confirmed this staffing model.

"I only remember the one NP... [and] there was a person at the front office. It is just like a regular doctor's office. You go in, you check in and she comes and sees you and then you leave." (Participant 17; Patient)

The funding for the clinic came solely from the CRPCN budget and there were no documents or interview data that showed clear documentation of funding beyond the one-year pilot project or a long-term funding agreement.

"Yes, it [Sheep River NP Clinic] was 100% PCN funded. We were contract workers. We put in our hours. Every month, we got paid once a month through the PCN budget." (Participant 1; Professional Organization Leader)

"The executive director of the CRPCN administered and managed all staffing and budgetary responsibilities "(Participant 2; Health Care Professional)

## Patient, Political and Provider Champions (Actors PTF, Political, SIF)

Champions advocated for the implementation and sustainability of Sheep River NP Clinic. Champions in this case included patients, political figures, and health care providers. Documents identified patients who wrote letters to the Minister of Health (MOH) (Document 16, 2015; Document 17, 2015; Document 18, 2016) and participated in newspaper interviews to try to prevent the clinic from closing (Document 1, 2011; Document 2, 2015; Document 3, 2015). The local Member of the Legislative Assembly (MLA) was a champion for the clinic and presented the issue of the clinic closure during a legislative assembly question period (Document 7, 2018). The Nurse Practitioner Association of Alberta actively discussed the pending closure of the clinic on social media and directed their members to write to the Minister of Health (MOH) to express provider concerns over the closure (Document 13, 2015).

Participants also identified champions who supported the implementation and struggled to sustain the Sheep River NP Clinic; CRPCN organizational leaders, two local physicians colocated at the same physical space, two NPs that were employees of the clinic, the provincial NP association, the provincial nursing regulator, as well as numerous patients/community members.<sup>51</sup>

"our local MLA[member of provincial parliament], did some media stuff on it [saving the Sheep River NP Clinic], there were patients sending tons of letters and calling the Minister's office" (Participant 2; Health Care Professional)

"...it was certainly important to me, I know there's many NPs that don't feel like the provincial nursing regulator did enough to profile the NPs, but I can tell you as I ended up doing more advocacy for the Nurse Practitioner role than... Not more but certainly as much as for the 470 NPs in the Province and the 37,800 RNs, and I advocated for both roles. But I advocated just as strongly for NPs role as I did for the RN role." (Participant 1; Professional Organization Leader)

#### Stakeholder Uncertainties (Political, SIF; Organizational, SIF)

Participants were open about their uncertainties pertaining to the decision-making process to close the Sheep River NP Clinic. These uncertainties were related to both political factors (accountability) and organizational factors (transparency). Participants (n=12) talked about their perceptions pertaining to a lack of accountability by government. For example, one community member in a newspaper interview discussed how the change in government resulted in a lack of interest to sustain this NP clinic pilot project (Document 1, 2015).

"Yeah. Because as soon as they [government] changed, and Health Minister took over, we started trying to contact the government and get meetings - trying to push it through, and we were told well, we don't need another nurse practitioner pilot project, and we have no plan for funding right now...." [Participant 2; Health Care Professional]

"you always get the feeling there is background stuff at the government level. But I know I was just mystified, like what on earth is going on that this is closing?" (Participant 1, Professional Organization Leader)

Another participant talked about a lack of response to the letters and briefing notes that were provided to inform the sustainability. This participant believed that the clinic was just not a priority at the time, despite the public and professional outcry to save it. Thus, demonstrating a lack of accountability to the public.

"My simple understanding was that even though the minister was given several letters and briefing notes and/or heads up in high level meetings - which are just meetings between the executive teams and the ministry with the minster and her team- that it was not a priority to them" (Participant 18, Health Organization Leader)

Participants also talked about their inability to understand a process; and described this as a loss of trust in a process. Most participants (n=14) were unclear about who was funding the Sheep River NP Clinic, unclear about the length and type of funding for the clinic, who made the decision to close the clinic and what that decision-making process was. One participant clearly states that he/she was not sure how the clinic was funded.

"I don't know [how the Sheep River NP Clinic was funded]. I am pretty sure it was a government funded thing, but I don't know where from." (Participant 9, Patient)

"No [I don't know the terms of funding for the Sheep River NP Clinic], but the fact that they were in operation for a while and then there was threat of closure, and there was a Ministerial intervention, so they stayed open. That would tell me, they were revisiting, pretty frequently, the decision whether or not to keep this thing open." (Participant 1, Professional Organization Leader"

"I don't know for sure [who made the decision to close the clinic]. I assumed it was the PCN. I assumed it was the funders" (Participant 16, patient)

"No. I do not know who was involved with the closure of this clinic. Participant 17, Patient"

# Insufficient Consultation/Engagement (Organization, SIF)

Participants talked frequently about insufficient stakeholder consultation and engagement. Of particular participant concern was the lack of communication to inform stakeholders, patients and community members about the clinic closure. Participants had contradictory opinions or an overall lack of knowledge on who planned, was involved or participated in the decision to open, run and close the clinic. For example, many patient/community members participants didn't know who was involved with the implementation or closure of the clinic.

"I don't remember anything because I don't remember there being a problem. It was just all of a sudden they [Sheep River NP Clinic] were closing" (Participant 17; Patient)

"I don't [know why the decision was made to close the Sheep River NP clinic], but I remember it happening very abruptly. I remember one day it was fine, and the next day it appeared that boxes were being brought in to pack up things" (Participant 9; Patient)

Healthcare professionals and health organizational participants had contradictory understandings about who was involved in planning, why the decision to close the clinic was made and their responses varied. Some participants believed that senior level administrators within the main provincial health care organization were responsible, some believed that the CRPCN was responsible and others believed the Ministry of Health and their respective government representatives made the decision to close the clinic. "Yeah, I don't really know, [who or how the decision to close the clinic was made]" (Participant 6, Heath Organization Leader)

"Yeah, no it was as firm ongoing funding as anything we did in the PCN. It was never intended to be just temporary, because you wouldn't attach 1600 patients to a clinic and then just say, oh, we're shutting it down." (Participant 6, Health Organization Leader)

"I don't know if the PCN was involved in the decision to close the clinic. No, I don't know who was involved with the decision to close the clinic" (Participant 10, Professional Organization Leader)

"I actually am in the dark [as to why the decision was made to close the Sheep River NP Clinic closed]" (Participant 13, Health Care Professional)

# Funding Failure (Financial Factor, SIF)

Participants described several factors that contributed to the funding failure for the Sheep River NP. No documents clearly indicated a commitment to ongoing funding past the initial oneyear pilot funding period. Fifteen of the 18 participants knew that at the time of the clinic closure NPs in Alberta did not have a sustainability funding model.

"I mean what can be argued was that you need to have a sustainable funding model dedicated towards nurse practitioners. It can't flow through another group's hands first. It can't flow through PCN hands first or through physician's hands first because then you're asking another profession to develop the nurse practitioner role in primary care." (Participant 3; Health Care Professional)

"[we need a funding model where] ...nurse practitioners are funded provincially instead of by local PCNs" (Participant 17; Patient)

"Well, stronger government vision about how primary health care should be developed. A government who'd actually taken it seriously to develop a funding model for Nurse Practitioners" (Participant 1; Professional Organization Leader)

# The Right Model (Workforce, SIF)

Several participants provided supportive statements which positively reflect that the NP model was the right model for this patient population. Participants made statements about the quality, as well as the comprehensiveness of care that these NPs were providing to patients. An informal evaluation conducted by the CRPCN in 2012 confirmed the value of the services that the NP Sheep River clinic was providing. For example:

"There was also a benefit to them [PCN physicians] as the NPs provide coverage for lab work follow up when some of the physicians are away. All physicians who were interviewed expressed confidence in the level of skill demonstrated by the NPs and stated that they were confident that the NPs are practicing safely: they always ask for a second opinion in more complex situations, and know their limitations really well." (Document 19, 2012)

"I think those patients that were getting care at the clinic probably really missing their NPs. I know they got good care. They got quality care that they don't get from their physicians, and they have no other alternative." (Participant 13; Health Care Professional)

"because they [the NPs] were treated as full professionals running their Primary Care practice, and they did a great job to demonstrate that it was valuable" (Participant 6; Health Organization Leader)

"I think it [Sheep River NP Clinic Closure] forced those of us that were really attracted to a more holistic medical model [ and found that within the NP model]- we lost that when we lost the NPs". (Participant 16, Patient) In spite of the recognized value by patients and physicians, participants also talked about the continued marginalization and devaluing of the NP role in this case.

"The problem is that we're not equal players. When we're in a PCN we're generally employees. And so that gives somebody else all the power and employee have none.... you really have no voice as an NP." (Participant 3; Professional Organization Leader)

"I think NPs forget they are pioneers in a very NP resistant environment, where docs have way more power and are doing an invisible lobby and sometimes even visible." (Participant 18; Health Organization Leader)

Lastly, the idea that making change, taking risks, supporting success and utilizing careful planning can encourage an innovation is supported by participants and interview data. Participants described the Sheep River NP clinic as a novel or new idea introduced in Alberta and thought it was an innovative, cost effective and successful model of care. Participants also felt that this NP service delivery model met the needs of patient populations more effectively than existing models. Document analysis highlighted that the CRPCN evaluation "confirmed that the care provided in the community of Okotoks was of a high standard and well regarded by the community, at a comparable cost to other models" (Document 19, 2012).

"I thought it was a really effective model of care and actually fulfilled the government's directions around optimizing nurse practitioner workforce." (Participant 1, Professional Organization Leader)

"But the reality is that even under a PC government that tends to be physician friendly they took the risk to actually implement an alternate primary care provider to better meet the needs of the community. That to me was an innovation in itself. Then I think of those NPs as kind of pioneers in that particular community and region in the province - I think they established their service delivery model to really meet the needs of the population." (Participant 18, Health Organization Leader)

## Innovation Barriers (Innovation Supports, SIF)

Barriers to innovation can emerge from attitudes and perceptions of organizational leaders, stakeholders or individuals; other barriers can emerge from political forces and yet others can emerge from economic and financial triggers. Participants in their interviews described various barriers which include a lack of identified long term funding, the lack of patient and stakeholder consultation, the economic and political climate at the time the clinic was closing, frequent changeover of primary government leaders, the lack of health policy alignment to support the clinic and NP role, and the lack accountability on the part of organization and government leaders to address entrenched power and policy arrangements which impacted the sustainability of this clinic. See Table 4-10.

"For me, the Sheep River Clinic highlighted the systemic barriers to integrating nurse practitioners into the primary care system." (Participant 10, Patient)

"That (barriers to keeping the Sheep River NP Clinic open) is a multifactorial issue" (Participant 1, Health Care Professional)

## Profound Social Proof (Innovation Specific, SIF)

The overwhelming influence that the actions and attitudes of individuals can have on an innovation; the greater the number of people who support the innovation, the more the innovation is perceived to be a success. Participant interviews alluded to the Sheep River NP Clinic being an "unmitigated or absolute success" (Participant 1, Professional Organization Leader). The outcomes evaluation conducted by the CRPCN was extremely positive, there were numerous champions of the clinic that were vocal within the community, the Sheep River NPs collected meticulous supportive data, and there were supportive comments about the clinic in media sources by patients and community members (CRPCN, 2009; CRPCN, 2012; CRPCN, 2015).

"So there was this profound social proof right in front of me about the kind of trust that this clinic had built with their population and with their clientele. And they explained a bit about the history of the clinic to me and explained that they had initially been started as a bit of a desperation move because the PCN that had been established in Okotoks was short of physicians, and they had this growing population and no way to meet the primary health care needs in a reasonable way" (Participant 1, Professional Organization Leader)

"To be fair we started out as a pilot; a pilot that kept running for five years because we were effective" (Participant 4; Health Care Professional)

"I certainly did not see, hear or know of any evidence that the Sheep River NP clinic wasn't anything but an unmitigated success. Here was a model within a PCN that was doing brilliant work." (Participant 1; Professional Organization Leader

"I felt that if it [Sheep River NP Clinic] was successful it would continue. I remember when it opened, I asked that question and the assumption that I had was if they're busy enough or stay busy, they'll stay open. I probably wouldn't have gone there knowing that they could lose their funding. I felt very confident that I wouldn't go to a doctor and be told all of a sudden, the doctor's not there anymore. I mean I just, so obviously whatever I've found out at that time I assumed it was, if it's successful, meaning they have the patient base, then why would you close something like that." (Participant 15; Patient)

## **Newly Emerging Themes**

#### **Emotional** Aftermath

Stakeholder and patient emotions; the aftereffects and hidden consequences that are often not captured, or regarded as important to consider with sustainability failure. The participant interview data powerfully demonstrated the depth of emotion many patients and stakeholders felt when the Sheep River NP Clinic closed. Words such as 'devastated', 'a sense of loss', 'frustrated' and 'shocked' were used by participants to describe how they felt about the abrupt closure of this clinic.

"Devastated.... I felt like part of me was missing and I was left high and dry [when the Sheep River NP Clinic closed.]" (Participant 16, Patient)

"I just felt it was devastating for the community and for all the people involved, the nurse practitioners and their staff. I know the clinic, the doctors and the other receptionists or admin assistants that were working there were devastated as well. There was a great connection between them and a great degree of professionality and friendliness and working together as well too. And then, of course, all the patients. We were just sort of scattered" (Participant 9, Patient)

"losing the [Sheep River] NP clinic was equivalent to having two physicians move out of town" (Participant 11, Health Care Professional)

"it's just not the disruption from the patients that had [received that level of service from the NPs] ... Those patients were just sort of pushed aside, myself included. We were just sort of left" (Participant 9, Patient).

## Innovation Challengers

In addition to the champions of this innovation, there were innovation *challengers;* those individuals who were in opposition, acted against or were an obstacle to the Sheep River NP Clinic sustainability. Innovation challengers included professional groups and organizations and/or individuals who used their title, position and power to exert a degree of influence on the Sheep River Clinic process, or initiated measures that restricted or protected themselves against competition from NPs based on the fear of losing power, status and/or income.

"I think NPs forgot they are pioneers in a very NP resistant environment, where docs have way more power and are doing an invisible lobby and sometimes even visible" (Participant 18, Health Organization Leader)

"[the closure of the Sheep River NP Clinic] was one big perfect storm - a maelstrom of organizational/political and professional power/lobby and corruption" (Participant 18, Health Organization Leader)

" I think [given] the physician lobby to invest in physician roles and PCN,

the new government was more focused on relationship management with physicians, hence they funded physician services [not NP services] "(Participant 13, Health Care Professional)

# **Technicolor Data**

Colorful, meaningful and meticulously collected evidence, in multiple forms illustrated the success of the Sheep River NP clinic and should have secured the clinic's sustainability. Participants spoke about the evidence that could have been used to validate the Sheep River NP clinic effectiveness and secured its sustainability. Evidence discovered in the documentation and confirmed in the interview's points to the Sheep River NP Clinic being an "unmitigated success". The outcomes evaluation conducted by the CRPCN was extremely positive (Document 19, 2012), the Sheep River NPs collected meticulous supportive data, and there were supportive comments about the clinic in media sources by patients and community members (Document 1, 2015; Document 3, 2015)

"I certainly did not see, hear or know of any evidence that the Sheep River NP clinic wasn't anything but an unmitigated success." (Participant 1, Professional Organization Leader)

"Here was a model within a PCN that was doing brilliant work." (Participant 3, Professional Organization Leader)

"...so, it was not all opaque the difference that they'd [Sheep River NPs] made in their community. They had data, right there in technicolor, you could see the difference they made in their community" (Participant 1, Professional Organization Leader)

Document analysis revealed the Sheep River NP Clinic had a positive evaluation in 2012, the NPs collected their own supportive outcomes data and the clinic had strong community/patient support (Document 19,2012).

In spite of this technicolor data, the Sheep River NP clinic closed its doors on January 15<sup>th</sup>, 2016.

#### Discussion

The findings (ten themes) from this case study align with the previously published Sustainability of Innovation and Policy Triangle Frameworks. However, three unique themes emerged from the data analysis for this case. These themes add additional depth to descriptions of sustainability and highlight the challenges and complexity of sustaining innovations in a dynamic health care environment. The results of this case study are grouped into four key discussion points: 1) despite evidence, "technicolor data", that points to the Sheep River NP clinic being effective and "an unmitigated success", it was a pilot project that was not sustained, 2) there was a lack of evidence-informed policy to support the decision making in regards to sustainability of this clinic. There was also a lack of accountability, transparency, consultation and communication on behalf of funders and governments regarding the decision making processes involved with the clinic's closure, 3) marginalization and devaluing of the NP role continues to be problematic in primary care settings in Alberta and this influences a sustainable funding solution for NPs, and 4) a major finding in this case study is three new emerging themes which further inform sustainability of innovations in PHC and add to our understanding of the complexity of the sustainability process. These themes together with the SIF and PTF captured the breadth of factors influencing sustainability of the PHC innovation in this case.

First and foremost, this case study analysis revealed that the reasons for closure of the Sheep River NP clinic were multifactorial in nature and combined with the 'perfect storm' of circumstances, the clinic ultimately closed. Canada is well known as the "Land of Pilot Projects". The country perpetually introduces and implements projects that are narrow in scale, prove that they work, and then hope that if successful, this will be enough to convince governments, funders and policy makers to do things differently.<sup>51</sup> The Sheep River NP Clinic was no different. The clinic was created as a one-year pilot project with no documented commitment that future sustainability or funding of the clinic was considered. A comprehensive evaluation of the clinic and other data sources known to decision makers provided evidence of the clinic's success. There is evidence that the CRPCN was hopeful the provincial government would create a sustainable funding solution for NPs, but when this did not happen the clinic ultimately closed.<sup>51</sup>

Second, the decision-making process of closing the Sheep River NP clinic lacked accountability, transparency and clarity. No documentation or information provided an accurate

picture of why the clinic closed; nor was there a clearly stated rationale for clinic closing. Important stakeholders, patients, community members and employees of the clinic were not consulted about the closures, nor did these stakeholders receive timely communication about closures decisions.<sup>23</sup> However, it is evident that the decision to close the clinic was made by the CRPCN due to a lack of jurisdictional support from the health care funder. In absence of formal funding policy, the CRPCN was able to make whatever decision they wanted, regardless of stakeholder consultation and technicolor data. Health service organizations, such as Primary Care Networks (PCNs) need to engage in transparent decision-making processes that use evidence to inform choices that support patient centered approaches to care.<sup>22</sup> Third, participants talked about the continued marginalization and devaluing of the NP role in Alberta. There is continued confusion about the NP role due to a lack of understanding of the role and uncertainty of how the NP can contribute to primary care. The issue is not unique to Alberta and continued limited stakeholder understanding of the NP role is a huge barrier to optimization and a sustainable funding solution for NPs across the country.<sup>22</sup>

Finally, three new themes (Emotional Aftermath, Technicolor Data, Challengers) emerged in the results of this case study. These themes capture additional components of the sustainability process in PHC which was not reflected in the SIF or PTF. These themes offer novel insights into the complexity of sustaining innovations such as the NP role in PHC and these findings raise important questions about the political, social and emotional nature of the sustainability process. These new themes emphasize the multi-facetedness and evolving nature of the sustainability of PHC innovations process. These themes further demonstrate that sustainability of innovations in PHC is composed of a variety of interrelated elements, which should be simultaneously assessed and managed to strengthen the ability of the health care system to promote health and wellbeing.

## Implications

Sustainability is crucial to health system reform, however the literature in this area of study is sparse. This study was informed by two conceptual models, the SIF<sup>11</sup> which has not been empirically tested and the PTF<sup>12</sup> which to this point has been used extensively in health policy research but only addresses the decision making and policy process of sustaining a PHC innovation. This study confirms that the five factors described in the SIF and the PTF are

applicable to sustainability assessment. However, three unique themes emerged from the data in this study to complement sustainability assessment.

## Limitations

One of the main limitations of case study research is the generalizability of results to other contexts. This case study occurred in Alberta, Canada and due to variable policy and NP initiatives across Canada it is difficult to generalize findings to all Canadian jurisdictions. However, this case produced rich and detailed data about the phenomenon of interest: sustainability failure. The second limitation is that this was a case study using qualitative analysis and no economic data was collected or analyzed. The CRPCN's funding formula utilizes rostering per patient and is not funded beyond this small per patient top up. As such, there is no explicit funding for NPs in the current Alberta PCN funding formula and the Sheep River NP clinic likely operated at a loss. The third limitation involves the potential for researcher bias, the primary investigator was a NP, as were two of her committee members. However, this limitation was mitigated through the involvement of a PhD committee member from the discipline of Public Health with expertise in policy analysis. The fourth limitation involves the length of time between when the Sheep River NP clinic closed and when interviews and data collection occurred. This could lead to recall bias as the participants may unintentionally forget certain important details of their involvement in the case.

### **Future Research**

Key areas for future research include: a) advancing substantive research on sustainability and sustainability failures of the NP role and other PHC innovations, b) advancing methods for sustainability research in PHC, and (c) uilizing mixed methods case studies and applying health policy and economic analysis to further understand the role of NPs in primary care. It is critical there is improved focus on understanding, measuring, and communicating the value of sustainability for the multiple stakeholders involved in the sustainability process.

### Summary

Canada's health system continues to experience significant PHC reform delays. Findings demonstrate that the closure of the Sheep River NP clinic was multifactorial, and combined with a 'perfect storm' of events, sustainability failed. Sustainability requires continuing effort through time, rather than representing a final state to be achieved. Our study offers new insights into the process of sustainability of organizational change, and elucidates the complement of strategies needed to make bottom-up change last in challenging contexts replete with competing priorities.

- Xia, L (2016). Is There A Sustainability Imperative for The Healthcare Industry? *Huffington Post.* <u>https://www.huffingtonpost.com/lisa-xia/is-there-a-</u> <u>sustainability b 10775666.html</u>
- Hutchison B, Levesque JF, Strumpf E, Coyle N. Primary health care in Canada: systems in motion. *Milbank Q*. 2011;89(2):256-88. doi: <u>10.1111/j.1468-0009.2011.00628.x</u>
- 3. Stonebridge C, Brichta J, Dinh T. A Road Map to Health System Sustainability: CASHC Compendium Report, 2011–16. The Conference Board of Canada - Economic forecasts, public policy, and organizational performance. <u>https://www.conferenceboard.ca/e-library/abstract.aspx?</u>

<u>did=8816&AspxAutoDetectCookieSupport=1</u>. Published May 15, 2017. Accessed March 20, 2020.

- Martin GP, Weaver S, Currie G, Finn R, Mcdonald R. Innovation sustainability in challenging health-care contexts: embedding clinically led change in routine practice. *Health Serv Manage Res.* 2012;25(4):190-9. doi: 10.1177/0951484812474246
- Naylor MD, Kurtzman ET. The role of nurse practitioners in reinventing primary care. *Health Aff* (Millwood). 2010;29(5):893-9. doi: 10.1377/hlthaff.2010.0440.
- Sangster-Gormley E, Martin-Misener R, Burge F. A case study of nurse practitioner role implementation in primary care: what happens when new roles are introduced? *BMC Nursing*. 2013;12(1). doi:10.1186/1472-6955-12-1Mattison, C. A., & Wilson, M. G. (2018). Rapid Synthesis: Enhancing Health System Integration of Nurse Practitioners in Ontario.

- Martínez-González NA, Djalali S, Tandjung R, et al. Substitution of physicians by nurses in primary care: a systematic review and meta-analysis. *BMC Health Services Research*. 2014;14(1). doi:10.1186/1472-6963-14-214.
- Martin-Misener R, Harbman P, Donald F, et al. Cost-effectiveness of nurse practitioners in primary and specialised ambulatory care: systematic review. *BMJ Open*. 2015;5(6). doi:10.1136/bmjopen-2014-007167.
- Swan M, Ferguson S, Chang A, Larson E, Smaldone A. Quality of primary care by advanced practice nurses: a systematic review. *International Journal for Quality in Health Care*. 2015;27(5):396-404. doi:10.1093/intqhc/mzv054.
- Registered Nurses Association of Ontario. Fully Utilizing Nurse Practitioners (NP): Prescribing Controlled Substances and Other Expanded Au.; 2017. https://rnao.ca/sites/rnao-ca/files/QP\_Day\_2017-\_Fully\_Utilizing\_NPs\_final\_public.pdf. Accessed May 5, 2020.
- 11. Fox A, Gardner G, Osborne S. A theoretical framework to support research of health service innovation. *Aust Health Rev.* 2015;39(1):70-75. doi:10.1071/ah14031
- Mattison CA, Wilson MG. Rapid Synthesis: Enhancing Health System Integration of Nurse Practitioners in Ontario
- 13. Walt G, Gilson L. Reforming the health sector in developing countries: the central role of policy analysis. Health Policy Plan. 1994;9(4):353-70. doi:10.1093/heapol/9.4.353
- Delvin ME, Braithwaite S, Plazas PC. Canadian nurse practitioner's quest for identity: A philosophical perspective. *International journal of nursing sciences*. 2018 Apr 10;5(2):110-4.

15. Canadian Nurses Association (2016, August). The Canadian Nurse Practitioner Initiative: A Ten Year Retrospective. <u>https://cna-aiic.ca/-/media/cna/page-content/pdf-en/canadian-nurse-practitioner-initiative-a-10-year-</u>

retrospective.pdf?la=en&hash=406D8F34F947CFD94DA04AE0CF56A739514D01B7 OR

Canadian Nurses Association (2016). Position Statement Nurse Practitioners. https://www.cna-aiic.ca/-/media/cna/page-content/pdf-en/the-nurse-practitioner-positionstatement\_2016.pdf?la=en&hash=B13B5142C8D02990439EF06736EA284126779BCC

- Dicenso A, Auffrey L, Bryant-Lukosius D, et al. Primary health care nurse practitioners in Canada. *Contemporary Nurse*. 2007;26(1):104-115. doi:10.5172/conu.2007.26.1.104
- Dicenso A, Bourgeault I, Abelson J, et al. Utilization of Nurse Practitioners to Increase Patient Access to Primary Healthcare in Canada – Thinking Outside the Box. *Nursing Leadership.* 2010;23(sp):239-259. doi:10.12927/cjnl.2010.22281
- Donald F, Martin-Misener R, Bryant-Lukosius D, et al. The Primary Healthcare Nurse Practitioner Role in Canada. *Nursing Leadership*. 2010;23(sp):88-113. doi:10.12927/cjnl.2013.22271
- Edwards N, Rowan M, Marck P, Grinspun D. Understanding Whole Systems Change in Health Care: The Case of Nurse Practitioners in Canada. *Policy, Politics, & Nursing Practice.* 2011;12(1):4-17. doi:10.1177/1527154411403816
- 20. Edwards N, Rowan M, Marck P, Grinspun D. Understanding Whole Systems Change in Health Care: The Case of Nurse Practitioners in Canada. *Policy, Politics, & Nursing Practice*. 2011;12(1):4-17. doi:10.1177/1527154411403816
- 21. Nurse Practitioner Association of Canada. (2019, May 24 email). President's Newsletter.

- 22. Wong ST, Farrally V. The utilization of nurse practitioners and physician assistants: A research synthesis. *Michael Smith Foundation for Health Research*; 2014.
- 23. Canadian Federation of Nurses Unions. (2018). Media Release.
  https://nursesunions.ca/wp-content/uploads/2017/11/2017-1113\_NursePractitionerWeek\_EN-1.pdf
- 24. El-Jardali, F., & Lavis, J. N. Issue Brief: Addressing the Integration of Nurse Practitioners in Primary Healthcare Settings in Canada. *McMaster Health Forum*. https://www.mcmasterforum.org/docs/default-source/product-documents/issuebriefs/nurse-practitioners-in-primary-healthcare-in-canada-ib.pdf?sfvrsn=2
- 25. Ergen E. Health Policy and Economics: The adoption of generic drugs in Greek healthcare system. DOI: <u>10.13140/RG.2.1.2261.6163</u>
- 26. Canadian Nurses Association. Primary Health Care The Time has Come. 2003: https://www.cna-aiic.ca/~/media/cna/page-content/pdffr/primaryhealthcare\_sept\_2003\_e.pdf?la=en
- 27. Canadian Nurses Association. Revised Position Statement Primary Health Care. 2015: <u>https://www.cna-aiic.ca/-/media/cna/page-content/pdf-en/primary-health-care-position-</u> <u>statement.pdf?la=en&hash=611712E3AAC033BF0C4E94FB3F07F30866F6B692</u>
- Scheirer MA, Dearing JW. An Agenda for Research on the Sustainability of Public Health Programs. *American Journal of Public Health*. 2011;101(11):2059-2067. doi:10.2105/ajph.2011.300193
- 29. Stirman SW, Kimberly J, Cook N, Calloway A, Castro F, Charns M. The sustainability of new programs and innovations: a review of the empirical literature and recommendations for future research. *Implementation Science*. 2012;7(1). doi:10.1186/1748-5908-7-17

- 30. Lennox L, Maher L, Reed J. Navigating the sustainability landscape: a systematic review of sustainability approaches in healthcare. *Implementation Science*. 2018;13(1). doi:10.1186/s13012-017-0707-4
- 31. Sibthorpe BM, Glasgow NJ, Wells RW. Emergent themes in the sustainability of primary health care innovation. *Medical Journal of Australia*. 2005;183(S10). doi:10.5694/j.1326-5377.2005.tb07185.x
- 32. Wakerman, J., Humphreys, J. S., Wells, R., Kuipers, P., Jones, J., Entwistle, P., & Kinsman, L. (2009). Features of effective primary health care models in rural and remote Australia: a case-study analysis. *Medical Journal of Australia, 191*(2), 88-91.
- 33. Wakerman J, Humphreys JS. Sustainable primary health care services in rural and remote areas: Innovation and evidence. *Australian Journal of Rural Health*. 2011;19(3):118-124. doi:10.1111/j.1440-1584.2010.01180.x
- 34. Patterson D. 2016 census shows Okotoks' once high growth rate now slowing. http://www.westernwheel.com/article/2016-census-shows-Okotoks-once-high-growthrate-now-slowing-20170208
- 35. Conrad K. Lack of Funding Closes Okotoks Clinic. <u>https://www.okotokstoday.ca/local-news/lack-of-funding-closes-okotoks-clinic-1531455</u>
- 36. Spencely S. A New Vision for RN Practice. College and Association of Registered Nurses of Alberta, 71(2), 3. <u>http://www.nurses.ab.ca/content/dam/carna</u> /pdfs/AB%20RN/2015/ABRN\_Summer15.pdf
- Calgary Rural Primary Care Network. 2009: Primary Care Service Project Charter.
   Calgary, AB: Canada. Calgary Rural Primary Care Network.

- Calgary Rural Primary Care Network. 2015:. Dear Minister Mandel. Calgary, AB: Canada. Calgary Rural Primary Care Network.
- 39. Yin RK. Case study research: Design and methods (5th ed.). Thousand Oaks, CA: Sage.
- 40. Ma F, Lv F, Xu P, et al. Task shifting of HIV/AIDS case management to Community Health Service Centers in urban China: a qualitative policy analysis. *BMC Health Serv* Res. 2015;15:253. doi: 10.1186/s12913-015-0924-y
- Ergen E. Health Policy and Economics: The adoption of generic drugs in Greek healthcare system. DOI: <u>10.13140/RG.2.1.2261.6163</u>
- 42. Greenhalgh T, Robert G, Bate P, Kyriakidou O, Macfarlane F, Peacock R. How to spread good ideas - A systematic review of the literature on diffusion, dissemination and sustainability of innovations in health service delivery and organisation. *Report for the National Co-ordinating centre for NHS Service Delivery and Organisation R & D (NCCSDO)*, 1-426
- 43. Chambers DA, Glasgow RE, Stange KC. The dynamic sustainability framework: addressing the paradox of sustainment amid ongoing change. *Implementation Science*. 2013;8(1). doi:10.1186/1748-5908-8-117
- 44. Creswell JW. *Research design: Qualitative, quantitative, and mixed methods approaches* (4th ed.). Los Angeles, CA: Sage
- Tuckett AG. Part II. rigour in qualitative research: complexities and solutions. Nurse Res. 2005;13(1):29-42. DOI:10.7748/nr2005.07.13.1.29.c5998.
- 46. Morse J. Qualitative Methods: The State of the Art. *Qual Health Res.* 1999;9(3):393-406. doi:10.1177/104973299129121938

- 47. Baker SE, Edwards R, Doidge M. How many qualitative interviews is enough?: expert voices and early career reflections on sampling and cases in qualitative research. *National Centre for Research Methods,* Southampton, UK: National Centre for Research Methods, Southampton. 2012 Mar 26.
- 48. Braun, V, Clarke V. Using thematic analysis in psychology. *Qual Research in Psych*.
  2006: 3(2), 77-101. DOI: 10.1191/1478088706qp063oa
- 49. Calgary Rural Primary Care Network. 2012. Evaluation: Family Practice Nurse Practitioner Clinic in Okotoks. Calgary, AB, Canada. Calgary Rural Primary Care Network.
- 50. Woltering L. 'Pilots Never Fail, Pilots Never Scale': Why the Global Development Community Needs a More Realistic Approach to Reaching Billions. <u>https://nextbillion.net/global-development-needs-realistic-approach-to-scale/</u>
- 51. Larrabee JH. Nurse to nurse evidence-based practice. McGraw Hill Professional; 2008

Figure 4-1: Policy Triangle Framework by Walt & Gilson<sup>35</sup>



Table 4-1: Policy Triangle Element<sup>35</sup>

Policy Triangle Framework Components	
Context	<ul> <li>The context includes the 'why' of the policy process and considers how political history, economic ideologies, characteristics of government systems and cultural and environmental factors influence the policy process.</li> <li>Asks why are certain policy decisions made and why do policies and agendas change (Walt &amp; Gilson, 1994; Ergen, 2012).</li> </ul>
Content	<ul> <li>The content is the 'what' of the policy process, and considers the conditions under which the policy process occurred; the research, evidence, and the rules, regulations and legislation that influenced the policy process.</li> <li>It considers the objectives of the policy, the evidence and the legislation supporting a policy decision (Walt &amp; Gilson, 1994; Ergen, 2012).</li> </ul>
Process	<ul> <li>Process is the 'how' and looks specifically at policy development and the process of policy-making.</li> </ul>

<ul> <li>Assists in understanding how public policy is made and implemented, whom it may influence, and how it is evaluated (Walt &amp; Gilson, 1994; Ergen, 2012).</li> </ul>

Figure 4-2: Health Service Innovation Framework by Fox, Gardner & Osborne<sup>34</sup>


Table 4-2: Sustainability of Innovation Framework<sup>34</sup>

	Sustainability of Innovation Framework
Political Factors	<ul> <li>This factor focuses on appropriate government alignment, linking with regional health plans, involvement of local and national champions, as well as staff involvement.</li> <li>Evidence indicates that a political focus on one particular policy can have a strong influence on the sustainability of an innovation.</li> <li>When assessing factors that influence health services innovation sustainability, questions to consider include: Was the policy aligned with government health planning? Is there a pending change in government that could impact sustainability? Who are the political 'champions' that can support the innovation? and What are the funding provisions surrounding the policy?</li> </ul>
Organizational	<ul> <li>Having flexibility and the ability to adapt an innovation to support the organizational environment is shown to encourage sustainability.</li> <li>When assessing factors that influence sustainability of a health services innovation, questions to consider include: What are the established communication pathways in the organization? What networking and communication needs to occur to include all elements of the organization? What levels of government and other organizational stakeholders should be involved? and How can an organization adapt the innovation to fit the local environment</li> </ul>
Workforce	<ul> <li>Research indicates that innovations aligned with the value and needs of employees are more readily adopted and sustained.</li> <li>When assessing factors that influence sustainability of a health services innovation, questions to consider include: Are the values and needs of employees examined in the planning and</li> </ul>

Innovation	<ul> <li>implementation of the innovation? Was sufficient recruitment, workforce and succession planning assessed? Are minimal changes to roles, policies and procedures expected? and Is regular communication with employees throughout the process anticipated</li> <li>Innovations that are flexible and adaptable, have the ability to respond to a dynamic funding and policy environment, and continue to evolve.</li> <li>These characteristics are an important feature of sustainability. When assessing factors that influence sustainability of health services innovations, questions to consider include: How are stakeholders involved in the innovation? Will they be consulted on the acceptability, quality and safety of the innovation? What barriers and supports to the innovation exist? and Will an employee the initiated and if a hearder (Terr et al. 2015)?</li> </ul>
Financial	<ul> <li>evaluation process be initiated, and if so by who (Fox et al., 2015)?</li> <li>One of the most important factors influencing sustainability of a health services innovation are the financial factors.</li> <li>These factors include the service delivery model and source and length of funding.</li> <li>When assessing factors that influence financial sustainability of a health services innovation, questions to consider include: What funding sources have been identified and secured? Is the budget adequate to meet the needs of an innovation running long term? What will happen when the funding ceases? and Is there any consideration of program sustainability in program planning</li> </ul>

Figure 4-3: Patient and Stakeholder Research Information Letters and Consent Forms



STAKEHOLDER/PROFESSIONAL VERSION

## Study Title: Sheep River Nurse Practitioner Clinic: A Case Study Supervisor/Principal Investigator: Dr. Tammy O'Rourke Assistant Professor University of Alberta, Faculty of Nursing EMAIL: torourke@ualberta.ca PHONE: 780-492-2699 Student Research Investigator:

Raelene Marceau MN. NP

. . . . .

PhD Candidate

Faculty of Nursing

University of Alberta

EMAIL: <a href="mailto:rroot@ualberta.ca">rroot@ualberta.ca</a>

**PHONE**: 780-618-5396

#### **Background**

My name is Raelene Marceau and I am a PhD candidate through the Faculty of Nursing at the University of Alberta. I would like to invite you to participate in my thesis research study examining the closure of the Sheep River Nurse Practitioner Clinic in Okotoks, AB. You have been identified through a key informant or via newspaper, media or public government documents, as an individual who had involvement or accessed this clinic.

#### <u>Purpose</u>

Sustainability often comes up in discussions and debates concerning health system transformation and primary health care reform. Numerous efforts aimed at re-organizing our health system have led to the introduction of new programs and services; however, few of these have been maintained over the long-term. This creates wasted resources, frustration and increases resistance to later initiatives to improve health care. To date, there has been a limited focus on how to achieve sustainability of health service innovations, which focus on changing the health system such as the NP role. This issue is poorly understood and there is little available literature on the sustainability of the NP role or how to sustain other programs or services in primary health care.

Nurse Practitioners (NPs) are healthcare professionals with additional education, either a Masters or PhD degree. NPs are legally able to assess and treat chronic and acute illnesses, prescribe medications, order diagnostic tests and comprehensively coordinate care to a broad population of patients in a wide variety of settings. In Canada, several issues prevent NPs from working in their professional roles. The most significant current issue is the lack of a sustainable funding to support this role which has resulted in a lack of utilization of NPs across the country. It is crucial we understand the limitations of how NPs are funded and through research, make recommendations towards creating sustainability of the NP role.

The purpose of this research is to closely examine the closure of the Sheep River Nurse Practitioner Clinic to better understand the following:

- 1. Why the Sheep River Nurse Practitioner Clinic closed and what were the factors that influenced clinic closure?
- 2. How was the decision to close the Sheep River Nurse Practitioner Clinic made and who was involved in the decision to close the clinic?
- 3. Was the Sheep River Nurse Practitioner clinic sustainable?

#### **Participation**

You have been identified as a potential participant in this study due to your involvement in the Sheep River Clinic or your professional position at the time of the Sheep River Clinic closure. If you volunteer to take part in this study, you will be asked to do the following:

- 1. Provide information about yourself, such as your age, occupation and where you live.
- 2. Complete an interview. Interviews will be held at a predetermined location in Okotoks, Alberta or at a location more suitable to your work/life needs. Alternatively, a telephone interview can be scheduled. The interview will take about 60-90 minutes, will be conducted by the student researcher, audio-taped and later transcribed for the purpose of data analysis

- 3. Respond to questions regarding the Sheep River Nurse Practitioner Clinic, including how you were involved and what your knowledge was of the clinic and its subsequent closure.
- 4. The interviews will be conducted in February, March and April of 2019.

#### <u>Benefits</u>

- The anticipated benefit of participating in this research is the opportunity to discuss your knowledge, insights and concerns related to the Sheep River Nurse Practitioner Clinic. An additional benefit is the opportunity to contribute to our improved understanding of how to sustain NP roles and other primary health care programs and services in our health system. However, it should be noted there might be no benefit to participating in this research.
- There are no costs to the participant for taking part in this research.
- There is no compensation or reimbursement for participating in this research.\

#### <u>Risks</u>

• Potential risks or discomforts include possible uncomfortable emotional feelings when asked questions during the interview. While this project may involve some professional and emotional risks, the utmost care will be taken to anonymize your information and protect your identity. If you do feel any emotional or psychological discomfort from being involved in the study, or for any other reason, you are free to limit your participation and/or withdraw from the study at any time.

#### Voluntary Participation

- Your participation in this study is voluntary and you are under no obligation to participate.
- You may decline to answer any question(s) you prefer not to answer by requesting to skip the question.

#### **Confidentiality & Anonymity**

• Every effort will be made to ensure confidentiality of information about you that is obtained during this research study. What you say during the interview will be audio taped and then transcribed, but names and any identifying information will be removed from the data. Data (audio recordings, personal information, interview notes and anonymized transcripts) will be kept by the research team for a period of 5 years

following the end of the study. The data will be kept in a safe place on a secure, encrypted computer drive that only the research team has access to.

• The results from this study will be presented in academic/professional journals read by government, policy and health care professionals to help them better understand the research topic. The results may also be presented in person to groups of government, policy and/or health care professionals. At no time, however, will your name be used or any identifying information revealed. If you wish to receive a copy of the results from this study, you may contact one of the researchers at the telephone number given below.

#### Rights of Research Participants:

- You may withdraw from this study until April 30th, 2019, after which your anonymized data will be included in the final research report. To withdraw from the study, participants should contact the researcher by phone or email. After you leave the study, we will not collect any more information about you for the study.
- The plan for this study has been reviewed by a Research Ethics Board at the University of Alberta. If you have questions about your rights or how research should be conducted, you can call (780) 492-2615. This office is independent of the researchers

#### *How to participate in this study:*

- 1. Review this letter of information.
- 2. Provide consent if you want to participate.
- 3. Wait for the researcher to contact you to schedule an interview

#### **Further Information**

- If you require any information about this study, or would like to speak to one of the researchers, please call Raelene Marceau (Student Investigator) or Dr. Tammy O'Rourke (PhD Supervisor/Co-investigator) at 780-492-2699.
- The plan for this study has been reviewed by a Research Ethics Board at the University of Alberta. If you have questions about your rights or how research should be conducted, you can call (780) 492-2615. This office is independent of the researchers.

#### **Conclusion**

We thank you in advance for thinking about being part of this study. Please do not hesitate to contact me with any questions relating to the study. *Sincerely,* Raelene Marceau MN, NP, PhD(c)

#### **CONSENT FORM**

Study Title: Sheep River Nurse Practitioner Clinic: A Case Study

#### **Supervisor /Principal Investigator:**

Dr. Tammy O'Rourke Assistant Professor University of Alberta, Faculty of Nursing EMAIL: <u>torourke@ualberta.ca</u> PHONE: 780-492-2699

#### **Student Research Investigator:**

Raelene Marceau MN, NP PhD Candidate Faculty of Nursing University of Alberta EMAIL: <u>rroot@ualberta.ca</u> PHONE: 780-618-5396

	Ye No
	S
Do you understand that you have been asked to be in a research study?	
Have you read and received a copy of the attached Information Sheet?	
Was this consent read aloud in its entirety?	
Do you understand the benefits and risks involved in taking part in this research study?	
Have you had an opportunity to ask questions and discuss this study?	
Do you understand that you are free to leave the study at any time, without	
having to give a reason and without affecting your health care and or Sage	
services?	

Has the issue of confidentiality been explained to you?	
Do you understand who will have access to your data, including personally identifiable information?	
Who explained this study to you?	
I agree to take part in this study.	
Name of Participant (please print) Signature of Participant	Date
Signature of witness: I believe that the person signing this form understands	s what is involved in

the study and voluntarily agrees to participate.

	-			
Name of Witness (please print)		Signature of Witness	]	Date

1. What can you recall about the Sheep River NP Clinic?
2. What was your involvement with the Sheep River NP Clinic?
3. Probe: What was your role? Stakeholder? Employee etc
4. Can you tell me why this clinic opened and what about this clinic was innovative?
5. Who was involved in the planning and opening of the clinic?
6. Who were the staff at the Sheep River NP Clinic?
7. What do you know about how the Sheep River NP Clinic was funded?
8. Probes: how was the clinic funded?
9. Do you know if there were formal policies in place to support the development of the Sheep River Clinic?
10. Probe: Say for example from Alberta Health and Wellness, Alberta Health or the Primary Care Network?
11. If there were a policies, were these linked or aligned to Alberta Health goals and vision at the time?
12. Probe: Do you know if this policy changed? If so, why?
13. Were NPs consulted on any decisions regarding health policy relevant to the Sheep River Clinic?
14. What do you understand about NPs and their role in PC?
15. What do you know about the PCN and how it was structured?
16. Probe: (patient) What kinds of 'things' did the NP do for you?
17. Who were the champions of the Sheep River Clinic?
18. Probes: Were there municipal, provincial government, PCN, AH directors as champions?
19. Were there other things happening in health care from a political perspective at the time of the closure of the
clinic?

20. There was an evaluation of the clinic conducted by the PCN in March 2012, do you know if the evaluation
influenced the decision to close the clinic?
21. Probe: Do you know if there were any other evaluations of the clinic during the time the clinic was operational or
prior to the closure of the clinic?
22. What is your understanding of why the Sheep River Clinic was closed?
23. Where and when was the decision to close the Sheep River NP clinic made?
24. Were you involved in the decision to close the clinic?
25. Who made the decision to close the Sheep River NP Clinic?
26. How did individuals communicate about the clinic closure within the PCN?
27. Probes: How was information shared about the closure of the clinic? Within the clinic, within the PCN and
Alberta Health?
28. Was the Sheep River NP clinic intended as a pilot project?
29. Do you know if the budget was adequate enough to meet the needs of running the clinic?
30. How were the decisions to staff the clinic made? Who made those decisions?
31. What kind of communication and updates did the PCN provide to the clinic staff on clinic outcomes and quality
of care?
32. How were the NPs recruited to the clinic and were there any staffing changes during the time the clinic was in
operation?
33. What might have prevented the closure of the clinic?
34. What were some barriers to keeping the clinic open and what efforts, if any, were made to avoid closure?
35. Probes: Staff turnover? Lack of key support? Lack of technical assistance? Financial?
36. What else could have been done to keep the clinic open?
37. What was the effect of closing the NP clinic on the community? Surrounding communities?
38. Do you know anyone else was involved in the planning, implementation or closure process

	<ul><li>39. How did you feel when the clinic closed? (Patient)</li><li>40. What was the impact on your community, on you, how long did it take you to find another family physician.</li><li>41. Did you have contact from any organization to obtain your feedback on whether this clinic should close? (Patient)</li></ul>
Closing Key Components	<ul><li>42. Is there anything more you would like to add? Is there anything you would like NOT included?</li><li>43. Thank you for your time.</li></ul>

### Table 4-4: A-Priori Preliminary Codes

Sustainability Innovation Framework Factors	RESEARCH QUESTIONS
Organization/Stakeholder	1. Why did the Sheep River NP Clinic close and what were the contextual
Systems	factors influencing clinic closure?
Political	2. What was the decision-making process encompassing clinic closure and
Workforce/Professional	what did this process entail?
Innovation Barriers	3. Was the Sheep River NP clinic sustainable and what lessons can be
Innovation Supports	garnered from clinic closure to inform organizations, funders, policy
Financial Funding	makers and governments on sustaining the NP role and other PHC
Context	health service innovations?
Why Decision Made	
Content	
What was the decision-making process	
Legislation/Evidence/Policy Supporting Decision	
Content What were the Political Conditions	
Process How Involvement in Decision	
Actors WhoParticipated in the Decision	

# Table 4-5: How Themes Support the Concepts in the Policy Triangle and Sustainability of Innovation Frameworks

Policy Triangle	Key Theme	Category	Informe	Key Theme	Quote	Particip	Docum
Framework			d by	Definition		ant #	ent #
Context - context	Motivation for Clinic	Motivation for	Document	The 'why' or	"Initially opened	# 2	
element includes the	<b>Opening and Closing</b>	Clinic Opening	s and	reasons and	as a pilot project		
'why' of the policy			Interview	rationale for	to fulfill a need:		
process and considers				clinic	20% of people in		
how political history,				opening.	Okotoks without		
economic ideologies,				These	a family		
characteristics of				include a	physician"		
government systems				shortage of			
and cultural and				family	"The pilot was		
environmental factors				physicians,	initiated in a		#22
influence the policy				lack of	response to a		
process. This element				access to	chronic lack of		
asks why are certain				primary care	access to primary		
policy decisions				and huge	care physicians		
made and why do				population	in the town of		
policies and agendas				boom in	Okotoks. As		
change				Okotoks	discussed above,		
				putting	all current		

			pressure on	physicians were		
			existing	at capacity and		
			primary care	there were		
			services.	limited		
				opportunities to		
				recruit more"		
Motivation for Clinic	Motivation for	Document	The 'why' or	"somebody	#17	
<b>Opening and Closing</b>	Clinic Closing	s and	reasons and	pulled funding"		
		Interview	rationale for			
		s	clinic	"Loss of	#17	
			closure.	AH/AHS		
			These	executives who		
			include the	were nurses and		
			Calgary	female"		
			Rural			
			Primary Care	"There is no		
			Network	money we would		
			(CRPCN)	want to spend to	#4	
			becoming	continue to fund		
			more fiscally	this"		
			accountable			
			and moving	"Currently, the		
			from a local	clinic operates at		#22
			model of	a cost of		
			primary care	\$297,557.48 per		

		managed by	annum, caring	
		each	for a panel of	
		community	approximately	
		in the	1600 patients.	
		CRPCN, to a	Unfortunately,	
		more	funding the	
		centralized	program at this	
		and less	amount prevents	
		individual	the local	
		focused	community from	
		approach.	funding other	
		The fact that	programs and	
		there was	services that also	
		Provincial	benefit patients	
		Government	of PCN member	
		election and	physicians	
		shift from	practicing in the	
		one political	area"	
		party		
		philosophy to		
		another, the		
		fact that there		
		were three		
		different		
		Ministers of		
		ivinisters of		

			Health in a			
			one-year			
			period etc			
Content - content	Just A Pilot Process	Document	The 'what' or	"We created the	#4	
element is the 'what'		s and	the planning,	budget. We were		
of the policy process,		Interview	processes,	very good with		
and considers the		s	structure,	our budget. We		
conditions under			staffing,	were very		
which the policy			physical	careful. We were		
process occurred; the			space of the	fiscally aware,		
research, evidence,			Sheep River	fiscally		
and the rules,			NP Clinic.	responsible"		
regulations and						
legislation that				"But, there didn't		
influenced the policy				seem to be any	#9	
process. It considers				staff changes.		
the objectives of the				There seemed to		
policy, the evidence				be the same		
and the legislation				administration		
supporting a policy				people and the		
decision				same nurse		
				practitioners that		
				were there. So		
				no, I loved the		

				continuity of it,		
				yes"		
				"My simple		
				understanding	#18	
				was that even		
				though the		
				minister was		
				given several		
				letters and		
				briefing notes		
				and/or heads up		
				in the MDM		
				meetings which		
				are just meetings		
				between the		
				executive teams		
				and the ministry		
				with the minster		
				and her team that		
				it was not a		
				priority to them"		
Process - the 'how'	Just A Pilot Process	Document	'How' the			
element looks		s and	Sheep River	"The nurse		
specifically at policy			NP Clinic	practitioner		#22

development and the		Interview	came to be in	program was	
process of policy-		s	existence	developed as a	
making. This element			(producing	pilot program,	
assists in			the pilot	aimed at	
understanding how			project,	enhancing access	
public policy is made			funding of	to a primary care	
and implemented,			the pilot	provider in the	
whom it may			project,	Town of	
influence, and how it			description of	Okotoks during a	
is evaluated			the project).	period of	
				physician	
				shortage"	
	Details of Day to Day	Document	The basic		
	Operation	s and	elements		
		interviews	(structure,		
			funding,		
			relationships)		
			of the		
			CRPCN		
			organization		
		Document	The basic		
		s and	elements		
		interviews	(structure,		

		funding,			
		relationships)			
		of Primary			
		Care			
		Networks in			
		Alberta			
	Document	The	"I think it was a	#4	
	s and	process/cours	PCN decision.		
	Interview	e of action	And that's only a		
	s	and choices	gut feeling,		
		made which	nothing concrete.		
		influenced	And just		
		the closure of	knowing what I		
		the Sheep	know about the		
		River NP	leadership of		
		Clinic. How	PCNs. I think the		
		was the	government		
		decision to	would've been		
		close the	hard pressed to		
		Sheep River	intervene again		
		NP Clinic	as they had in the		
		made.	past. Even		
			though it was a		
			different		
			government, I		

Image: space of the space of			4.1.4.4		
Image: state in the state			think that they		
Image: second					
PCN board. That would be my read of it from the political will at the time" "So I don't think they were a bunch of evil pcople, I think they were trying to make a decision for the best use of their resources that they'd received.			to overturn the		
would be my read of it from the political will at the time" "So I don't think they were a bunch of evil people, I think they were trying to make a decision for the best use of their resources that they'd received.			decision of a		
Image: second			PCN board. That		
Image: series of the series			would be my		
Image: series of the series			read of it from		
So I don't think they were a bunch of evil people, I think they were trying to make a decision for the best use of their resources that they'd received.			the political will		
Image: space of the space of			at the time"		
Image: space of the space of					
bunch of evil people, I think they were trying to make a decision for the best use of their resources that they'd received.			"So I don't think		
Image: state in the state in			they were a	#1	
Image: state of the state of			bunch of evil		
Image: state in the state			people, I think		
decision for the best use of their resources that they'd received.			they were trying		
best use of their resources that they'd received.			to make a		
resources that they'd received.			decision for the		
they'd received.			best use of their		
			resources that		
			they'd received.		
And it got to the			And it got to the		
point where they					
no longer had the					
resources to					
support the clinic					

				out of the		
				envelope that		
				they received"		
Actors/Participants	Patient, Political and	Document	Individuals	4 NPs		
Actors are the	<b>Provider Champions</b>	s and	who directly	1 Other		
participants or the		Interview	participated	healthcare		
'who' element of the		S	or had	provider		
policy process. A key			interest in	1 Physician		
determinant of policy			regards to the	4 Organizational		
change is the group			Sheep River	Leaders		
of individuals			NP Clinic	8		
involved and these				Patient/communi		
often include:				ty members		
politicians,		Document	Individuals	" it was certainly	#1	
government and		s and	and	important to me,		
organizational		Interview	Stakeholders	I know there's		
employees and		s	who	many NPs that		
leaders, community			supported,	don't feel like		
members and various			defended or	CARNA did		
stakeholders. This			fought for the	enough to profile		
element asks who are				the NPs, but I		

the essential 'actors'			Sheep River	can tell you as		
involved in the policy			NP Clinic	President, I		
process, and who				ended up doing		
plays a key role in				more advocacy		
the planning,				for the Nurse		
implementation and				Practitioner role		
evaluation of the				than Not more		
innovation				but certainly as		
				much as for the		
				470 NPs in the		
				Province and the		
				37,800 RNs, and		
				I advocated for		
				both roles. But I		
				advocated just as		
				strongly for NP		
				role as I did for		
				the RN role"		
Sustainability of	Key Theme		Definition			
Innovation						
Framework						
Organization Factors	Insufficient	Document	Deficient	"Only on its	#3	
_	Consultation/engagement	s and	knowledge,	closing were		
Having flexibility		Interview	participation,	they (NPs) not		
and the ability to		S	involvement,	consulted. I		

adapt an innovation	interest or	know that, but I	
to support the	investment in	don't know	
organizational	the Sheep	anything else	
environment is	River NP	with their	
shown to encourage	Clinic	consultation. I	
sustainability. When	Process	believe they	
assessing factors that	and/or	were quite	
influence sustainabili	closure.	involved setting	
ty of a health care		it up because I	
innovation, questions		think that's the	
to consider include:		gist I remember.	
What are the		But I can say for	
established		sure that they	
communication		weren't involved	
pathways in the		in it's closing. I	
organization? What		think it came	
networking and		down like just a	
communication needs		big shutter,	
to occur to include all		"Done, we're	
elements of the		done, see ya."	
organization? What			
levels of government		"I mean I guess,	
and other		I don't live in	#17
organizational		Okotoks, I live in	
stakeholders should		High River, so	

be involved? and					maybe I didn't	
How can an					know about it in	
organization adapt					the community	
the innovation to fit					that if there was	
the local environment					anything going	
					on in the	
					Okotoks	
					community. But,	
					I just feel like I	
					wasn't consulted,	
					it was just, it's	
					just closing"	
	Innovation Barriers	Lack of	Document	Unable to		
		Transparency	s and	clearly or	"Data sharing	
			Interview	easily	between the	#9
			s	understand a	Department,	
				process; a	AHS and PCN	
				loss of trust	physicians is	
					limited, and	
					systems to do so	
					are not well	
					developed"	

This factor focuses		s and	al policies	say that the		
Political Factors -	Just a Pilot Process	Document	Organization	"Well, I would	#1	
				program"		
				the PCN		
				performance of		
				reflect on		
				PCNs and none		
				relate directly to		
				few of them		
				healthcare, but		
				area of primary		
				to the overall		
				measures related		
				a variety of		
				AHS each report		
				Department and		
				program. The		
				the PCN		
				performance of		
				on the		
				little information		#31
				"The Department provides very		#31

on appropriate			Interview	were in place	mandate of that		
government			s	or position to	clinic and the		
alignment, linking				support the	service they		
with regional health				Sheep River	provided were		
plans, involvement of				NP Clinic.	completely		
local and national					aligned with the		
champions as well as					directions in		
staff involvement.					Alberta health at		
Evidence indicates					the time"		
that a political focus							
on one particular							
policy can have a	Innovation Barriers	Lack of	Document	Fear,	"I think the	#2	
strong influence on		Accountability	s and	Uncertainty,	minister of		
the sustainability of			Interview	Obscurity,	Health should've		
an innovation. When			s	not clearly	stepped in"		
assessing factors that				understandin			
influence health care				g or	"He says he's		
innovation				withholding	approached		
sustainability,				knowledge	Health Minister		
questions to consider				surrounding	Sarah Hoffman		
include: Was the				the decision-	and was told		
policy aligned with				making	inquiries would		
government health				process of	have to come		
planning? Is there a				the closure of	from the		
pending change in				the Sheep	Wildrose		

government that			River NP	health critic and		
could impact			Clinic	when he raised		
sustainability? Who				the issue in the		
are the political				Legislature she		
'champions' that can				said it was the		
support the				first time she's		
innovation? and				heard of it"		
What are the funding						
provisions						
surrounding the		Document	The	"Well, we had a	#17	
policy?		s and	organization,	new government		
		Interview	political and	that didn't know		
		S	community	what they were		
			mood/opinio	doing. You had a		
			ns during the	new deputy		
			time the	minister that did		
			Sheep River	not have one day		
			NP Clinic	of healthcare		
			was open and	experience. You		
			in particular	had the oil drop I		
			closing	think by what		
				was it, 50% or		
				something insane		
				in less than a		
				year? You had		

	all of the sudden	
	this massive	
	deficit. You had	
	the loss of a	
	deputy minister	
	and a CEO both	
	of whom were	
	nurses. The CEO	
	of AHS was	
	booted out in a	
	way, like not	
	booted out but	
	was treated so	
	poorly that she	
	left"	
	"The problem	#15
	was that	
	occurred in the	
	change of	
	government, and	
	so, in that six	
	month period of	
	time, there was	

			three different		—
			health ministers"		
	Document	The response	"Because as soon	#1	
	s and	of the	as the		
	Interview	Provincial	government		
	S	Government	changed, and the		
		to the	new health		
		pending	minister took		
		closure of the	over, we started		
		Sheep River	trying to contact		
		NP Clinic.	the government		
			and getting		
			meetings and		
			trying to push it		
			through, and we		
			were told well,		
			we don't need		
			another nurse		
			practitioner pilot		
			project, and we		
			have no plan for		
			funding right		
			now, so it was		
			pushed and		
			pushed"		

			"My simple	#18	
			understanding		
			was that even		
			though the		
			minister was		
			given several		
			letters and		
			briefing notes		
			and/or heads up		
			in the MDM		
			meetings which		
			are just meetings		
			between the		
			executive teams		
			and the ministry		
			with the minster		
			and her team that		
			it was not a		
			priority to them"		
Patient, Political and	Document	Efforts made	"I think the	#10	
Provider Champions	s and	by	political will		
	Interview	organizationa	from decision		
	s	l, political	makers, whether		
		stakeholders	it be senior		

			and community members to prevent clinic closing."	bureaucrats or the actual minister of health or Associate Minister of health, to see it through, to see the nurse practitioner pilot project, fully funded and have a sustainable funding"		
Financial Factors - One of the most important factors influencing sustainability of a health care innovation are the financial factors. These	Funding Failure	Document s and Interview s	Funding that was obtained for the of the Sheep River NP Clinic	"With support from the PCN executive and physician leadership within the community, a decision was made to fund the	#18	

factors include the	NP project fo	r 1
service delivery	year"	
model and source and		
length of funding.	"It was a PC	N
When assessing	initiative, so i	it
factors that influence	was funded b	у
financial	the PCN"	
sustainability of a		
health care		
innovation, questions	"And, and I	
to consider include:	mean what ca	ın
What funding sources	be argued wa	s #10
have been identified	that you need	to
and secured? Is the	have a	
budget adequate to	sustainable	
meet the needs of an	funding mode	el
innovation running	dedicated	
long term? What will	towards nurse	2
happen when the	practitioners.	It
funding ceases? and	can't flow	
Is there any	through anoth	ner
consideration of	group's hands	3
program	first, flow	
sustainability in	through PCN	's
program planning	hands first or	

					through physicians hands first because then you're asking another profession to develop the nurse practitioner role in primary care"	
Workforce Factors -	The Right Model	NP Value	Document	Statements	"There was also	#8
Research indicates	The Right Would		s and	which	a benefit to them	<i>"</i> 0
that innovations			Interview	positively	as the NPs	
aligned with the			s	support the	provide coverage	
value and needs of				NP role and	for labwork	
employees are more				illustrate the	follow up when	
readily adopted and				value of the	some of the	
sustained. When				role	physicians are	
assessing factors that					away. All	
influence				Statements	physicians who	
sustainability of a				that reflect	were interviewed	
health care				the treatment	expressed	

innovation, questions		of NPs as not	confidence in the		
to consider include:		being valued,	level skill		
Are the values and		supported,	demonstrated by		
needs of employees		welcome or	the NPs and		
examined in the		insignifican	stated that they		
planning and		C	were confident		
implementation of			that the NPs are		
the innovation? Was			practicing safely:		
sufficient			they always ask		
recruitment,			for a second		
workforce and			opinion in more		
succession planning			complex		
assessed? Are			situations, and		
minimal changes to			know their		
roles, policies and			limitations really		
procedures expected?			well"		
and Is regular			wen		
communication with			"The problem		
employees			too is that we're	#3	
throughout the			not equal	π3	
process anticipated			players. When		
process anticipated			we're in a PCN		
			we're generally		
			employees. And		
			so that gives		

			none. So you really have no voice as an NP" "I still don't think that trying to integrate NPs into traditional primary care processes, which is now a PCN process, is the way to go for NPs. I think it's hopeless"	#3	
Innovation Specific	Innovation Barriers	A number of	" For me, The	#10	
Factors - Innovations that are flexible and adaptable, have the		different factors, when combined	Sheep River Clinic highlighted the		

ability to respond to a	together	systemic barriers				
dynamic funding and	prevented the	to integrating				
policy environment,	Sheep River	nurse				
and continue to	NP Clinic	practitioners into				
evolve. These	from being	the primary care				
characteristics are an	sustainable	system, even				
important feature of		though they're				
sustainability. When		well-integrated				
assessing factors that		and now they				
influence		have not for				
sustainability of		profits. Yeah,				
health care		and certainly the	#3			
innovations,		evidence, it's				
questions to consider		overwhelming				
include: How are		evidence"				
stakeholders involved						
in the innovation?		"What I saw was				
Will they be		a pattern that I				
consulted on the		had started to see				
acceptability, quality		and continue to				
and safety of the		see, and that				
innovation? What		wherever NPs				
barriers and supports		put in because				
to the innovation		there's a lack of				
exist? and Will an		physicians so				
evaluation process be				we're filler. As		
-------------------------	---------------------------	--------------	-----------------	--------------------	-----	--
initiated, and if so by				soon as positions		
who				can be found		
				we're replaced		
				with physician"		
				"That (barriers to	#10	
				keeping the		
				Sheep River NP		
				Clinic open) is a		
				multifactorial		
				issue"		
	Stakeholder Uncertainties	Lack of	Unable to			
		Transparency	clearly or	"I do find there's	#3	
			easily	a significant lack		
			understand a	of learning, and		
			process; a	this goes across		
			loss of trust	all occupations,		
			in a process	there's a lack of		
			Stakeholders	learning between		
			that had little	health		
			awareness or	occupations as to		
			understandin	what one		
			g of the facts	provider		
			in the Sheep	provides in		

	River NP	relation to	
	clinic process	another. I find it	
		quite amazing.	
		"I don't know. I	
		am pretty sure it	
		was a	#15
		government	
		funded thing, but	
		I don't know	
		where from"	
	Statements	"I thought it was	#18
	that the	a really effective	
	Sheep River	model of care	
	NP clinic	and actually	
	was an	fulfilled the	
	advanced	government's	
	novel/new	directions around	
	idea	optimizing nurse	
	introduced in	practitioner	
	Alberta; was	workforce"	
	the first		
	clinic of its		

		kind to open	"But the reality		
		in Alberta.	is that even	#18	
			under a PC		
			government that		
			tends to be		
			physician		
			friendly they		
			took the risk to		
			actually		
			implement an		
			alternate primary		
			care provider to		
			better meet the		
			needs of the		
			community. That		
			to me was an		
			innovation in		
			itself. Then I		
			think those NPs		
			as kind of		
			pioneers in that		
			particular		
			community and		
			region in the		
			province I think		

			established their		
			service delivery		
			model to really		
			meet the needs		
			of the		
			population"		
The Right Model	NP Value	Statements		#1	
		which	"We have our		
		positively	own role to play.		
		support the	We are not to fit		
		NP role and	in with		
		illustrate the	physicians"		
		value of the			
		role.	"The problem		
		Statements	too is that we're		
		that reflect	not equal		
		the treatment	players. When	#13	
		of NPs as not	we're in a PCN		
		being valued,	we're generally		
		supported,	employees. And		
		welcome or	so that gives		
		insignifican	somebody else		
			all the power and		
			employee has		
			none. So you		

			really have no voice as an NP"	
<b>Profound Social Proof -</b>	Social Proof	Statements	"To be fair we	
Successful/Sustainable	Successful/Sustainab	which	started out as a	#4
	le	demonstrate	pilot; a pilot that	
		and validate	kept running for	
		that the	five years	
		Sheep River	because we were	
		NP clinic	effective"	
		was		#1
		sustainable	"I certainly did	
		and effective.	not see, hear or	
			know of any	
			evidence that the	
			Sheep River NP	
			clinic wasn't	
			anything but an	#1
			unmitigated	
			success"	
			"Here was a	
			model within a	

ſ			PCN that was	
			doing brilliant work"	
			work"	

# Table 4-6: Emerging Theme Definitions to Develop a New Conceptual Framework to AssessSustainability of Innovations in PHC

Emerging Themes	Key Category	Informed By	Definition	Quotes	Participa nt #	Doc #
Emotions/Aftermath	Impact	Documents/Interviews	The	"Losing the NP clinic was		
			unpleasant	equivalent to having two	#9	
			emotions,	physicians move out of town"		
			aftereffects			
			and	"DevastatedI felt like part of	#15	
			consequences	me was missing"		
			of the Sheep	"I was left high and dry"		
			River NP		#13	
			Clinic closing.	"All of us patients - we were just		
				sort of scattered"		
					#18	
				"Devastated, it's a step		
				backwards"		
				"I'm not sure but I think these		
				people are caught in the middle	#3	
				of something that's very tragic,"		

				"And I can't believe they did that. I mean that's unethical. You don't just change providers on people and not ask if that's okay, or "would you mind, or this is what we're doing." It's just, "Okay, well sorry they're out, these guys are in." I just think that's abysmal. I would hate to be treated like that as a patient, so disrespectful."		
Professional Turf	Challengers	Documents/Interviews	Challengers -	"Yes. They (senior	#17	
This is an amousing them.			Individuals	administration through our health		
This is an emerging theme because it is was found to			who were in	organizations) would do anything to put obstacles in the way and to		
have a great impact on the			opposition, acted against	support the physicians not the		
sustainability of this clinic. It			or were an	NPs. Yes. Yes. Yes"		
is a very specific (and often			obstacle in the	111 5. 105. 105. 105		
under looked) component			Sheep River	"I wouldn't say it's jealousy. They		
that can greatly influence				just don't want us, it's just an		

and impact the sustainability			Clinic clinic	animosity that we're cutting into		
process and it is not			process.	their paycheck and their wallets"	#13	
specifically addressed in						
either one of the existing						
frameworks.	Protectionism	Documents/Interviews	Professional	"Physician's groups are old boys	#18	
	(Fear)		Groups,	clubs in rural Alberta"		
			Organizations			
			and/or	"But it's self-interest and I don't	#3	
			individuals	really blame them. Boy, if		
			who initiated	somebody had handed me that		
			measures that	package too (PCN) I'd be		
			restricted or	protecting it. I mean can		
			protected	completely really understand, it's		
			themselves	just that nurse practitioners are		
			against	taking the brunt, we're just not		
			competition	welcome, and that's the hurtful		
			from NPs	part, it's a social injustice part,		
			based upon	right, it's an injustice, but I'm not		
			fear of losing	surprised that's why I think we		
			status,	need something completely		
			income, power	separate"		
			etc.			
				"Physicians and the	#13	
				administration of the Health	1110	
				Authority at that time, who never		

Power and	Documents/Interviews	Professional	<ul> <li>wanted nurse practitioners</li> <li>because they wanted to keep the docs happy, and they didn't want us"</li> <li>"There was tons of politics where the docs were truly undermining the spread and scale up of NP role integration in the province yes. Tons"</li> </ul>		
Positional		groups,	maelstorm of	#18	
Authority		Organizations	organizational/political and		
(Influence		and/or	professional power/lobby and		
		individuals	corruption"		
		who use their			
		title, position	"I think NPs forgot they are		
		and power to	pioneers in a very NP resistant		
		exert a degree	environment, where docs have	#13	
		of influence	way more power and are doing		
		on the Sheep	an invisible lobby and sometimes		
		River NP	even visible"		
		clinic process.			

			"I suspect the doctors The		
			#13Physician's Association put		
			pressure on the Government to		
			protect their own interests"		
Technicolor Data	Profound social	Detailed,	"So it was not at all opaque, the	#1	
This is an emerging theme	proof	'colorful',	difference that they'd made in		
because even though they		meaningful	their community. They had the		
had impeccable data		data, through	data, right there in technicolor,		
collected on this clinic) that		the evaluation	you could see the difference that		
illustrated very positive		of the Sheep	they'd made in their community"		
outcomes), the clinic was		River NP			
still not sustainable. The		clinic and			
current frameworks do not		collected by			
address this specific		the NPs			
component. We are NOT		themselves			
currently using evidence or		which			
outcomes to support the		illustrated the			
sustainability process. It is		success and			
not specifically addressed in		sustainability			
these frameworks. This needs		of the clinic.			
to be reflected in the new		Profound			
framework.		social proof,			
		the influence			
		of words and			

actions from	
people around	
us that	
positively	
impacted the	
success and	
sustainability	
of the Sheep	
River NP	
Clinic.	

# Table 4-7: Participant Characteristics

Stakeholder Group	Perspective	Gender	Discipline	Role
Internal Stakeholders	Organizational n=4	Female n=8	Nursing n=4	Healthcare providers n=6
(those who participated	Professional n=6	Male n=2	Medicine n=1	Healthcare Management
in some capacity in the	Community/Patient n=0		Healthcare n=5	n=4
Sheep River NP Clinic				
Process)				
External Stakeholders	Organizational=0	Female n=7	N/A	Community
(Those who had	Professional=0	Male n=1		Member/Patient n=8
minimal knowledge of	Community			
the Sheep River NP	Member/Patient n=8			
Clinic Process but may				
have been involved in				
another capacity)				

# Table 4-8: Document Summary

#	Date	Title	Author	Source	Туре
1	N.D. (No Date)	Calgary Rural Primary Care	Calgary	Calgary Rural	Operating Strategy
		Network Operating Strategy	Rural	Primary Care	
			Primary	Network	
			Care		
			Network		
2	N.D.	Letter addressed to Minister of	Community	Personal	Letter
		Health Honorable Stephan	Member	Documentatio	
		Mandel		n	
3	N.D.	Letter addressed to Minister of	Community	Personal	Letter
		Health Honorable Stephen	Member	Documentatio	
		Mandel		n	
4	N.D.	Letter addressed to Minister of	Community	Personal	Letter
		Health Honorable Stephan	Member	Documentatio	
		Mandel		n	
5	N.D.	Letter addressed to Minister of	Community	Personal	Letter
		Health Honorable Stephan	Member	Documentatio	
		Mandel		n	

6	9/11/10	Primary Care Service Charter:	Calgary	Calgary Rural	Project Charter
		Family Practice NP Pilot	Rural	Primary Care	
			Primary	Network	
			Care		
			Network		
7	27/10/11	Okotoks nurse practitioner clinic	Journalist	Okotoks	Newspaper Article
		first of its kind in Alberta		Western	
				Wheel	
8	2012	<b>Evaluation Family Practice</b>	Calgary	Calgary Rural	Evaluation
		Nurse Practitioner Clinic in	Rural	Primary Care	
		Okotoks	Primary	Network	
			Care		
			Network		
9	2012	Office of the Auditor General	Auditor	Office of the	Report
		Report of the Auditor General of	General	Auditor	
		Alberta		General	
		on Primary Care Networks			
10	2013	Calgary Rural Primary Care	Calgary	Calgary Rural	Organization structure and chart
		Network Organizational	Rural	Primary Care	
		Structure	Primary	Network	
			Care		
			Network		

11	01/02/13	Family Practice NP Initiative -	Calgary	Calgary Rural	Project Charter Update
		Updated Charter	Rural	Primary Care	
			Primary	Network	
			Care		
			Network		
12	24/04/15	Letter addressed to Minister of	Community	Personal	Letter
		Health Honorable Stephen	Member	Documentatio	
		Mandel		n	
13	15/05/15	Letter addressed to Minister of	Community	Personal	Letter
		Health Honorable Sarah	Member	Documentatio	
		Hoffman		n	
14	06/11/15	Calgary Rural Primary Care	Sheep	Sheep River	Letter
		Network Letter Sheep River	River	Nurse	
		Nurse Practitioner Clinic Closure	Nurse	Practitioner	
			Practitioner	Clinic	
			S		
15	01/12/15	Video MLA in	Highwood	same	Video
		Alberta Legislature	MLA		
16	01/12/15	Member Legislative Assembly's	Community	Facebook	Public Social Media Post
		Facebook Post on Facebook	Member		

17	01/12/15	Facebook Comment on Member	Community	Facebook	Public Social Media Post
		Legislative Assembly's	Member		
		Facebook Post			
18	07/12/15	Nurse Practitioner's Office in	Journalist	High River	Newspaper Article
		Sheep River Clinic to Close		Online	
		https://okotoksonline.com/local/s			
		heep-river-clinic-to-close			
19	07/12/15	Comment on article "Sheep	Community	Facebook	Public Social Media Post
		River Clinic to Close"	Member		
20	07/12/15	Comment on article "Sheep	Community	Facebook	Public Social Media Post
		River Clinic to Close"	Member		
21	08/12/15	Nurses Looking For Work	Journalist	Okotoks	Newspaper Article
		Following Sheep River NP		Western	
		Clinic Closure		Wheel	
22	12/02/15	Calgary Rural Primary Care	Physician	Calgary Rural	Letter
		Network letter to Minister	Lead	Primary Care	
		Mandel	Calgary	Network	
			Rural		
			Primary		
			Care		
			Network		

23	16/12/15	Lack of funding closes Okotoks	Journalist	Okotoks	Newspaper Article
		clinic		Western	
				Wheel	
24	24/12/15	Comment on article "Okotoks	Community	Facebook	Public Social Media Post
		Nurse Practitioner Clinic Could	Member		
		Have Been Saved if Policy was			
		in Place"			
25	02/01/16	Letter addressed to Minister of	Community	Personal	Letter
		Health Honorable Sarah	Member	Documentatio	
		Hoffman		n	
26	14/01/16	Comment on article "Okotoks	Community	Facebook	Public Social Media Post
		Nurse Practitioner Clinic Could	Member		
		Have Been Saved if Policy was			
		in Place"			
27	14/01/16	Comment on article "Okotoks	Community	Facebook	Public Social Media Post
		Nurse Practitioner Clinic Could	Member		
		Have Been Saved if Policy was			
		in Place			
28	14/01/16	Comment on article "Okotoks	Professiona	Facebook	Public Social Media Post
		Nurse Practitioner Clinic Could	1		
		Have Been Saved if Policy was	Association		
		in Place			

29	24/01/16	Comment on article "Okotoks	Professiona	Facebook	Public Social Media Post
		Nurse Practitioner Clinic Could	1		
		Have Been Saved if Policy was	Association		
		in Place"			
30	19/09/16	Calgary Rural Primary Care On	Researcher	Calgary Rural	Email
		Site Visit		Primary Care	
				Network	
31	2017	Office of the Auditor General	Auditor	Office of the	Report
		Report of the Auditor General of	General	Auditor	
		Alberta		General	
		on Primary Care Networks			



Motivation	Participant Response
Change in local community decision making power	"Because local decision making was taken away in Okotoks, the centralized approach was
	detrimental to any local health initiatives (even if they were deemed successful). The PCN
	lost its ability to have discretionary decision-making" (Participant 11; Health Care
	Professional)
Physician pressure and lobbying	"I think the physician lobby to invest in physician roles and PCN, the new deputy minister
	who was more focused on relationship management with physicians" (Participant 18;
	Health Organization Leader)
Lack of awareness and marginalization; of the NP role	"We need to practice to our full scope of practice. We need to be recognized for that and
	what we can contribute to the welfare of people, and their health needs. We are to
	work as partners. We are not there to be competition. The focus is always the patient"
	(Participant 13; Health Care Professional)
Lack of stakeholder, community and patient consultation	"and that's the biggest horror of all is that none of the patients were asked if that was okay
	with them. I think I was just so disgusted. I think that breach is everything about patient
	provider relationships. And I can't believe they did that. I mean that's unethical. You don't
	just change providers on people and not ask if that's okay, or "would you mind, or this is
	what we're doing." It's just, "Okay, well sorry they're out, these guys are in." I just think
	that's abysmal. I would hate to be treated like that as a patient, so disrespectful"
	(Participant 3; Health Care Professional)

Economic slowdown	"I think we need to put the oil drop [oil prices decreasing significantly in Alberta] into	
	timeline - provincial economic stability when this clinic was in the process of closing."	
	Participant 18, Health Organization Leader	

Innovation Barriers	Participant Response
Lack of identified long term funding	"Had the government said, you know what- this is really innovative PCN. We need you to keep
	this going, so we're going to create a funding model that allows you to continue this
	"(Participant 1, Professional Organization Leader)
Lack of patient and stakeholder consultation	"I suppose if the patients would have gotten notice that there was possibility that it could
	have been closed. I just kind of felt like I got the letter that said "We're closing" like a month
	from now or two weeks from now or whatever it was" (Participant 17, Patient)
Lack of health policy alignment to support a funding model for the NP role	"The Barrier - It just was the politics and the money" participant 6, Health Care Management
	Well, I think the government should pay for quality and not [a healthcare professional who]
	provides for one thing, We are getting more and more to the point where with can tell physicians
	based on data that they're not doing a good job" (Participant 6, Health Organization Leader)
Lack of organizational (CRPCN) accountability	
	"My simple understanding was that even though the minister of health was given several letters
	and briefing notes and/or heads up in the MDM meetings which are just meetings between the
	executive teams and the ministry with the minister and her team that it was not a priority to
	them"(Participant 18, Health Organization Leader)
Professional Opposition.	"Barriers - like I said I think the physician lobby to invest in physician roles was a barrier"
	(Participant 18, Health Organization Leader)

# Chapter Five: Summary and Synthesis

Raelene Marceau Dr. Kathleen Hunter Dr. Stephanie Montesanti Dr. Tammy O' Rourke

## **Chapter Five: Summary and Synthesis**

The overarching purpose of this thesis was three-fold. Firstly, to assess the state of the literature in regards to the closure of clinics utilizing NP services. Secondly, to critically analyze current NP funding models across Canada to provide guidance on improving funding mechanisms to sustain NP roles. Lastly, to examine an NP clinic closure using case study design to generate understanding on sustaining the NP role and PHC innovations. This series of studies establishes a foundation and direction for development of a program of research in the area of sustainability of PHC innovations. This thesis is the first work to explore the closure of NP clinics in PHC settings. The findings contribute to the literature, inform future innovation assessment and planning, and have the potential to enhance governments, funders and policy-makers understanding of the elements that contribute to NP role sustainability and sustainability of other important PHC innovations.

In Chapter Two, a scoping review was undertaken to examine the clinic closures with NP services as the main provider. The aim of this review was to explore evidence in three main areas: a) the factors which influence decisions to close clinics that include NPs; b) factors which influence the sustainability of PHC programs and services and; c) factors which influence the sustainability of the NP role in PHC. The key findings of this review demonstrate a lack of published, peer reviewed research to advance the understanding of clinic closures specifically using NPs; limited evidence of peer reviewed research pertaining to sustainability of PHC innovations, and a dearth of peer reviewed, published research informing sustainability of the NP role in PHC. A summary of the state of the literature on sustainability of the NP role and other PHC programs and services was presented, including existing gaps in knowledge to provide direction for future research.<sup>1-10</sup>

Chapter Three, a critical analysis of the current state of Canadian NP funding models using the Sustainability of Innovation Framework was presented.<sup>2</sup> This chapter emphasizes the challenges of sustaining the NP role in Canada and the current structure of funding models for NPs. The findings demonstrate that there is variability of jurisdictional funding initiatives and policies that can influence sustainability, Canadian jurisdictions need to reform NP funding approaches and transition to evidence based, flexible funding models that support PHC reform and respond to variations in NP practice and patient panels.<sup>11-17</sup> Findings also illustrate that NP wage and benefit compensation need to be harmonized, interprofessional teams utilizing NPs must be expanded and a coordinated national NP workforce strategy is necessary. Governments, funders and policy makers must consider the influence of existing outdated policies and adopt new and innovative solutions to funding for Canadian NPs to ensure sustainability of the NP role and to further support and advance needed PHC reform.<sup>18</sup>

In Chapter Four, the Sustainability of Innovation (Fox et al., 2015) and Policy Triangle Frameworks are used to inform an exploratory single case study of a NP clinic closure in Alberta, Canada.<sup>2,19,34,35</sup> The contextual factors influencing and the decision-making process surrounding clinic closure were examined. This chapter provides an in-depth explanation of this NP clinic's sustainability failure. Multiple contextual factors influenced clinic closure including: lack of stakeholder and patient consultation, lack of accountability and transparency in regards to clinic planning, frequent government leader changeover, lack of policy alignment for planning, implementation and evaluation of the clinic, and a rapidly changing economic landscape. Data from participants and documents confirmed that the Sustainability of Innovation and Policy Triangle Frameworks, while thorough in their assessment of factors influencing sustainability, did not completely capture the complexity of sustaining innovations in PHC environments such as primary care. In its entirety, this thesis has made important contributions to knowledge, theory, practice and policy and the contributions of these areas are important to note.

# Contributions

## **Knowledge** Contribution

This thesis contributes substantially to the state of the science on sustainability of the NP role and other innovations in PHC in the following ways: 1) demonstrates a paucity of peer reviewed evidence on clinic closures involving NP services, as well as the sustainability of innovations in PHC including the NP role, 2) demonstrates through critical analysis there is wide variation in NP funding models across Canadian jurisdictions, as well as frequent use of short term 'pockets' of funding to fill gaps in health services rather than establishing flexible funding models to suit a wide variety of NP practice environments and patient populations, and 3) the identification of three new themes in the case study as additional concepts to consider when examining the sustainability of any innovation in PHC. These themes demonstrate the

complexity and evolving nature of the study of sustainability and have been utilized to inform construction of a new conceptual framework to assess sustainability of PHC innovations. The practice and policy contributions are discussed, after which a newly visualized theoretical framework, and its significance for knowledge is presented.

# **Practice Contribution**

Over a decade ago, Dr. Paul Uhlig described NPs as a 'disruptive innovation' with the potential to be a catalyst in achieving a transformed health system and for the past five decades, researchers have consistently validated the safety and quality of care provided by NPs.<sup>21-27</sup> Despite this evidence, the NP role remains absent or underutilized in some Canadian jurisdictions and NPs in this country continue to struggle with issues that prevent optimization of the role. <sup>12,24,26,28,29</sup> The NP role continues to be labelled as an innovation thought to be important to achieving PHC reform.<sup>12</sup> The concept of innovation; a new method or idea having influence and impact, is one that has been central to the NP profession since it was established.<sup>12</sup> However, decades later we are no closer to sustaining a stable environment for the NP role. This implores an important question; is the NP role still an innovation today or would it be better labelled as is a health care improvement that has failed to be sustained.<sup>30</sup> Innovation does not occur until there is a break from the current paradigm on which the product or health service innovation is based. Only then can we consider it an innovation.<sup>30</sup> There has been incremental changes across Canada in terms of legislation, regulation and the numbers of practicing NPs has increased substantially.<sup>12</sup> Yet, we have failed to achieve the full health system paradigm shift necessary to label the NP role an innovation. Limitations of outdated and entrenched health policies, a lack of policy action and a lack of evidence informed policy design, continue to impact NP role optimization across Canada.<sup>31</sup>

This thesis demonstrates the literature informing sustainability of PHC innovations such as the NP role is lacking, and what does exist is poorly organized and thus not suited to informing evidence-based policy and practice. Our failure in understanding the topic of sustainability, has resulted in a primary care model that is unsustainable and an overall sustainability failure of the NP role. This thesis moves the meter forward in terms of providing a critical lens and evidence on what is needed to improve sustainability of the NP role in PHC.

# **Policy Contribution**

The notion that health service innovations and organizations should learn from failures small and large; has obvious appeal. Yet health care organizations that effectively learn from failure are relatively rare.<sup>32</sup> Pervasive barriers embedded in health care organizational systems make learning from failure difficult.<sup>32</sup> The inability to effectively learn from failures in policy is highlighted in this thesis. The paucity of peer reviewed literature on clinic closures involving NPs, the limited literature on the topic of sustainability of PHC innovations such as the NP role, and the gaps in the literature informing health care sustainability, illustrates the challenges of studying failure in health systems.<sup>9</sup>

In terms of policy, this dissertation demonstrates the following: a) literature suggests that there has been a failure by federal and provincial governments and policy-makers to ensure sustainability of the NP role in Canada as a supportive element in PHC reform, b) in the case study of sustainability failure of a PHC innovation, closure occurred despite conflicting stakeholder support, evaluation success and supportive outcome evidence, c) there was failure to utilize evidence informed policy and processes to support the planning, implementation and evaluation of this innovation in PHC, d) there was failure of governments, funders, policy makers and organizations to engage, be accountable and transparent in regards to decision making processes in this case, d) there was a failure of a consistent health policy approach throughout various election cycles and government changeovers, e) most importantly, a failure of governments, funders and policy-makers involved in this case to ethically consider how sustainability failure impacted patients, stakeholders, communities, organizations, and the health system as a whole, and f) finally a failure to engage in patient centred accountability on the part of governments, funders and policy makers in regards to the long-term sustainability of the Sheep River NP Clinic. An enhanced understanding of sustainability failure in this thesis points to a lack of policy response, policy inaction and policy failure on behalf of governments, policymakers and funders in Canada.

## **Minimizing Policy Failure: Recommendations for Action**

Currently, the policy context is understood to be much more complex than previously recognized. It is imperative that governments, funders and policy-makers discover how to minimize policy failure and use it to inform future endeavours. It is important to understand how past sustainability failures can inform future innovation planning and how governments, funders and health policy-makers can utilize expert evidence to inform decision making throughout the process.<sup>39</sup> Embedding a culture of effective policy learning to improve health policy design and delivery is necessary in order to sustain innovations in PHC. <sup>36–38,41,42</sup>

# **Evidence** and **Evaluation**

To minimize policy failure, there is an urgent need for longitudinal studies that systematically evaluate the introduction of health service innovations over time. This is to better understand the impact of factors that receive little attention on how they affect the sustainability process. There is a demand to systematically assess what sustainability strategies will work best in what context and under what conditions, keeping the complexities involved in mind. This requires an adequate understanding of the 'politics' surrounding the various processes, and in particular, the power relationships between the various stakeholders involved.<sup>33</sup>

To avoid policy failure, prolonged evaluation of health service innovations is fundamental to enable sustainable implementation and wider spread. It may also be necessary to frame research evidence differently to convince various stakeholders of its effects. Such reframing is especially important in times of resource constraints, where perspectives on evidence may emphasize cost savings or efficiencies.<sup>33</sup>

# Pilot 'Projectitis'

Sweeping policies such as those that support the Canada Health Act (CHA) and are formulated at the national level need some degree of consistent delivery at other jurisdictional levels.<sup>33</sup> This is especially true in Canada, where the jurisdictional governments have a separate degree of political authority and the general guidelines of the CHA are reinterpreted to fit into local contexts. Jurisdictions often use pilot projects as a method of testing a particular health policy to ensure it is the correct solution before investing significant amounts of money. While pilot projects are worthwhile, their benefit may only extend to a small group, ceases on

completion of the project, and the lessons learned from the pilot often fail to translate beyond the final report to sustainable outcomes. This thesis illustrates there is a need to translate the lessons learned from pilots into key principles, generalizable beyond a few pilot projects, to inform health policy and health service planning.<sup>33</sup> This thesis also demonstrates that it is necessary to account for the variability of jurisdictional initiatives and policies that could influence the sustainability of any pilot project and the planning of pilot projects needs to include anticipated long term implementation based on the evaluation.

# **Government Accountability and Transparency**

Government and organizational leaders tend not to be held accountable for the outcomes of their policy decisions. <sup>41</sup> In the event of policy failure, it is likely that they will have moved on or out of office. Consequently, they are easily attracted to the prospect of short-term results. This can lead to the pushing through of policies quickly, rather than getting involved in the protracted and frustrating details of how things might work in practice.<sup>41</sup> Often a culture of protectionism shields government and organizational leaders from meaningful scrutiny when a policy underperforms or fails.<sup>42</sup> Stronger accountability is needed to counterbalance the ambiguity and unavoidable uncertainty of decision making.<sup>42</sup> It is not enough to identify who is responsible for the decision once failure becomes apparent; there needs to be proactive methods which offer assurance that the responsible individuals have properly considered the risks associated with major projects before they begin.<sup>42</sup>

Competent governments and organizations are led by open and transparent leaders. There is a growing consensus that transparency lies at the heart of effective governments and organizations, and that this is an essential ingredient of 21st-century policy-making.<sup>42</sup> A transparent government or organization is described as being open about their actions, willing to facilitate accessibility of information and responsive to new ideas, demands and needs. Together, these 'building' blocks of transparency are seen to support a number of benefits: improving the evidence base for policy making, lessening the risk of policy failure, strengthening integrity, and building public trust in the policy process.<sup>43</sup>

Governments, funders, policy-makers and organizational leaders must establish prerequisites for accountability and transparency when making important policy or program decisions.<sup>44</sup> To conclude this section, these officials need to provide information that enables

stakeholders to understand the consequences of any decision and to be aware of the responsive actions taken by government when these deviations occur.<sup>44</sup> They need to be responsible for ensuring that legislation, regulations and commitments are made public, which enables stakeholders to understand who is accountable for what. These same officials must disclose decisions and results, which enables stakeholders to know whether or not agreed standards and commitments have been met. Particularly, the publication of government decisions, meeting records, audit reports and monitoring and evaluation data, which provide the information to examine whether or not set outcomes have been achieved.<sup>44.</sup>

#### Utilizing an Ethical Lens and Person-Centred Policy Approach

In addition to the two previous recommendations, there must be a consideration to ethics in the use of public health funds. Governments, funders and policy-makers may not consider the ethical consequences of building and implementing an innovative program without sustainable funding.<sup>45</sup> Organizational and political bodies must use an ethical lens and consider the implications as to whether it is ethical to develop interventions without adequate support to sustain them. Governments and policy-makers must also pay attention to ways in which a person-centred focus can be implemented in the decision-making process of sustaining PHC health services innovations. A person-centred approach is a key component of quality in health care systems but difficult to achieve in practice.<sup>46</sup> Evidence suggests that person-centred participation is a fundamental component for designing effective and sustainable health care systems and this premise can be applied to sustaining health service innovations. Utilizing a person-centred approach in the health policy process requires the engagement of patients as equal partners in decision making in order to ensure that their health needs and perspectives are incorporated into the health policy agenda. Increased person-centred involvement in health policy decision making in regards to sustaining innovations in PHC is associated with improved quality, safety, cost-effectiveness and overall better health outcomes. In this reasoning, person centred involvement is indispensable in the development and implementation of health policies, and the sustaining of health innovations in PHC.<sup>46</sup>

This thesis demonstrates it is necessary to develop further understanding and guidance on evaluating the ethics of emotional responses to sustainability failures and patient engagement in sustaining innovations such as the NP role in PHC. Schell et al, noted that health service innovations able to sustain themselves are more likely to produce outcomes that are lasting and healthier.<sup>47</sup> Maximising the benefits accrued from studying the sustainability of health service innovations is an essential task for researchers and health care planners. A number of reasons have been advanced to explain the limited focus on the sustainability of health service innovations in PHC.<sup>45</sup> Key among these is the fact that sustainability is often conceptualised as a final phase of program development after planning, implementation and evaluation phases. Many current frameworks to assess sustainability do not take into consideration the importance of sustainability in designing health service innovations. Pluye et al, argue that health service innovation planning, implementation and sustainability are concomitant processes and advocate for the re-conceptualization of sustainability to ensure that it is integrated into the design of interventions and planned for in advance.<sup>48</sup> One key recommendation to improve sustainability of health service innovations, is the use of conceptually and theoretically informed approaches to guide the design, development, implementation, evaluation and sustainability of these innovative health service.<sup>45</sup>

The purpose of this next section is to propose a new conceptual framework, based on three new findings that emerged from a case study of the closure of a NP clinic in Alberta Canada. This new conceptual framework is intended to guide organizations, governments, funders and policy-makers through an early assessment process to better achieve sustainability of health service innovations in PHC.

#### **Theoretical Contribution**

#### Primary Health Care Sustainability of Innovation Assessment Framework

Three new concepts emerged in the case study presented in Chapter 4. These concepts are not currently recognized or included in the assessment of the sustainability of innovations in either the PTF or the SIF. These frameworks are lacking the following: utilization of evidence informed policy and processes, consideration and understanding how stakeholder emotions are affected by sustainability failures, and an understanding how protectionism, fear and individual/organizational opposition impacts sustainability of health services innovations.<sup>36-38,41</sup> The core factors of the SIF and PTF and the emerging data identified in this dissertation, have

been incorporated into a new conceptual sustainability framework. A conceptual framework is defined as the end result of bringing together a number of related concepts which are joined together to illustrate a larger map of possible relationships.<sup>49</sup> The Sustainability of Primary Health Care Innovations Assessment Framework takes the SIF, a theoretical framework suitable for the examination of sustainability of any health service innovation, and utilizes the five factors that impact sustainability, refining them for further understanding and clarity. It merges the SIF with the healthy policy concepts of the PTF in consideration of the broad policy process that is often going on 'behind the scenes'.<sup>2,50</sup> This new envisioned conceptual framework adds the three emerging themes (concepts) identified in the case study to the five factors from the SIF for a total of eight contextual factors that should be considered when assessing the sustainability of innovations in PHC. The purpose of this framework is to help create a shared understanding of the components necessary to assess sustainability amongst governments, funders, policy-makers, practitioners, researchers and evaluators.

# **Contextual Factor 1: Political**

This factor focuses on the small and large political forces that may impact the innovation. This includes considering whether election cycles and government changeover are occurring, is there appropriate government policy alignment, is the innovation linked with regional and organizational health plans, is there accountability and transparency in regards to the decision making surrounding the innovation, and does the innovation involve stakeholder input; local and national champions and staff and community involvement. This factor also focuses on who are the champions, advocates and early adopters of the innovation.<sup>2,50</sup>

# **Contextual Factor 2: Organizational**

This factor focuses on organizations having a sound policy process to assist with health service innovation decision- making at the planning, implementation and evaluation phases of a project. Organizations must have adaptability and be responsive to any changes the innovation will bring forth. They must have established communication pathways and must be transparent in conversations at all levels within the organization. This includes conversations and engagement with organizational and non-organizational stakeholders.<sup>2,50</sup>

# **Contextual Factor 3: Financial**

This factor focuses on clearly outlining a funding model, a source of funding, the amount and length of funding from innovation outset. This factor also focuses on assessing long term fiduciary viability of the innovation in the planning phase and not when the innovation funding term is about to end. Financial viability requires stakeholder input and involvement from the outset to ensure that all potential sources of funding have been identified and exhausted.<sup>2,50</sup>

# **Contextual Factor 4: Workforce**

This factor focuses on health service innovations that have a clear human resource and workforce strategy including recruitment, retention and workforce planning succession. The workforce factor also brings attention policies to support regular, clear, concise and transparent communication pathways with organization employees, including policies and procedures to support organizational roles, structure and function.<sup>2,50</sup>

# **Contextual Factor 5: Innovation Specific**

This factor focuses on creating a policy and organizational environment that supports the innovation. It asks the questions: how complex is the innovation and the environment that the innovation is being introduced too? Is the innovation flexible and adaptive enough to respond to a dynamic funding, policy and organizational environment, and continue to evolve? How are stakeholders involved in the innovation and what communication pathways can be formulated to be transparent and clear about the innovation. What additional barriers and supports to the innovation exist? .<sup>2,50</sup>

## **New Contextual Factors**

The new conceptual model incorporates the three emerging themes from thesis research and labels and defines them as additional factors to be included in the assessment of the sustainability of an innovation in PHC.

# New Factor 1: Evidence

This factor focuses on utilizing an evidenced informed policy process to support sustaining an innovation and avoid failure. Evidence includes utilizing sound evaluation strategies and asking what evidence exists to support the innovation. As demonstrated by previous literature; evaluating innovations longitudinally, not just a short window in time, is an important feature of sustainability of health service innovations.<sup>41</sup>

#### New Factor 2: Challengers

Challengers are those individuals who are fiercely opposed to the innovation. They use protectionism, lobbying and power to influence decision making regarding innovation. The health care sector has many stakeholders, each with an agenda. Often, challengers have substantial resources and power to influence public policy and opinion by challenging the innovation. Identifying the challengers of the innovation in the planning stage is an important feature of sustainability. Learning to anticipate challengers and preparing a plan to deal with opposition to health service innovations is a crucial part of the sustainability assessment process.<sup>51</sup>

# New Factor 3: Emotion Aftermath

An often-overlooked aspect of sustaining an innovation is factoring stakeholder, patient and community member's emotions in assessment of sustainability. Individuals, groups, organizations; including key stakeholders, do not make decisions on facts alone. Emotions drive behaviors and decision-making. Ask how key individuals, groups and stakeholders feel about the innovation, about the costs involved, about how they will feel if the innovation fails.<sup>52</sup>

#### New Conceptual Framework Flow

Researchers have conceptualized sustainability as the final stage in the life cycle of an innovation and it has often been measured as an outcome instead of a process.<sup>53</sup> Limited research has focused on sustainability as a set of processes that occur in the innovation life cycle and evidence now emphasizes the importance of early innovation sustainability planning.<sup>53</sup> Current

research demonstrates that sustainability is likely influenced by early implementation processes, but that conceptualizing sustainability as a process does not emphasize the longitudinal perspective needed for sustainability research. The Primary Health Care Sustainability of Innovations Assessment Framework has been created as an assessment tool and can be utilized during the planning, implementation and evaluative component of any health service innovation. The framework focuses on early assessment of the contextual factors influencing sustainability of a health service innovation and utilizes the outcome of early sustainability assessment as a guide for implementation and evaluation. The framework flows in a left to right direction and the health policy process at the top of the framework funnels down to illustrate how specific health service innovations are very likely to be influenced by the social, economic and political aspects of the policy environment.<sup>54</sup> There are eight contextual factors which influence sustainability. Each factor contains key points to guide the assessment for sustainability of a health service innovation. Moving through the circle and ensuring all the major contextual factor points have been addressed, the framework directs its focus to the outcome of sustainability assessment. It prompts the assessor to consider the innovation in its new sustainable state and to ask whether the innovation is sustainable or not. Criteria to achieve an ideal state of sustainability include: outcome targets are well defined, there is alignment with funding and health policy, workforce and human resource planning is addressed, barriers to funding have been assessed, the health service delivery model has been well thought out, all decisions and planning are patient focused, strategic alignment with determinants of health has been considered and the emotional impact of the innovation has been explored.<sup>54</sup> If after assessment, the health services innovation is considered sustainable, future planning which includes focusing on long term evaluation can occur.<sup>54</sup> If it is determined by the assessor that sustainability is not achievable, the results of the assessment can be utilized to help further improve the sustainability of the health services innovation. See Framework Figure 5-1.

#### **Strengths and Limitations of this Framework**

The new conceptual framework is appropriate to assess sustainability across a broad spectrum of health service innovations and may be suitable to use beyond the confines of PHC.
The framework is intended to be applied very early in the innovation planning process but can be utilized at any point in the planning or implementation of an innovation. While the concepts of this framework were developed from the use of several bodies of knowledge, its key limitations should be highlighted. Given the limited evidence informing the Sustainability of Innovations Framework in PHC, testing the sustainability of a variety of PHC innovations is recommended and examining both sustainability success and failure may be an important element to consider in framework testing. Focusing on the connection between implementation, sustainability and scale up and spread may also offer further conceptual insight in this area of research. Additionally, conceptual frameworks, although based on highly regarded concepts and information, are often only a starting point to inquiry and may look considerably different following application in practice, evaluation and with revision.<sup>2</sup>. Operationalizing concepts in this proposed framework in several contexts will provide a broader understanding of its use and development. If future research in this field is to effectively inform health policy and implementation and scale up and spread must be used and tested.<sup>2</sup>

#### **Future Research**

The following section explains the scholarly and practical implications of this study that will be fruitful for future study. This thesis provides new information regarding the sustainability of innovations and more importantly the avoidance of sustainability failures such as the Sheep River Clinic, as well as the broad NP role in PHC. There are several implications for researchers: (1) using conceptual, theoretical and evidenced informed approaches that guide the design, development, implementation, evaluation and sustainability of any health service innovation is an important contribution to knowledge generation; (2) conducting research on the events that occur early in the innovation planning process, as well as incorporating necessary longitudinal research that examines the continuation of innovation activities or outcomes beyond the initial implementation is critical; 3) conducting research which traces the processes that occur in successful innovation sustainability along with parallel examinations of innovations that did not sustain. This includes comparing NP role implementation projects that were successfully sustained with those that were not.<sup>53,54</sup> This can be illuminated by future in-depth and comparative case studies of what happened and why. Ideally, these case studies would collect

data in an initial wave before the end of external funding and would then collect additional waves of data a year or more after the funding ended; <sup>53,54</sup> 4) conducting research which further examines the variability of Canadian jurisdictional initiatives and policies that influence sustainability of the NP role and other innovations in PHC; 5) advancing the concepts, measurement of the determinants, and outcomes of sustainability through robust prospective designs will also be critical; 6) studying failure in relation to the concepts of innovation, sustainability and optimization; how they are interconnected and how the use of the concepts together can provide solutions towards necessary PHC reform; and 7) coalescing and consolidating the knowledge base, language and terms to define and describe sustainability<sup>53</sup>

Improving the research-based knowledge about sustainability could initiate several desirable consequences. First, governments, policy-makers and funders would have more effective guidance on how to increase the likelihood of sustaining effective innovations. This would ensure a richer and deeper evidence base on how to sustain innovations in PHC.<sup>53</sup> Second, innovation management and coordination could be improved, as health care leaders became more aware of the strategies needed to achieve long-term beneficial outcomes. Third, improved linkage between short-term funding for developing new innovations and longer-term support for sustained programs to improve health outcomes. Finally, creating in-depth knowledge about sustainability and its influences can contribute to the broader research agendas for translation and dissemination of effective sustainability practices into widespread use.<sup>54</sup>

## **Knowledge Dissemination**

Without well planned dissemination, research information will not reach target audiences and in turn will not be effectively understood or utilized. The use of appropriate communication tools is necessary for mobilizing the knowledge generated from this research in order to reach target audiences. A central goal of a communication and dissemination plan is to maximise opportunities to promote, communicate and disseminate research results. This will ensure that organizations, governments, funders and policy- makers can contribute to, and act on the findings in a timely fashion.<sup>55</sup>

## **Objectives**

The dissemination and communication plan for this thesis has four main objectives, namely to: a) to educate, raise interest and awareness around utilizing the Primary Health Care Sustainability of Innovation Assessment Framework for use in assessing a wide variety of PHC and non-PHC health service innovations., b) encourage patients/community members to actively become involved in health policy issues regarding sustaining innovations in PHC, c) identify opportunities to engage among stakeholders and policy-makers to educate on sustaining innovations in PHC, including the introduction of the newly created framework, and 4) disseminate results in strategic and targeted ways.

This dissertation will be disseminated using a variety of communication channels and tools to communicate the results and outcomes of this case study and to actively engage relevant stakeholders as necessary. Research is most effectively disseminated using multiple methods and targeting multiple audiences. As such, there are five key intended audiences for this research. See Table 5-1:

- 1. Governments, Health Policy-makers, Funders and Organizations
- 2. Health Care Professionals
- 3. Communities/Patients/General Public
- 4. Academic Institutions
- 5. Media

## **Communication Avenues**

#### Social Media (Twitter/Facebook)

Use of social media (Facebook, Twitter, YouTube) contributes to establishing and maintaining public engagement about the research findings, any reports, media briefings and publications. Alerts of publications and published articles will be linked on researchers' own personal Facebook and Twitter.

## **Project brochures**

Researchers will provide study brochures summarizing key results. Two different brochures, one for academics/professionals/ leaders and one for non-academics will be published and distributed to target audiences as well as research participants.

#### Academic publications

The three studies in this thesis are intended for publication with Chapter 1: Scoping Review already accepted for publication in Policy, Politics and Nursing Practice. The other two papers will be submitted in appropriate peer reviewed journals.

#### Presentations at academic conferences and workshops

The studies produced in this thesis will target high-profile academic conferences and workshops organized by national and international organizations that involve and/or represent health policy, nursing or nurse practitioner associations.

#### Press Release/Media Reports

Media reports (articles, interviews, online reports, etc.) are based on press releases. One press release will be issued for the Case Study in the local Okotoks Community Paper. The other press release will be prepared for the Nurse Practitioner Association of Alberta. The press release will announce the major findings and policy recommendations.

## Policy briefs and reports

One policy brief with recommendations will be issued for the Case Study and distributed to Government Leaders, Policy Makers and Funders as well as major health organizations.

## **Expected Results of Dissemination Strategy**

The following are the expected results of this dissemination strategy: a) the researcher will create publicity of the research findings to attract potential future stakeholders and ensure maximum impact, b) the researcher will communicate to target audiences about policy and future research implications, c) the researcher will promote active participation in the research findings e.g. via the attendance in workshops. This dissemination strategy offers the breadth to reach out to multiple audiences and has the potential to create more in-depth interactive work with key audiences such as governments and policy-makers to influence system, organizational and health policy change.

#### Conclusion

This thesis explored the topic of sustainability failure of the NP role, a PHC innovation that has failed to be optimized throughout much of Canada and offers new insights into the process of sustainability of the NP role and other innovations in PHC. Achieving sustainability is an evolving process that requires continuing effort over time. This thesis is the foundation from which future research will be developed to improve the knowledge base of sustainability of the NP role and other innovations in PHC.

#### **Chapter 5: References**

- Considine J, Fielding K. Sustainable workforce reform: case study of Victorian nurse practitioner roles. *Aust Health Rev.* 2010;34(3):297-303. https://doi.org/10.1071/AH08727
- 2. Fox A, Gardner G, Osborne S. A theoretical framework to support research of health service innovation. *Aust Health Rev.* 2015;39(1):70-75. doi:10.1071/ah14031
- Hovlid E, Bukve O, Haug K, Aslaksen AB, Von plessen C. Sustainability of healthcare improvement: what can we learn from learning theory?. *BMC Health Serv Res.* 2012;12:235. <u>https://doi.org/10.1186/1472-6963-12-235</u>
- Keating SF, Thompson JP, Lee GA. Perceived barriers to the sustainability and progression of nurse practitioners. *International emergency nursing*, 2010 18(3), 147-153. <u>https://doi.org/10.1016/j.ienj.2009.09.003</u>
- Lennox L, Maher L, Reed J. Navigating the sustainability landscape: a systematic review of sustainability approaches in healthcare. *Implementation Science*. 2018;13(1). doi:10.1186/s13012-017-0707-4
- Martin GP, Weaver S, Currie G, Finn R, Mcdonald R. Innovation sustainability in challenging health-care contexts: embedding clinically led change in routine practice. *Health Serv Manage Res.* 2012;25(4):190-9. doi: 10.1177/0951484812474246
- Moore JE, Mascarenhas A, Bain J, Straus SE. Developing a comprehensive definition of sustainability. *Implement Sci.* 2017;12(1):110. doi: 10.1186/s13012-017-0637-1.

- Sibthorpe BM, Glasgow NJ, Wells RW. Emergent themes in the sustainability of primary health care innovation. *Medical Journal of Australia*. 2005;183(S10). doi:10.5694/j.1326-5377.2005.tb07185.x
- Stirman SW, Kimberly J, Cook N, Calloway A, Castro F, Charns M. The sustainability of new programs and innovations: a review of the empirical literature and recommendations for future research. *Implementation Science*. 2012;7(1). doi:10.1186/1748-5908-7-17
- Wakerman J, Humphreys JS. Sustainable primary health care services in rural and remote areas: Innovation and evidence. *Australian Journal of Rural Health*. 2011;19(3):118-124. doi:10.1111/j.1440-1584.2010.01180.x
- 11. Aggarwal M, Hutchison B. *Toward A Primary Care Strategy for Canada* (Rep.).: http://www.cfhi-fcass.ca/Libraries/Reports/Primary-Care-Strategy-EN.sflb.ashx
- 12. Canadian Federation of Nurses Unions. Fulfilling Nurse Practitioners' Untapped Potential in Canada's Health Care System. Media Release. <u>https://nursesunions.ca/wpcontent/uploads/2018/06/CFNU\_UntappedPotential-Final-EN.pdf</u>
- Hutchison B, Abelson J, Lavis J. Primary care in Canada: so much innovation, so little change. *Health Aff (Millwood)*. 2001;20(3):116-31. DOI:10.1377/hlthaff.20.3.116.
- Hutchison B. A long time coming: primary healthcare renewal in Canada. *Health Pap*. 2008;8(2):10-24. DOI: 10.12927/hcpap.2008.19704
- Hutchison B, Levesque JF, Strumpf E, Coyle N. Primary health care in Canada: systems in motion. *Milbank Q.* 2011;89(2):256-88. doi: <u>10.1111/j.1468-0009.2011.00628.x</u>
- 16. El-Jardali F,Lavis JN. Issue Brief: Addressing the Integration of Nurse Practitioners in Primary Healthcare Settings in Canada. *McMaster Health Forum*.

https://www.mcmasterforum.org/docs/default-source/product-documents/issuebriefs/nurse-practitioners-in-primary-healthcare-in-canada-ib.pdf?sfvrsn=2

17. Martin-Misener R, Donald F, Kilpatrick K, Bryant-Lukosius D, Rayner J, Landry V, Viscardi V, McKinlay R. Benchmarking for nurse practitioner patient panel size and comparative analysis of nurse practitioner pay scales: update of a scoping review. Hamilton, ON: McMaster University. 2015 Mar.

https://fhs.mcmaster.ca/ccapnr/documents/ScopingReviewNPPanelSize-

FINALPrinted17042015SJ\_000.pdf

- Johnson JE. Working together in the best interest of patients. J Am Board Fam Med.
   2013;26(3):241-3. doi: 10.3122/jabfm.2013.03.130075.
- Walt G, Gilson L. Reforming the health sector in developing countries: the central role of policy analysis. *Health Policy Plan*. 1994;9(4):353-70. doi:10.1093/heapol/9.4.353
- 20. Woltering L. 'Pilots Never Fail, Pilots Never Scale': Why the Global Development Community Needs a More Realistic Approach to Reaching Billions.
   <u>https://nextbillion.net/global-development-needs-realistic-approach-to-scale/. Published</u>
   <u>2019</u>
- Uhlig, P. 2006 (April 3). "Collaborative Practice." Presentation, NPAO Accord Project Conference: Ahead of the Curve.
- 22. Bauer JC. Nurse practitioners as an underutilized resource for health reform: evidencebased demonstrations of cost-effectiveness. *J Am Acad Nurse Pract*. 2010;22(4):228-31. doi:10.1111/j.1745-7599.2010.00498.x

- 23. Brassard A. Making the case for NPs as primary care providers. *Am Nurse*. 2013: 45(3), 13-13.
- 24. Dicenso A, Auffrey L, Bryant-Lukosius D, et al. Primary health care nurse practitioners in Canada. *Contemporary Nurse*. 2007;26(1):104-115. doi:10.5172/conu.2007.26.1.104
- 25. Fund ME, Swanson-Hill A. Cost-effectiveness of nurse practitioner care. *Kansas Nurse*.
  2014: 89(1), 12-15. <u>http://amandasivewright.weebly.com/</u>
  uploads/3/0/8/4/30840229/reference 2.pdf. Published 2014. Accessed
- 26. Gould ON, Johnstone D, Wasylkiw L. Nurse practitioners in Canada: Beginnings, benefits, and barriers. *Journal of the American Academy of Nurse Practitioners*. 2007;19(4):165-171. doi:10.1111/j.1745-7599.2007.00210.x
- 27. Poghosyan L, Lucero R, Rauch L, Berkowitz B. Nurse practitioner workforce: a substantial supply of primary care providers. *Nurs Econ.* 2012;30(5):268-74, 294.
- 28. Carter N, Martin-Misener R, Kilpatrick K, et al. The role of nursing leadership in integrating clinical nurse specialists and nurse practitioners in healthcare delivery in Canada. *Nurs Leadersh (Tor Ont*). 2010;23 Spec No 2010:167-85. https://doi.org/10.12927/cjnl.2010.22274
- 29. DiCenso A, & Bryant-Lukosius D. Clinical nurse specialists and nurse practitioners in Canada. <u>http://www.cfhi-fcass.ca/SearchResultsNews/10-06-01/b9cb9576-6140-4954-aa57-2b81c1350936.aspx</u>. Published 2010. Accessed
- Jayaratna N, Wood. B. Every Innovation is a New Thing but not Every New Thing is an Innovation. *Projectics / Proyéctica / Projectique*, 2008: 0(1), 13-25. doi:10.3917/proj.000.0013.

- 31. Esmail N, Barua B. Is Canada's Health Act a Barrier to Reform. *Fraser Institute*. <u>https://www.fraserinstitute.org/studies/is-the-canada-health-act-a-barrier-to-reform</u>. Published 1 May 2018. Accessed
- 32. Edmondson AC. Learning from failure in health care: frequent opportunities, pervasive barriers. *BMJ Quality & Safety*, 2014: 13(suppl 2), ii3-ii9.
- 33. Nolte E. (2018). How do we ensure that innovation in health service delivery and organization is implemented, sustained and spread. Policy Brief. Copenhagen: WHO Regional Office for Europe.
  <u>http://www.euro.who.int/\_\_data/assets/pdf\_file/0004/380731/pb-tallinn-03-eng.pdf?ua=1</u>
  Published 2018. Accessed
- 34. Simon HA. A behavioral model of rational choice. *The quarterly journal of economics*, 1955: 69(1), 99-118. DOI: 10.2307/1884852
- 35. Clemens E, Cook J. Politics and Institutionalism: Explaining Durability and Change. *Annual Review of Sociology*, 1995: 25, 441-466. <u>www.jstor.org/stable/223512</u>
- 36. Herzlinger, R. E. (2006). Why innovation in health care is so hard. Harvard business review, 84(5), 58.
- 37. Howlett M. The lessons of failure: learning and blame avoidance in public policymaking. *International Political Science Review*. 2012;33(5):539-555. doi:10.1177/0192512112453603
- 38. MacNeil M, Koch M, Kuspinar A, Juzwishin D, Lehoux P, Stolee P. Enabling health technology innovation in Canada: Barriers and facilitators in policy and regulatory processes. *Health Policy*. 2019;123(2):203-214. doi: 10.1016/j.healthpol.2018.09.018.

- Braithwaite J. Changing how we think about healthcare improvement. BMJ.
   2018;361:k2014. doi: 10.1136/bmj.k2014.
- 40. Daddow O. *How to Reduce the Risk of Policy Failure: Three actions you can take.* 2010 https://apolitical.co/en/solution\_article/how-to-reduce-risk-policy-failure
- 41. Hudson B, Hunter D, Peckham S. Policy failure and the policy-implementation gap: can policy support programs help?, *Policy Design and Practice* 2019:2:1, 1-14, DOI: 10.1080/25741292.2018.1540378
- 42. Benoit G, McCrae J, Shepheard M. Accountability in Modern Government: Recommendations for Change. <u>https://www.instituteforgovernment.org.uk</u> /sites/default/files/publications/Accountability\_modern\_government\_WEB.pdf.
  Published April 2018. Accessed January 18, 2020
- 43. Gavelin K, Burall S, Wilson R. Open Government: beyond static measures. A paper produced by Involve for the OECD. <u>http://www.oecd.org/gov/46560184.pdf</u>. Published July 2009. Accessed Accessed January 18, 2020
- 44. Paschke A, Dimancesco D, Vian T, Kohler JC, Forte G. Increasing transparency and accountability in national pharmaceutical systems. *Bulletin of the World Health Organization*. 2018: 96(11), 782.
- 45. Walugembe DR, Sibbald S, Ber MJL, Kothari A. Sustainability of public health interventions: where are the gaps? *Health Research Policy and Systems*. 2019;17(1). doi:10.1186/s12961-018-0405-y

- 46. Souliotis K, Agapidaki E, Peppou LE, Tzavara C, Samoutis G, Theodorou M. Assessing Patient Participation in Health Policy Decision-Making in Cyprus. *Int J Health Policy* Manag. 2016;5(8):461-466. <u>https://doi.org/10.15171/ijhpm.2016.78</u>
- 47. Schell SF, Luke DA, Schooley MW, et al. Public health program capacity for sustainability: a new framework. *Implement Sci. 20*13;8:15. <u>https://doi.org/10.1186/1748-5908-8-15</u>
- Pluye P, Potvin L, Denis J-L. Making public health programs last: conceptualizing sustainability. *Eval Prog Plann*. 2004;27(2):121–33.
- 49. Imenda S. Is there a conceptual difference between theoretical and conceptual frameworks? *Journal of Social Sciences*, 2013:38(2), 185-195.
  https://doi.org/10.1080/09718923.2014.11893249
- Walt G, Gilson L. Reforming the health sector in developing countries: the central role of policy analysis. *Health Policy Plan*. 1994;9(4):353-70. doi:10.1093/heapol/9.4.353
- 51. Dixon-Woods M, Amalberti R, Goodman S, Bergman B, Glasziou P. Problems and promises of innovation: why healthcare needs to rethink its love/hate relationship with the new. BMJ quality & safety. 2011 Apr 1;20(Suppl 1):i47-51.
- 52. Lerner JS, Li Y, Valdesolo P, Kassam KS. Emotion and decision making. Annual review of psychology. 2015 Jan 3;66:799-823.
- 53. Shelton RC, Cooper BR, Stirman SW. The Sustainability of Evidence-Based Interventions and Practices in Public Health and Health Care. Annual Review of Public Health. 2018;39(1):55-76. doi:10.1146/annurev-publhealth-040617-014731

- 54. Scheirer MA, Dearing JW. An Agenda for Research on the Sustainability of Public Health Programs. *American Journal of Public Health*. 2011;101(11):2059-2067. doi:10.2105/ajph.2011.300193
- 55. Straus SE, Tetroe J, Graham I. Defining knowledge translation. *Canadian Medical Association Journal*. 2009;181(3-4):165-168. doi:10.1503/cmaj.081229

Dissemination and Use of Results for Different Target Groups							
Aims:	Governments, Funders,	Experts,	Health Care	Media	Patients		
	Health Policy Makers,	Academics,	Professional		<b>Community Members</b>		
	Organizations at Different	Researchers					
	Levels						
	<ul> <li>Education/Involve in discussion</li> <li>Disseminate results to above groups</li> <li>Use and build on results for future policy making and project funding</li> <li>Sustainability Assessment Framework</li> </ul>	<ul> <li>Education &amp; Involve in discussion</li> <li>Disseminate results to</li> <li>Use and build on results for future research</li> <li>Sustainability Assessment Framework</li> </ul>	<ul> <li>Education &amp; Involve in discussion</li> <li>Disseminate results to</li> <li>Use and build on results to involve in future sustainability research</li> <li>Sustainability</li> </ul>	<ul> <li>Education &amp; Involve in discussion</li> <li>Disseminate results to</li> <li>Use and build on results to educate general public</li> <li>Sustainability Assessment</li> </ul>	<ul> <li>Education &amp; Involve in discussion</li> <li>Disseminate results to</li> <li>Use and build on results to involve in future research</li> <li>Sustainability Assessment Framework</li> </ul>		
			Assessment Framework	Framework			

Table 5-1: Dissemination and Use of Results for Different Target Groups

				221
Measu	Stakeholder workshops	• Stakeholder	Stakeholder     Regular press	Stakeholder workshops
res to	• Policy briefs Policy papers	workshops	workshops releases	• Materials developed for
	Social Media	• Academic/expert	Materials     Events open to	patient/community
Comm	• YouTube	conference	• Publications the press	• Publications aimed at
unicate		presentations	aimed at expert • Interviews	layperson: project
		• Academic/expert	audience: project • Social Media	summary brochures,
		publications	summary • YouTube	information in the
		• Social	brochures,	media
		Media/YouTube	information in the	Social Media/YouTube
			media	
			Social Media	
			• YouTube	

Figure 5-1: Primary Health Care Sustainability of Innovation Assessment Framework



# Sustainability of Innovations in Primary Healthcare Integrative Framework

## **Bibliography (all cited works)**

- Stonebridge C, Brichta J, Dinh T. A Road Map to Health System Sustainability: CASHC Compendium Report, 2011–16. The Conference Board of Canada - Economic forecasts, public policy, and organizational performance. <u>https://www.conferenceboard.ca/elibrary/abstract.aspx?did=8816&AspxAutoDetectCookieSupport=1</u>. Published 15 May 2017. Accessed February 13, 2020
- Martin GP, Weaver S, Currie G, Finn R, Mcdonald R. Innovation sustainability in challenging health-care contexts: embedding clinically led change in routine practice. *Health Serv Manage Res.* 2012;25(4):190-9. doi: 10.1177/0951484812474246
- Sangster-Gormley E, Martin-Misener R, Burge F. A case study of nurse practitioner role implementation in primary care: what happens when new roles are introduced? *BMC Nursing*. 2013;12(1). doi:10.1186/1472-6955-12-1.
- Brown LJ, Mcintyre EL. The contribution of Primary Health Care Research, Evaluation and Development-supported research to primary health care policy and practice. *Aust J Prim Health*. 2014;20(1):47-55. doi:10.1071/py12058
- Canadian Nurses Association. Primary Health Care The Time Has Come. https://www.cna-aiic.ca/~/media/cna/page-content/pdffr/primaryhealthcare\_sept\_2003\_e.pdf?la=en. Published September 2003
- 6. World Health Organization. International Conference on Primary Health Care (1978 : Alma-Ata, USSR), World Health Organization & United Nations Children's Fund (UNICEF). (1978). Primary health care: report of the International Conference on Primary Health Care, Alma-Ata, USSR, 6-12 September 1978 / jointly sponsored by the World Health Organization and the United Nations Children's Fund. World Health Organization. <u>https://apps.who.int/iris/handle/10665/39228</u>
- Canadian Nurses Association. *Primary care, Primary Health Care*. Canadian Nurse. https://canadian-nurse.com/articles/issues/2015/june-2015/cna-now?page=8

- Scheirer MA, Dearing JW. An Agenda for Research on the Sustainability of Public Health Programs. *American Journal of Public Health*. 2011;101(11):2059-2067. doi:10.2105/ajph.2011.300193
- Shelton RC, Cooper BR, Stirman SW. The Sustainability of Evidence-Based Interventions and Practices in Public Health and Health Care. *Annual Review of Public Health*. 2018;39(1):55-76. doi:10.1146/annurev-publhealth-040617-014731
- Walugembe DR, Sibbald S, Ber MJL, Kothari A. Sustainability of public health interventions: where are the gaps? *Health Research Policy and Systems*. 2019;17(1). doi:10.1186/s12961-018-0405-y
- Fleiszer AR, Semenic SE, Ritchie JA, Richer M-C, Denis J-L. The sustainability of healthcare innovations: a concept analysis. *Journal of Advanced Nursing*. 2015;71(7):1484-1498. doi:10.1111/jan.12633
- Buykx P, Humphreys JS, Tham R, et al. How do small rural primary health care services sustain themselves in a constantly changing health system environment? *BMC Health Services Research*. 2012;12(1). doi:10.1186/1472-6963-12-81
- Lennox L, Maher L, Reed J. Navigating the sustainability landscape: a systematic review of sustainability approaches in healthcare. *Implementation Science*. 2018;13(1). doi:10.1186/s13012-017-0707-4
- Sibthorpe BM, Glasgow NJ, Wells RW. Emergent themes in the sustainability of primary health care innovation. *Medical Journal of Australia*. 2005;183(S10). doi:10.5694/j.1326-5377.2005.tb07185.x
- 15. Wakerman J. Innovative rural and remote primary health care models: What do we know and what are the research priorities? *Australian Journal of Rural Health*. 2009;17(1):21-26. doi:10.1111/j.1440-1584.2008.01032.x
- Wakerman J, Humphreys JS. Sustainable primary health care services in rural and remote areas: Innovation and evidence. *Australian Journal of Rural Health*. 2011;19(3):118-124. doi:10.1111/j.1440-1584.2010.01180.x

- Stirman SW, Kimberly J, Cook N, Calloway A, Castro F, Charns M. The sustainability of new programs and innovations: a review of the empirical literature and recommendations for future research. *Implementation Science*. 2012;7(1). doi:10.1186/1748-5908-7-17
- Hutchison B, Levesque JF, Strumpf E, Coyle N. Primary health care in Canada: systems in motion. *Milbank Q.* 2011;89(2):256-88. doi: <u>10.1111/j.1468-0009.2011.00628.x</u>
- Canadian Nurses Association. (2019). Advanced practice nursing: a pan-Canadian framework. Ottawa: Canadian Nurses Association. <u>https://www.cna-aiic.ca/-</u> /media/cna/page-content/pdf-en/apn-a-pan-canadian-framework.pdf
- Donald F, Martin-Misener R, Bryant-Lukosius D, et al. The Primary Healthcare Nurse Practitioner Role in Canada. *Nursing Leadership*. 2010;23(sp):88-113. doi:10.12927/cjnl.2013.22271
- 21. El-Jardali F, Lavis JN. Issue Brief: Addressing the Integration of Nurse Practitioners in Primary Healthcare Settings in Canada. Hamilton, Canada: McMaster Health Forum, 6 July 2011. https://www.mcmasterforum.org/docs/default-source/productdocuments/issue-briefs/nurse-practitioners-in-primary-healthcare-in-canadaib.pdf?sfvrsn=2
- Dicenso A, Bourgeault I, Abelson J, et al. Utilization of Nurse Practitioners to Increase Patient Access to Primary Healthcare in Canada – Thinking Outside the Box. *Nursing Leadership.* 2010;23(sp):239-259. doi:10.12927/cjnl.2010.22281
- Kaasalainen S, Martin-misener R, Kilpatrick K, et al. A historical overview of the development of advanced practice nursing roles in Canada. *Nurs Leadersh* (Tor Ont). 2010;23 Spec No 2010:35-60. DOI: 10.12927/cjnl.2010.22268
- 24. Dicenso A, Auffrey L, Bryant-Lukosius D, et al. Primary health care nurse practitioners in Canada. *Contemporary Nurse*. 2007;26(1):104-115. doi:10.5172/conu.2007.26.1.104
- Edwards N, Rowan M, Marck P, Grinspun D. Understanding Whole Systems Change in Health Care: The Case of Nurse Practitioners in Canada. *Policy, Politics, & Nursing Practice.* 2011;12(1):4-17. doi:10.1177/1527154411403816

- Mattison CA, Wilson MG. Rapid Synthesis: Enhancing Health System Integration of Nurse Practitioners in Ontario.
- Ashby M. (2006). Barriers to nurse practitioner reimbursement. *The Pennsylvania Nurse*, 61(2), 19.
- Brassard A. Scope of practice wins in Iowa, Nevada, Oregon and Rhode Island. *The American Nurse*, 2013:11.
- Gould ON, Johnstone D, Wasylkiw L. Nurse practitioners in Canada: Beginnings, benefits, and barriers. *Journal of the American Academy of Nurse Practitioners*. 2007;19(4):165-171. doi:10.1111/j.1745-7599.2007.00210.x
- 30. Canadian Nurses Association. The Primary Health Care Approach. <u>https://www.cna-aiic.ca/-/media/cna/page-content/pdf-en/fs02\_primary\_health\_care\_approach\_june\_2000\_e.pdf?la=en&hash=C31F23CD2036</u> 6955F267C04D737B32163F7D5623
- 31. Esmail N, Barua B. Is Canada's Health Act a Barrier to Reform. Fraser Institute. https://www.fraserinstitute.org/studies/is-the-canada-health-act-a-barrier-to-reform
- Martin-Misener R, Bryant-lukosius D, Harbman P, et al. Education of advanced practice nurses in Canada. *Nurs Leadersh* (Tor Ont). 2010;23 Spec No 2010:61-84. DOI:10.12927/cjnl.2010.22269
- 33. Tapper L, Trevoy J. Key Issues Impacting Nurse Practitioner Practice and Integration in Alberta. Nurse Practitioner Association of Alberta. <u>https://albertanps.com/wpcontent/uploads/2015/03/Key-Issues-for-Nurse-Practitioners-of-Alberta.pdf</u>
- Howlett M. The lessons of failure: learning and blame avoidance in public policymaking. *International Political Science Review*. 2012;33(5):539-555. doi:10.1177/0192512112453603
- 35. Fox A, Gardner G, Osborne S. A theoretical framework to support research of health service innovation. *Aust Health Rev.* 2015;39(1):70-75. doi:10.1071/ah14031

- 36. Walt G, Gilson L. Reforming the health sector in developing countries: the central role of policy analysis. *Health Policy Plan.* 1994;9(4):353-70. doi:10.1093/heapol/9.4.353
- 37. Greenhalgh T, Robert G, Bate P, Kyriakidou O, Macfarlane F, Peacock R. How to spread good ideas A systematic review of the literature on diffusion, dissemination and sustainability of innovations in health service delivery and organisation. *Report for the National Co-ordinating centre for NHS Service Delivery and Organisation R & D (NCCSDO)*, 1-426
- Chambers DA, Glasgow RE, Stange KC. The dynamic sustainability framework: addressing the paradox of sustainment amid ongoing change. *Implementation Science*. 2013;8(1). doi:10.1186/1748-5908-8-117
- 39. Yin RK. Case study research: Design and methods (5th ed.). Thousand Oaks, CA: Sage.
- 40. Bourke B. Positionality: Reflecting on the research process. *Qualitative Report*, 2015: 19(33).
- 41. Hovlid E, Bukve O, Haug K, Aslaksen AB, Von plessen C. Sustainability of healthcare improvement: what can we learn from learning theory?. *BMC Health Serv Res.* 2012;12:235. <u>https://doi.org/10.1186/1472-6963-12-235</u>
- 42. Dicenso A, Auffrey L, Bryant-Lukosius D, et al. Primary health care nurse practitioners in Canada. *Contemporary Nurse*. 2007;26(1):104-115. doi:10.5172/conu.2007.26.1.104
- 43. https://doi.org/10.5172/conu.2007.26.1.104
- 44. Donald F, Martin-Misener R, Bryant-Lukosius D, et al. The Primary Healthcare Nurse Practitioner Role in Canada. *Nursing Leadership*. 2010;23(sp):88-113. doi:10.12927/cjnl.2013.22271
- 45. Sangster-Gormley E, Martin-Misener R, Burge F. A case study of nurse practitioner role implementation in primary care: what happens when new roles are introduced? *BMC Nursing*. 2013;12(1). doi:10.1186/1472-6955-12-1

- 46. Muldoon, L. K., Hogg, W. E., & Levitt, M. (2006). Primary care (PC) and primary health care (PHC): what is the difference? *Canadian Journal of Public Health/Revue Canadienne de Sante'e Publique*, 409–411. https://www.jstor.org/stable/41995754
- 47. Starfield, B. (1998). *Primary care: balancing health needs, services, and technology*.Oxford University Press.
- Canadian Nurses Association. (2015). Primary care, Primary Health Care. Canadian Nurse. https://www.cna-aiic.ca/~/media/cna/page-content/pdf-en/primary-health-careposition-statement.pdf
- World Health Organization. (1978, September 6–12). Primary health care: report of the International Conference on Primary Health Care, Alma-Ata, USSR, 6–12.
   <u>https://apps.who.int/iris/handle/10665/39228</u>
- 50. Moore JE, Mascarenhas A, Bain J, Straus SE. Developing a comprehensive definition of sustainability. *Implementation Science*. 2017;12(1). doi:10.1186/s13012-017-0637-1
- 51. Canadian Nurses Association. Advanced practice nursing: a pan-Canadian framework. Ottawa: Canadian Nurses Association. https://www.cna-aiic.ca/-/media/cna/page-content/pdf-en/apn-a-pan-canadian-framework.pdf
- Dicenso A, Bourgeault I, Abelson J, et al. Utilization of Nurse Practitioners to Increase Patient Access to Primary Healthcare in Canada – Thinking Outside the Box. *Nursing Leadership.* 2010;23(sp):239-259. doi:10.12927/cjnl.2010.22281
- 53. Canadian Federation of Nurses Unions. Media Release. https://nursesunions.ca/wpcontent/uploads/2017/11/2017-11-13 NursePractitionerWeek EN-1.pdf
- Gould ON, Johnstone D, Wasylkiw L. Nurse practitioners in Canada: Beginnings, benefits, and barriers. *Journal of the American Academy of Nurse Practitioners*. 2007;19(4):165-171. doi:10.1111/j.1745-7599.2007.00210.x
- 55. Sangster-Gormley E. Interprofessional Collaboration: Co-workers' Perceptions of Adding Nurse Practitioners to Primary Care Teams. *Quality in Primary Care*, 2015: *23*(3).

http://primarycare.imedpub.com/interprofessional-collaboration-coworkersperceptionsof-adding-nurse-practitioners-toprimary-care-teams.php?aid=5485

- 56. Arksey H, O'Malley L. Scoping studies: Towards a methodological framework. *International Journal of Social Research Methodology*, 2005: 8(1), 19–32. <u>https://doi.org/10.1080/1364557032000119616</u>
- Arenson A. Questionable logic behind clinic closure. *Winnipeg Free Press*. Published 04 August 2017. Accessed January 30, 2018
- Bruineman M. Clinic suffers from lack of funding: *Examiner*. Published 06 September 2001. Accessed January 30, 2018
- Bruineman M. Clinic closure creates anxieties for seniors: Sandy cove facility closing Nov 19. *Examiner*. Published 18 September 2001. Accessed January 30, 2018
- 60. Conrad K. Lack of funding closes clinic. Okotoks Western Wheel. http://www.westernwheel.com/article/Lack-of-funding-closes-clinic-20151211.
   Published 11 December 2015. Accessed January 30, 2018
- Gilchrist L. Two medical clinics facing closure. *Peterborough this Week*. Published 16
   September 2005. Accessed January 30, 2018
- 62. Gilchrist L. Clinics' pending closure to see 2,000 lose primary health care. *Peterborough this Week.* Published 23 December 2005. Accessed January 30, 2018
- Guest A, Australia's first stand-alone nurse practitioner clinic closes [Radio broadcast]. *The World Today, ABC News*, 17.
   <u>https://www.abc.net.au/radio/programs/worldtoday/australias-first-stand-alone-nurse-practitioner/4017410</u>. Published 17 May 2012. Accessed January 30, 2018
- King K. Family health teams prepare for transition. *Peterborough Examiner*. Published
   16 December 2005. Accessed January 30, 2018
- Kovach J. Ottawa not to blame if medical centre closes: MP. *Peterborough Examiner*. Published 22 December 2001. Accessed January 30, 2018

- 66. Looming Walk In. Looming walk-in closure disappoints. Vernon Morning Star. https://www.vernonmorningstar.com/news/looming-walk-in-closure-disappoints/.
   Published 5 December 2015. Accessed January 30, 2018
- Marchen, J. Minister vows to look into clinic's dilemma. *Peterborough Examiner* Published 18 December 2010. Accessed January 30, 2018
- McEachern L. Project redefines nursing role: Nurse practitioner pilot project wrapping up final year. *Evening News*. Published 21 October 2002. Accessed January 30, 2018
- Punch R. Funding runs out for clinics. *Peterborough Examiner*. Published 15 September 2005. Accessed January 30, 2018
- Punch R. Nowhere to go for patients: *Peterborough Examiner*. Published 16 September 2005. Accessed January 30, 2018
- Wedley B. Clinic's closure causes concern. *Peterborough Examiner*. Published 12 January 2006. Accessed January 30, 2018
- Herhalt C. Care clinic no longer needed. *Waterloo Region Record*. Published 21 April 2015. Accessed January 30, 2018
- Ramsay J. Elmvale nurse practitioner clinic closes. *Barrie-Advance*.
   <u>https://www.simcoe.com/news-story/4247236-elmvale-nurse-practitioner-clinic-closes/</u>
   Published 29 November 2013. Accessed January 30, 2018
- 74. Hutton R. Doors to close on Niagara Falls nurse practitioner clinic. *Niagara This Week*. <u>https://www.niagarathisweek.com/news-story/4040234-doors-to-close-on-niagara-falls-nurse-practitioner-clinic/.</u> Published 20, August 2013. Accessed January 30, 2018
- 75. MEI. (2016, April 12) Super nurse clinics: When bureaucracy prevents better access to care. <u>https://www.iedm.org/59836-super-nurse-clinics-when-bureaucracy-prevents-better-access-to-care/</u>
- 76. Carter, M. (2015, August 27). Dawson creek clinic facing closure. Prince George Citizen

- 77. Buykx P, Humphreys JS, Tham R, et al. How do small rural primary health care services sustain themselves in a constantly changing health system environment? *BMC Health Services Research*. 2012;12(1). doi:10.1186/1472-6963-12-81
- Gruen RL, Elliott JH, Nolan ML, Lawton PD, Parkhill A, McLaren CJ, Lavis JN.
   Sustainability science: an integrated approach for health-programme planning. *The Lancet*. 2008:372(9649), 1579–1589. https://doi.org/10.1016/S0140-6736(08)61659-1
- 79. Lennox L, Maher L, Reed J. Navigating the sustainability landscape: a systematic review of sustainability approaches in healthcare. *Implementation Science*. 2018;13(1). doi:10.1186/s13012-017-0707-4
- Wakerman J, Humphreys JS. Sustainable primary health care services in rural and remote areas: Innovation and evidence. *Australian Journal of Rural Health*. 2011;19(3):118-124. doi:10.1111/j.1440-1584.2010.01180.x
- Stirman SW, Kimberly J, Cook N, Calloway A, Castro F, Charns M. The sustainability of new programs and innovations: a review of the empirical literature and recommendations for future research. *Implementation Science*. 2012;7(1). doi:10.1186/1748-5908-7-17
- Sibthorpe BM, Glasgow NJ, Wells RW. Emergent themes in the sustainability of primary health care innovation. *Medical Journal of Australia*. 2005;183(S10). doi:10.5694/j.1326-5377.2005.tb07185.x
- Martin GP, Weaver S, Currie G, Finn R, Mcdonald R. Innovation sustainability in challenging health-care contexts: Embedding clinically led change in routine practice. *Health Serv Manage Res. 2012;25(4):190-9*.doi:10.1177/0951484812474246.
- 84. Considine J, Fielding K. Sustainable workforce reform: case study of Victorian nurse practitioner roles. Aust Health Rev. 2010;34(3):297-303. https://doi.org/10.1071/AH08727
- 85. Fox A, Gardner G, Osborne S. A theoretical framework to support research of health service innovation. Aust Health Rev. 2015;39(1):70-75. doi:10.1071/ah14031

- Keating SF, Thompson JP, Lee GA. Perceived barriers to the sustainability and progression of nurse practitioners. *International emergency nursing*, 2010 18(3), 147-153. <u>https://doi.org/10.1016/j.ienj.2009.09.003</u>
- 87. Fleiszer AR, Semenic SE, Ritchie JA, Richer M-C, Denis J-L. The sustainability of healthcare innovations: a concept analysis. *Journal of Advanced Nursing*. 2015;71(7):1484-1498. doi:10.1111/jan.12633
- Greenhalgh T, Robert G, Macfarlane F, Bate P, Kyriakidou O. Diffusion of innovations in service organizations: systematic review and recommendations. *Milbank Q*. 2004;82(4):581-629. https://doi.org/10.1111/j.0887-378X.2004.00325.x
- Scheirer, M. A., Hartling, G., & Hagerman, D. (2008). Defining sustainability outcomes of health programs: Illustrations from an on-line survey. *Evaluation and Program Planning*, 31(4), 335–346. <u>https://doi.org/10.1016/j.evalprogplan.2008.08.004</u>
- 90. Proctor E, Luke D, Calhoun A, et al. Sustainability of evidence-based healthcare: research agenda, methodological advances, and infrastructure support. *Implement Sci.* 2015;10:88. <u>https://doi.org/10.1186/s13012-015-0274-5</u>
- 91. Canadian Nurses Association (2020). Nursing Statistics. <u>https://www.cna-aiic.ca/en/nursing-practice/the-practice-of-nursing/health-human-resources/nursing-statistics</u>
- 92. Canadian Federation of Nurses Union (2018). *Fulfilling nurse practitioners untapped potential in Canada's healthcare system*. <u>https://nursesunions.ca/wp-content/uploads/2018/06/CFNU\_UntappedPotential-Final-EN.pdf</u>
- 93. Carter N, Martin-Misener R, Kilpatrick K, et al. The role of nursing leadership in integrating clinical nurse specialists and nurse practitioners in healthcare delivery in *Canada. Nurs Leadersh* (Tor Ont). 2010;23 Spec No 2010:167-85. https://doi.org/10.12927/cjnl.2010.22274
- 94. Fox A, Gardner G, Osborne S. A theoretical framework to support research of health service innovation. *Aust Health Rev.* 2015;39(1):70-75. doi:10.1071/ah14031

- 95. Canadian Nurses Association. (2003). Health Human Resource Component, Literature Review Report, Recruitment and Retention of Primary Health Care Nurse Practitioners in Canada. https://cna-aiic.ca/~/media/cna/page-content/pdfen/04 hhr appendixc.pdf?la=en
- 96. Canadian Nurses Association (2016, August). The Canadian Nurse Practitioner Initiative: A10-year Retrospective. <u>https://cna-aiic.ca/~/media/cna/page-content/pdf-</u> en/canadian-nurse-practitioner-initiative-a-10-year-retrospective.pdf?la=en
- 97. Dicenso A, Auffrey L, Bryant-Lukosius D, et al. Primary health care nurse practitioners in Canada. *Contemporary Nurse*. 2007;26(1):104-115. doi:10.5172/conu.2007.26.1.104
- 98. Dicenso A, Bourgeault I, Abelson J, et al. Utilization of Nurse Practitioners to Increase Patient Access to Primary Healthcare in Canada – Thinking Outside the Box. *Nursing Leadership*. 2010;23(sp):239-259. doi:10.12927/cjnl.2010.22281
- 99. Donald F, Martin-Misener R, Bryant-Lukosius D, et al. The Primary Healthcare Nurse Practitioner Role in Canada. *Nursing Leadership*. 2010;23(sp):88-113. doi:10.12927/cjnl.2013.22271
- 100. Edwards N, Rowan M, Marck P, Grinspun D. Understanding Whole Systems Change in Health Care: The Case of Nurse Practitioners in Canada. *Policy, Politics, & Nursing Practice.* 2011;12(1):4-17. doi:10.1177/1527154411403816
- 101. Gould ON, Johnstone D, Wasylkiw L. Nurse practitioners in Canada: Beginnings, benefits, and barriers. *Journal of the American Academy of Nurse Practitioners*. 2007;19(4):165-171. doi:10.1111/j.1745-7599.2007.00210.x
- 102. De Witt L, Ploeg J. Critical analysis of the evolution of a Canadian nurse practitioner role. *Canadian Journal of Nursing Research Archive*. 2005;1:37(4).
- 103. Wong ST, Farrally V. The utilization of nurse practitioners and physician assistants: A research synthesis. Michael Smith Foundation for Health Research; 2014.

- 104. Green LV, Savin S, Lu Y. Primary care physician shortages could be eliminated through use of teams, nonphysicians, and electronic communication. *Health Affairs*. 2013; 1;32(1):11-9.
- 105. Geurts-Laurant MG, Reeves D, Hermens RP, Braspenning JC, Grol RP, Sibbald BS. Substitution of doctors by nurses in primary care. *Cochrane Database of Systematic Reviews*. 2004;4:CD001271.
- 106. Mundinger MO, Kane RL, Lenz ER, Totten AM, Tsai WY, Cleary PD, Friedewald WT, Siu AL, Shelanski ML. Primary care outcomes in patients treated by nurse practitioners or physicians: a randomized trial. *Jama*. 2000 Jan 5;283(1):59-68.
- Peterson LE, Phillips RL, Puffer JC, Bazemore A, Petterson S. Most family physicians work routinely with nurse practitioners, physician assistants, or certified nurse midwives. *J Am Board Fam Med* 2013; 26:244 –245.
- 108. Reay T, Patterson EM, Halma L, Steed W. Introducing a nurse practitioner: experiences in a rural Alberta family practice clinic. *Can J Rural Med.* 2006;11(2):101-7.
- 109. Nurse Practitioner Association of Canada. (2019, May 24 email). President's Newsletter.
- Canadian Institute for Health Information. *Health Workforce in Canada, 2019 Quick Stats*. Ottawa, ON: CIHI; 2020.
- DiCenso A, Bryant-Lukosius D. Clinical nurse specialists and nurse practitioners in Canada. *Canadian Health Services Research Foundation*. 2010 Jun.
- 112. Foster W, Kim P, Christenson D. Ten Non Profit Funding Models. Stanford Social Innovation Review. 2009: https://ssir.org/articles/entry/ten\_nonprofit\_funding\_models#
- 113. Canadian Nurses Association (2009b). Recommendations of the Canadian Nurse Practitioner Initiative: A Progress Report.
- 114. Delvin ME, Braithwaite S, Plazas PC. Canadian nurse practitioner's quest for identity: A philosophical perspective. *International journal of nursing sciences*. 2018: 10;5(2):110-4.
- 115. Dueck, S. Timely Care Care Clinic Now Serving Steinbach Patients. 2017 https://steinbachonline.com/local/timely-care-clinic-now-serving-steinbach-patients

- 116. Murphy GT, Birch S, Mackenzie A, Rigby J, Langley J. An integrated needs-based approach to health service and health workforce planning: applications for pandemic influenza. *Healthcare Policy*. 2017:13(1):28.
- 117. Divisions of Family Practice. (n.d.). Contracting 200 NPs into Primary Care Networks. https://www.divisionsbc.ca/sites/default/files/inlinefiles/NP%20Backgrounder%20and%20FAQ%20-%20Final%20181023.pdf
- 118. Sangster-Gormley E, Martin-Misener R, Burge F. A case study of nurse practitioner role implementation in primary care: what happens when new roles are introduced? *BMC Nursing*. 2013;12(1). doi:10.1186/1472-6955-12-1
- 119. Government of Alberta. PCN Nurse Practitioner Support Program. 2019: https://www.alberta.ca/assets/documents/health-pcn-nurse-practitioner-support-programguide.pdf
- Martin-Misener R, Donald F, Kilpatrick K, Bryant-Lukosius D, Rayner J, Landry V, Viscardi V, McKinlay R. Benchmarking for nurse practitioner patient panel size and comparative analysis of nurse practitioner pay scales: update of a scoping review. Hamilton, ON: McMaster University. 2015 Mar. <u>https://fhs.mcmaster.ca/ccapnr/documents/ScopingReviewNPPanelSize-</u> FINALPrinted17042015SJ 000.pdf
- 121. Heale R, Butcher M. Canada's First Nurse Practitioner-Led Clinic: A Case Study in Healthcare Innovation. *Nursing leadership* (Toronto, Ont.). 2010;23(3):21-9.
- 122. King, R., Tod, A., & Sanders, T. Regulation of advanced nurse practitioners: understanding the current processes in the UK. *Nurs Stand*, 2017: 32(14), 43-50.
- 123. Buppert, C. Billing for nurse practitioner services: Guidelines for NPs, physicians, employers, and insurers. *Medscape Nurses*, 2002: 4(1).
- 124. Delamaire, M. L., & Lafortune, G. Nurses in advanced roles: a description and evaluation of experiences in 12 developed countries. Organization for Economic Cooperation and Development, 2010.

- 125. Kleinpell R, Scanlon A, Hibbert D, et al. Addressing Issues Impacting Advanced Nursing Practice Worldwide. Online J Issues Nurs. 2014;19(2):5. DOI: 10.3912/OJIN.Vol19No02Man05
- 126. Nixon SA, Yeung E, Shaw JA, Kuper A, Gibson BE. Seven-Step Framework for Critical Analysis and Its Application in the Field of Physical Therapy. Phys Ther. 2017;97(2):249-257. DOI:10.2522/ptj.20160149
- 127. Greenhalgh T, Robert G, Bate P, Kyriakidou O, Macfarlane F, Peacock R. How to spread good ideas A systematic review of the literature on diffusion, dissemination and sustainability of innovations in health service delivery and organisation. *Report for the National Co-ordinating centre for NHS Service Delivery and Organisation R & D (NCCSDO)*, 1-426
- 128. Chambers DA, Glasgow RE, Stange KC. The dynamic sustainability framework: addressing the paradox of sustainment amid ongoing change. *Implementation Science*. 2013;8(1). doi:10.1186/1748-5908-8-117
- 129. Norful AA, De jacq K, Carlino R, Poghosyan L. Nurse Practitioner-Physician Comanagement: A Theoretical Model to Alleviate Primary Care Strain. *Ann Fam Med.* 2018;16(3):250-256. doi: 10.1370/afm.2230.
- Poghosyan L, Liu J, Norful AA. Nurse practitioners as primary care providers with their own patient panels and organizational structures: A cross-sectional study. *Int J Nurs Stud.* 2017;74:1-7. doi: 10.1016/j.ijnurstu.2017.05.004
- HayGroup. (2013). Final Report Submitted to: AOHC, AFHTO and NPAO: Developing a Provincial Compensation Structure for Primary Care Organizations. https://www.afhto.ca/wp-content/uploads/Developing-a-Provincial-Compensation-Structure-for-IPCOs-Final-Report-June-2013.pdf
- Canadian Institute of Health Information. (2018). Nursing in Canada. https://www.cihi.ca/en/nursing-in-canada-2018

- 133. Canadian Nurses Association. (n.d.). Health human resource component literature review report. Canadian Nurse Practitioner Initiative. https://www.cna-aiic.ca/-/media/cna/pagecontent/pdfen/04\_hhr\_appendixc.pdf?la=en&hash=A824BB5E65D3BB8CE9456F1E6FBC72AD97 CB6971
- Hutchison B. A long time coming: primary healthcare renewal in Canada. *Health Pap*. 2008;8(2):10-24. DOI: 10.12927/hcpap.2008.19704
- 135. Hutchison B, Abelson J, Lavis J. Primary care in Canada: so much innovation, so little change. *Health Aff* (Millwood). 2001;20(3):116-31. DOI:10.1377/hlthaff.20.3.116
- Hutchison B, Levesque JF, Strumpf E, Coyle N. Primary health care in Canada: systems in motion. *Milbank Q.* 2011;89(2):256-88. doi: <u>10.1111/j.1468-0009.2011.00628.x</u>
- 137. El-Jardali, F & Lavis, J. (2011). Issue Brief: Addressing the Integration of Nurse Practitioners in Primary Healthcare Settings in Canada. *McMaster Health Forum*. https://www.mcmasterforum.org/docs/default-source/product-documents/issuebriefs/nurse-practitioners-in-primary-healthcare-in-canada-ib.pdf?sfvrsn=2
- 138. Martin-Misener R, Donald F, Kilpatrick K, Bryant-Lukosius D, Rayner J, Landry V, Viscardi V, McKinlay R. Benchmarking for nurse practitioner patient panel size and comparative analysis of nurse practitioner pay scales: update of a scoping review. Hamilton, ON: McMaster University. 2015 Mar.
- Johnson JE. Working together in the best interest of patients. J Am Board Fam Med. 2013;26(3):241-3. doi: 10.3122/jabfm.2013.03.130075..
- 140. Naylor MD, Kurtzman ET. The role of nurse practitioners in reinventing primary care. Health Aff (Millwood). 2010;29(5):893-9. doi: 10.1377/hlthaff.2010.0440.
- 141. Aggarwal M, Hutchison BG. Toward a primary care strategy for Canada. Ottawa, Ontario: *Canadian Foundation for Healthcare Improvement*; 2013 Jan.
- 142. Harkless G, Vece L. Systematic review addressing nurse practitioner reimbursement policy: Part one of a four-part series on critical topics identified by the 2015 nurse

practitioner research agenda. *J Am Assoc Nurse Pract*. 2018;30(12):673-682. doi:10.1097/JXX.000000000000121

- 143. Considine J, Fielding K. Sustainable workforce reform: case study of Victorian nurse practitioner roles. *Aust Health Rev.* 2010;34(3):297-303. https://doi.org/10.1071/AH08727
- 144. Fox A, Gardner G, Osborne S. A theoretical framework to support research of health service innovation. *Aust Health Rev.* 2015;39(1):70-75. doi:10.1071/ah14031
- 145. Hovlid E, Bukve O, Haug K, Aslaksen AB, Von plessen C. Sustainability of healthcare improvement: what can we learn from learning theory?. *BMC Health Serv Res.* 2012;12:235. <u>https://doi.org/10.1186/1472-6963-12-235</u>
- 146. Keating SF, Thompson JP, Lee GA. Perceived barriers to the sustainability and progression of nurse practitioners. *International emergency nursing*, 2010 18(3), 147-153. <u>https://doi.org/10.1016/j.ienj.2009.09.003</u>
- 147. Lennox L, Maher L, Reed J. Navigating the sustainability landscape: a systematic review of sustainability approaches in healthcare. *Implementation Science*. 2018;13(1). doi:10.1186/s13012-017-0707-4
- 148. Martin GP, Weaver S, Currie G, Finn R, Mcdonald R. Innovation sustainability in challenging health-care contexts: embedding clinically led change in routine practice. *Health Serv Manage Res.* 2012;25(4):190-9. doi: 10.1177/0951484812474246
- Moore JE, Mascarenhas A, Bain J, Straus SE. Developing a comprehensive definition of sustainability. *Implement Sci.* 2017;12(1):110. doi: 10.1186/s13012-017-0637-1.
- 150. Sibthorpe BM, Glasgow NJ, Wells RW. Emergent themes in the sustainability of primary health care innovation. *Medical Journal of Australia*. 2005;183(S10). doi:10.5694/j.1326-5377.2005.tb07185.x
- 151. Stirman SW, Kimberly J, Cook N, Calloway A, Castro F, Charns M. The sustainability of new programs and innovations: a review of the empirical literature and recommendations for future research. *Implementation Science*. 2012;7(1). doi:10.1186/1748-5908-7-17

- 152. Wakerman J, Humphreys JS. Sustainable primary health care services in rural and remote areas: Innovation and evidence. *Australian Journal of Rural Health*. 2011;19(3):118-124. doi:10.1111/j.1440-1584.2010.01180.x
- 153. Aggarwal M, Hutchison B. *Toward A Primary Care Strategy for Canada* (Rep.).: http://www.cfhi-fcass.ca/Libraries/Reports/Primary-Care-Strategy-EN.sflb.ashx
- 154. Canadian Federation of Nurses Unions. Fulfilling Nurse Practitioners' Untapped Potential in Canada's Health Care System. Media Release. <u>https://nursesunions.ca/wpcontent/uploads/2018/06/CFNU\_UntappedPotential-Final-EN.pdf</u>
- 155. Hutchison B, Abelson J, Lavis J. Primary care in Canada: so much innovation, so little change. *Health Aff (Millwood)*. 2001;20(3):116-31. DOI:10.1377/hlthaff.20.3.116.
- 156. Hutchison B. A long time coming: primary healthcare renewal in Canada. *Health Pap*. 2008;8(2):10-24. DOI: 10.12927/hcpap.2008.19704
- 157. Hutchison B, Levesque JF, Strumpf E, Coyle N. Primary health care in Canada: systems in motion. *Milbank Q*. 2011;89(2):256-88. doi: 10.1111/j.1468-0009.2011.00628.x
- 158. El-Jardali F,Lavis JN. Issue Brief: Addressing the Integration of Nurse Practitioners in Primary Healthcare Settings in Canada. *McMaster Health Forum*. https://www.mcmasterforum.org/docs/default-source/product-documents/issuebriefs/nurse-practitioners-in-primary-healthcare-in-canada-ib.pdf?sfvrsn=2
- Martin-Misener R, Donald F, Kilpatrick K, Bryant-Lukosius D, Rayner J, Landry V, Viscardi V, McKinlay R. Benchmarking for nurse practitioner patient panel size and comparative analysis of nurse practitioner pay scales: update of a scoping review. Hamilton, ON: McMaster University. 2015 Mar.
   <u>https://fhs.mcmaster.ca/ccapnr/documents/ScopingReviewNPPanelSize-</u> FINALPrinted17042015SJ 000.pdf
- 160. Johnson JE. Working together in the best interest of patients. J Am Board Fam Med.
  2013;26(3):241-3. doi: 10.3122/jabfm.2013.03.130075.

- 161. Walt G, Gilson L. Reforming the health sector in developing countries: the central role of policy analysis. *Health Policy Plan*. 1994;9(4):353-70. doi:10.1093/heapol/9.4.353
- 162. Woltering L. 'Pilots Never Fail, Pilots Never Scale': Why the Global Development Community Needs a More Realistic Approach to Reaching Billions. <u>https://nextbillion.net/global-development-needs-realistic-approach-to-scale/. Published</u> 2019
- Uhlig, P. 2006 (April 3). "Collaborative Practice." Presentation, NPAO Accord Project Conference: Ahead of the Curve.
- 164. Bauer JC. Nurse practitioners as an underutilized resource for health reform: evidencebased demonstrations of cost-effectiveness. J Am Acad Nurse Pract. 2010;22(4):228-31. doi:10.1111/j.1745-7599.2010.00498.x
- 165. Brassard A. Making the case for NPs as primary care providers. *Am Nurse*. 2013: 45(3), 13-13.
- 166. Dicenso A, Auffrey L, Bryant-Lukosius D, et al. Primary health care nurse practitioners in Canada. *Contemporary Nurse*. 2007;26(1):104-115. doi:10.5172/conu.2007.26.1.104
- 167. Fund ME, Swanson-Hill A. Cost-effectiveness of nurse practitioner care. *Kansas Nurse*.
  2014: 89(1), 12-15. <u>http://amandasivewright.weebly.com/</u>
  uploads/3/0/8/4/30840229/reference 2.pdf. Published 2014. Accessed
- 168. Gould ON, Johnstone D, Wasylkiw L. Nurse practitioners in Canada: Beginnings, benefits, and barriers. *Journal of the American Academy of Nurse Practitioners*. 2007;19(4):165-171. doi:10.1111/j.1745-7599.2007.00210.x
- Poghosyan L, Lucero R, Rauch L, Berkowitz B. Nurse practitioner workforce: a substantial supply of primary care providers. *Nurs Econ.* 2012;30(5):268-74, 294.
- 170. Carter N, Martin-Misener R, Kilpatrick K, et al. The role of nursing leadership in integrating clinical nurse specialists and nurse practitioners in healthcare delivery in Canada. Nurs Leadersh (Tor Ont). 2010;23 Spec No 2010:167-85. https://doi.org/10.12927/cjnl.2010.22274

- 171. DiCenso A, & Bryant-Lukosius D. Clinical nurse specialists and nurse practitioners in Canada. <u>http://www.cfhi-fcass.ca/SearchResultsNews/10-06-01/b9cb9576-6140-4954-</u> aa57-2b81c1350936.aspx. Published 2010. Accessed
- Jayaratna N, Wood. B. Every Innovation is a New Thing but not Every New Thing is an Innovation. *Projectics / Proyéctica / Projectique*, 2008: 0(1), 13-25. doi:10.3917/proj.000.0013.
- 173. Esmail N, Barua B. Is Canada's Health Act a Barrier to Reform. *Fraser Institute*. https://www.fraserinstitute.org/studies/is-the-canada-health-act-a-barrier-to-reform. Published 1 May 2018. Accessed
- 174. Edmondson AC. Learning from failure in health care: frequent opportunities, pervasive barriers. *BMJ Quality & Safety*, 2014: 13(suppl 2), ii3-ii9.
- 175. Nolte E. (2018). How do we ensure that innovation in health service delivery and organization is implemented, sustained and spread. Policy Brief. Copenhagen: WHO Regional Office for Europe.

http://www.euro.who.int/\_\_data/assets/pdf\_file/0004/380731/pb-tallinn-03-eng.pdf?ua=1

- 176. Published 2018. Accessed
- 177. Simon HA. A behavioral model of rational choice. *The quarterly journal of economics*, 1955: 69(1), 99-118. DOI: 10.2307/1884852
- Clemens E, Cook J. Politics and Institutionalism: Explaining Durability and Change. Annual Review of Sociology, 1995: 25, 441-466. <u>www.jstor.org/stable/223512</u>
- 179. Herzlinger, R. E. (2006). Why innovation in health care is so hard. Harvard business review, 84(5), 58.
- 180. Howlett M. The lessons of failure: learning and blame avoidance in public policymaking. *International Political Science Review*. 2012;33(5):539-555. doi:10.1177/0192512112453603

- 181. MacNeil M, Koch M, Kuspinar A, Juzwishin D, Lehoux P, Stolee P. Enabling health technology innovation in Canada: Barriers and facilitators in policy and regulatory processes. *Health Policy*. 2019;123(2):203-214. doi: 10.1016/j.healthpol.2018.09.018.
- Braithwaite J. Changing how we think about healthcare improvement. BMJ.
   2018;361:k2014. doi: 10.1136/bmj.k2014.
- 183. Daddow O. How to Reduce the Risk of Policy Failure: Three actions you can take. 2010 https://apolitical.co/en/solution\_article/how-to-reduce-risk-policy-failure
- Hudson B, Hunter D, Peckham S. Policy failure and the policy-implementation gap: can policy support programs help?, *Policy Design and Practice* 2019:2:1, 1-14, DOI: 10.1080/25741292.2018.1540378
- 185. Benoit G, McCrae J, Shepheard M. Accountability in Modern Government: Recommendations for Change. <u>https://www.instituteforgovernment.org.uk</u> /sites/default/files/publications/Accountability\_modern\_government\_WEB.pdf. Published April 2018. Accessed January 18, 2020
- 186. Gavelin K, Burall S, Wilson R. Open Government: beyond static measures. A paper produced by Involve for the OECD. <u>http://www.oecd.org/gov/46560184.pdf</u>. Published July 2009. Accessed Accessed January 18, 2020
- 187. Paschke A, Dimancesco D, Vian T, Kohler JC, Forte G. Increasing transparency and accountability in national pharmaceutical systems. *Bulletin of the World Health Organization*. 2018: 96(11), 782.
- 188. Walugembe DR, Sibbald S, Ber MJL, Kothari A. Sustainability of public health interventions: where are the gaps? *Health Research Policy and Systems*. 2019;17(1). doi:10.1186/s12961-018-0405-y
- 189. Souliotis K, Agapidaki E, Peppou LE, Tzavara C, Samoutis G, Theodorou M. Assessing Patient Participation in Health Policy Decision-Making in Cyprus. *Int J Health Policy* Manag. 2016;5(8):461-466. <u>https://doi.org/10.15171/ijhpm.2016.78</u>

- Schell SF, Luke DA, Schooley MW, et al. Public health program capacity for sustainability: a new framework. *Implement Sci.* 2013;8:15. <u>https://doi.org/10.1186/1748-5908-8-15</u>
- 191. Pluye P, Potvin L, Denis J-L. Making public health programs last: conceptualizing sustainability. *Eval Prog Plann*. 2004;27(2):121–33.
- 192. Imenda S. Is there a conceptual difference between theoretical and conceptual frameworks? *Journal of Social Sciences*, 2013:38(2), 185-195. https://doi.org/10.1080/09718923.2014.11893249
- 193. Walt G, Gilson L. Reforming the health sector in developing countries: the central role of policy analysis. *Health Policy Plan*. 1994;9(4):353-70. doi:10.1093/heapol/9.4.353
- 194. Dixon-Woods M, Amalberti R, Goodman S, Bergman B, Glasziou P. Problems and promises of innovation: why healthcare needs to rethink its love/hate relationship with the new. BMJ quality & safety. 2011 Apr 1;20(Suppl 1):i47-51.
- 195. Lerner JS, Li Y, Valdesolo P, Kassam KS. Emotion and decision making. Annual review of psychology. 2015 Jan 3;66:799-823.
- 196. Shelton RC, Cooper BR, Stirman SW. The Sustainability of Evidence-Based Interventions and Practices in Public Health and Health Care. Annual Review of Public Health. 2018;39(1):55-76. doi:10.1146/annurev-publhealth-040617-014731
- Scheirer MA, Dearing JW. An Agenda for Research on the Sustainability of Public Health Programs. *American Journal of Public Health*. 2011;101(11):2059-2067. doi:10.2105/ajph.2011.300193
- Straus SE, Tetroe J, Graham I. Defining knowledge translation. *Canadian Medical Association Journal*. 2009;181(3-4):165-168. doi:10.1503/cmaj.081229
- 199. Considine J, Fielding K. Sustainable workforce reform: case study of Victorian nurse practitioner roles. *Aust Health Rev.* 2010;34(3):297-303. https://doi.org/10.1071/AH08727

- 200. Fox A, Gardner G, Osborne S. A theoretical framework to support research of health service innovation. *Aust Health Rev.* 2015;39(1):70-75. doi:10.1071/ah14031
- 201. Hovlid E, Bukve O, Haug K, Aslaksen AB, Von plessen C. Sustainability of healthcare improvement: what can we learn from learning theory?. *BMC Health Serv Res.* 2012;12:235. https://doi.org/10.1186/1472-6963-12-235
- 202. Keating SF, Thompson JP, Lee GA. Perceived barriers to the sustainability and progression of nurse practitioners. *International emergency nursing*, 2010 18(3), 147-153. <u>https://doi.org/10.1016/j.ienj.2009.09.003</u>
- 203. Lennox L, Maher L, Reed J. Navigating the sustainability landscape: a systematic review of sustainability approaches in healthcare. *Implementation Science*. 2018;13(1). doi:10.1186/s13012-017-0707-4
- 204. Martin GP, Weaver S, Currie G, Finn R, Mcdonald R. Innovation sustainability in challenging health-care contexts: embedding clinically led change in routine practice. *Health Serv Manage Res.* 2012;25(4):190-9. doi: 10.1177/0951484812474246
- 205. Moore JE, Mascarenhas A, Bain J, Straus SE. Developing a comprehensive definition of sustainability. *Implement Sci.* 2017;12(1):110. doi: 10.1186/s13012-017-0637-1.
- 206. Sibthorpe BM, Glasgow NJ, Wells RW. Emergent themes in the sustainability of primary health care innovation. *Medical Journal of Australia*. 2005;183(S10). doi:10.5694/j.1326-5377.2005.tb07185.x
- 207. Stirman SW, Kimberly J, Cook N, Calloway A, Castro F, Charns M. The sustainability of new programs and innovations: a review of the empirical literature and recommendations for future research. *Implementation Science*. 2012;7(1). doi:10.1186/1748-5908-7-17
- 208. Wakerman J, Humphreys JS. Sustainable primary health care services in rural and remote areas: Innovation and evidence. *Australian Journal of Rural Health*. 2011;19(3):118-124. doi:10.1111/j.1440-1584.2010.01180.x
- 209. Aggarwal M, Hutchison B. *Toward A Primary Care Strategy for Canada* (Rep.).: http://www.cfhi-fcass.ca/Libraries/Reports/Primary-Care-Strategy-EN.sflb.ashx

- 210. Canadian Federation of Nurses Unions. Fulfilling Nurse Practitioners' Untapped Potential in Canada's Health Care System. Media Release. <u>https://nursesunions.ca/wpcontent/uploads/2018/06/CFNU\_UntappedPotential-Final-EN.pdf</u>
- 211. Hutchison B, Abelson J, Lavis J. Primary care in Canada: so much innovation, so little change. *Health Aff (Millwood)*. 2001;20(3):116-31. DOI:10.1377/hlthaff.20.3.116.
- 212. Hutchison B. A long time coming: primary healthcare renewal in Canada. *Health Pap*.
  2008;8(2):10-24. DOI: 10.12927/hcpap.2008.19704
- 213. Hutchison B, Levesque JF, Strumpf E, Coyle N. Primary health care in Canada: systems in motion. *Milbank Q*. 2011;89(2):256-88. doi: 10.1111/j.1468-0009.2011.00628.x
- 214. El-Jardali F,Lavis JN. Issue Brief: Addressing the Integration of Nurse Practitioners in Primary Healthcare Settings in Canada. *McMaster Health Forum*. https://www.mcmasterforum.org/docs/default-source/product-documents/issuebriefs/nurse-practitioners-in-primary-healthcare-in-canada-ib.pdf?sfvrsn=2
- 215. Martin-Misener R, Donald F, Kilpatrick K, Bryant-Lukosius D, Rayner J, Landry V, Viscardi V, McKinlay R. Benchmarking for nurse practitioner patient panel size and comparative analysis of nurse practitioner pay scales: update of a scoping review. Hamilton, ON: McMaster University. 2015 Mar. https://fhs.mcmaster.ca/ccapnr/documents/ScopingReviewNPPanelSize-FINALPrinted17042015SJ\_000.pdf
- 216. Johnson JE. Working together in the best interest of patients. J Am Board Fam Med.
  2013;26(3):241-3. doi: 10.3122/jabfm.2013.03.130075.
- Walt G, Gilson L. Reforming the health sector in developing countries: the central role of policy analysis. *Health Policy Plan*. 1994;9(4):353-70. doi:10.1093/heapol/9.4.353
- 218. Woltering L. 'Pilots Never Fail, Pilots Never Scale': Why the Global Development Community Needs a More Realistic Approach to Reaching Billions. <u>https://nextbillion.net/global-development-needs-realistic-approach-to-scale/. Published</u> 2019

- Uhlig, P. 2006 (April 3). "Collaborative Practice." Presentation, NPAO Accord Project Conference: Ahead of the Curve.
- 220. Bauer JC. Nurse practitioners as an underutilized resource for health reform: evidencebased demonstrations of cost-effectiveness. J Am Acad Nurse Pract. 2010;22(4):228-31. doi:10.1111/j.1745-7599.2010.00498.x
- 221. Brassard A. Making the case for NPs as primary care providers. *Am Nurse*. 2013: 45(3), 13-13.
- 222. Dicenso A, Auffrey L, Bryant-Lukosius D, et al. Primary health care nurse practitioners in Canada. *Contemporary Nurse*. 2007;26(1):104-115. doi:10.5172/conu.2007.26.1.104
- Fund ME, Swanson-Hill A. Cost-effectiveness of nurse practitioner care. *Kansas Nurse*.
   2014: 89(1), 12-15. <u>http://amandasivewright.weebly.com/</u>
   <u>uploads/3/0/8/4/30840229/reference 2.pdf</u>. Published 2014. Accessed
- 224. Gould ON, Johnstone D, Wasylkiw L. Nurse practitioners in Canada: Beginnings, benefits, and barriers. *Journal of the American Academy of Nurse Practitioners*. 2007;19(4):165-171. doi:10.1111/j.1745-7599.2007.00210.x
- 225. Poghosyan L, Lucero R, Rauch L, Berkowitz B. Nurse practitioner workforce: a substantial supply of primary care providers. *Nurs Econ.* 2012;30(5):268-74, 294.
- 226. Carter N, Martin-Misener R, Kilpatrick K, et al. The role of nursing leadership in integrating clinical nurse specialists and nurse practitioners in healthcare delivery in Canada. *Nurs Leadersh (Tor Ont)*. 2010;23 Spec No 2010:167-85. https://doi.org/10.12927/cjnl.2010.22274
- 227. DiCenso A, & Bryant-Lukosius D. Clinical nurse specialists and nurse practitioners in Canada. <u>http://www.cfhi-fcass.ca/SearchResultsNews/10-06-01/b9cb9576-6140-4954-aa57-2b81c1350936.aspx</u>. Published 2010. Accessed
- 228. Jayaratna N, Wood. B. Every Innovation is a New Thing but not Every New Thing is an Innovation. *Projectics / Proyéctica / Projectique*, 2008: 0(1), 13-25. doi:10.3917/proj.000.0013.

- 229. Esmail N, Barua B. Is Canada's Health Act a Barrier to Reform. *Fraser Institute*. <u>https://www.fraserinstitute.org/studies/is-the-canada-health-act-a-barrier-to-reform</u>. Published 1 May 2018. Accessed
- 230. Edmondson AC. Learning from failure in health care: frequent opportunities, pervasive barriers. *BMJ Quality & Safety*, 2014: 13(suppl 2), ii3-ii9.
- 231. Nolte E. (2018). How do we ensure that innovation in health service delivery and organization is implemented, sustained and spread. Policy Brief. Copenhagen: WHO Regional Office for Europe.

http://www.euro.who.int/\_\_data/assets/pdf\_file/0004/380731/pb-tallinn-03-eng.pdf?ua=1

- 232. Published 2018. Accessed
- 233. Simon HA. A behavioral model of rational choice. *The quarterly journal of economics*, 1955: 69(1), 99-118. DOI: 10.2307/1884852
- 234. Clemens E, Cook J. Politics and Institutionalism: Explaining Durability and Change. Annual Review of Sociology, 1995: 25, 441-466. <u>www.jstor.org/stable/223512</u>
- 235. Herzlinger, R. E. (2006). Why innovation in health care is so hard. Harvard business review, 84(5), 58.
- 236. Howlett M. The lessons of failure: learning and blame avoidance in public policymaking. *International Political Science Review*. 2012;33(5):539-555. doi:10.1177/0192512112453603
- 237. MacNeil M, Koch M, Kuspinar A, Juzwishin D, Lehoux P, Stolee P. Enabling health technology innovation in Canada: Barriers and facilitators in policy and regulatory processes. *Health Policy*. 2019;123(2):203-214. doi: 10.1016/j.healthpol.2018.09.018.
- 238. Braithwaite J. Changing how we think about healthcare improvement. BMJ.2018;361:k2014. doi: 10.1136/bmj.k2014.
- 239. Daddow O. *How to Reduce the Risk of Policy Failure: Three actions you can take.* 2010 https://apolitical.co/en/solution\_article/how-to-reduce-risk-policy-failure

- Hudson B, Hunter D, Peckham S. Policy failure and the policy-implementation gap: can policy support programs help?, *Policy Design and Practice* 2019:2:1, 1-14, DOI: 10.1080/25741292.2018.1540378
- 241. Benoit G, McCrae J, Shepheard M. Accountability in Modern Government: Recommendations for Change. <u>https://www.instituteforgovernment.org.uk</u> /sites/default/files/publications/Accountability\_modern\_government\_WEB.pdf.
   Published April 2018. Accessed January 18, 2020
- 242. Gavelin K, Burall S, Wilson R. Open Government: beyond static measures. A paper produced by Involve for the OECD. <u>http://www.oecd.org/gov/46560184.pdf</u>. Published July 2009. Accessed Accessed January 18, 2020
- 243. Paschke A, Dimancesco D, Vian T, Kohler JC, Forte G. Increasing transparency and accountability in national pharmaceutical systems. *Bulletin of the World Health Organization*. 2018: 96(11), 782.
- Walugembe DR, Sibbald S, Ber MJL, Kothari A. Sustainability of public health interventions: where are the gaps? *Health Research Policy and Systems*. 2019;17(1). doi:10.1186/s12961-018-0405-y
- 245. Souliotis K, Agapidaki E, Peppou LE, Tzavara C, Samoutis G, Theodorou M. Assessing Patient Participation in Health Policy Decision-Making in Cyprus. *Int J Health Policy* Manag. 2016;5(8):461-466. <u>https://doi.org/10.15171/ijhpm.2016.78</u>
- 246. Schell SF, Luke DA, Schooley MW, et al. Public health program capacity for sustainability: a new framework. *Implement Sci.* 2013;8:15. <u>https://doi.org/10.1186/1748-5908-8-15</u>
- 247. Pluye P, Potvin L, Denis J-L. Making public health programs last: conceptualizing sustainability. *Eval Prog Plann*. 2004;27(2):121–33.
- 248. Imenda S. Is there a conceptual difference between theoretical and conceptual frameworks? *Journal of Social Sciences*, 2013:38(2), 185-195. https://doi.org/10.1080/09718923.2014.11893249

- 249. Walt G, Gilson L. Reforming the health sector in developing countries: the central role of policy analysis. *Health Policy Plan*. 1994;9(4):353-70. doi:10.1093/heapol/9.4.353
- 250. Dixon-Woods M, Amalberti R, Goodman S, Bergman B, Glasziou P. Problems and promises of innovation: why healthcare needs to rethink its love/hate relationship with the new. BMJ quality & safety. 2011 Apr 1;20(Suppl 1):i47-51.
- 251. Lerner JS, Li Y, Valdesolo P, Kassam KS. Emotion and decision making. Annual review of psychology. 2015 Jan 3;66:799-823.
- 252. Shelton RC, Cooper BR, Stirman SW. The Sustainability of Evidence-Based Interventions and Practices in Public Health and Health Care. Annual Review of Public Health. 2018;39(1):55-76. doi:10.1146/annurev-publhealth-040617-014731
- 253. Scheirer MA, Dearing JW. An Agenda for Research on the Sustainability of Public Health Programs. *American Journal of Public Health*. 2011;101(11):2059-2067. doi:10.2105/ajph.2011.300193
- 254. Straus SE, Tetroe J, Graham I. Defining knowledge translation. *Canadian Medical Association Journal*. 2009;181(3-4):165-168. doi:10.1503/cmaj.081229